## FORM 6-K

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Report of Foreign Private Issuer

Pursuant to Rule 13a-16 or 15d-16 under the Securities Exchange Act of 1934

For the month of July, 2005

Commission File Number 333-125037

## **Royal Dutch Shell plc**

(Exact name of registrant as specified in its charter)

#### **England and Wales**

(State or Other Jurisdiction of Incorporation or Organization)

Carel van Bylandtlaan 30 2596 HR The Hague The Netherlands Tel: (011 31 70) 377 9111

(Address and telephone number of Registrant's principal executive offices)

Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F:

Form 20-F ☑ Form 40-F.o

| Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(1):                           |
|---|
| Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(7):                           |
| Indicate by check mark whether by furnishing the information contained in this Form, the registrant is also thereby furnishing the information to the |

Yes o No 🗹

If "Yes" is marked, indicate below the file number assigned to the registrant in connection with Rule 12g3-2(b): 82-\_\_\_\_

## **TABLE OF CONTENTS**

Royal Dutch Shell plc

DESCRIPTION OF ROYAL DUTCH SHELL ORDINARY SHARES

DESCRIPTION OF ROYAL DUTCH SHELL AMERICAN DEPOSITARY RECEIPTS

**SIGNATURES** 

**EXHIBITS** 

EX-99.1

EX-99.2

EX-99.4

#### **Royal Dutch Shell plc**

On May 19, 2005, Royal Dutch Shell plc (the "Registrant" or "Royal Dutch Shell") commenced an exchange offer (the "Exchange Offer") for all of the Ordinary Shares of Royal Dutch Petroleum Company (N.V. Koninklijke Nederlandsche Petroleum Maatschappij) ("Royal Dutch"). Concurrently, The "Shell" Transport and Trading Company, p.l.c. ("Shell Transport") published definitive documentation with respect to a scheme of arrangement of Shell Transport under English law pursuant to which Shell Transport would become a subsidiary of the Registrant (the "Scheme", and together with the Exchange Offer, the "Transaction").

On July 20, 2005, the Registrant, Royal Dutch and Shell Transport announced that all conditions to the Exchange Offer and the Scheme had been satisfied or waived and that the Transaction had been completed. The press release with respect to such announcement is attached hereto as Exhibit 99.1 and incorporated by reference herein.

Upon consummation of the Exchange Offer and the Scheme on July 20, 2005, the Registrant became the parent company of Royal Dutch, Shell Transport and, through Royal Dutch and Shell Transport, of the Royal Dutch/Shell Group of Companies.

Pursuant to the terms of the Exchange Offer and the Scheme, holders of Ordinary Shares of Royal Dutch ("Royal Dutch Ordinary Shares"), holders of Shell Transport ordinary shares (the "Shell Transport Ordinary Shares"), holders of Shell Transport bearer warrants and holders of American depositary receipts representing Shell Transport Ordinary Shares (the "Shell Transport ADRs") received, respectively:

for each Royal Dutch Ordinary Share held in New York registry form tendered:

1 Royal Dutch Shell Class A American depositary receipt (each, a "Class A ADR"; each Class A ADR represents 2 Royal Dutch Shell Class A ordinary shares (each, a "Class A ordinary share"))

for each Royal Dutch Ordinary Share held in bearer or Hague registry form tendered:

2 Class A ordinary shares

for each Shell Transport Ordinary Share (including Shell Transport Ordinary Shares to which holders of Shell Transport Bearer Warrants are entitled):

0.287333066 Royal Dutch Shell Class B ordinary shares (each, a "Class B ordinary share")

for each Shell Transport ADR:

0.861999198 Royal Dutch Shell Class B American depositary receipts (each, a "Class B ADR"; each Class B ADR represents 2 Class B ordinary share, Class A ADRs and Class B ADRs

## are collectively referred to as "ADRs")

The issuance of the Class A ordinary shares under the Exchange Offer was registered under the Securities Act of 1933 (the "Securities Act") pursuant to the Registrant's registration statement on Form F-4 (File No. 333-125037) (the "F-4 Registration Statement") filed with the Securities and Exchange Commission (the "SEC") on May 18, 2005 and declared effective on May 19, 2005. Registration under the Securities Act was not required in connection with the issuance of the Class B ordinary shares pursuant to the Scheme in accordance with Section 3(a)(10) of the Securities Act. The issuance of the Class A ADRs and the Class B ADRs was registered under the Securities Act pursuant to the Registrant's registration statements on Form F-6 (File Nos. 333-125035 and 333-125038, respectively) filed with the SEC on May 18, 2005 and declared effective on May 19, 2005. The F-4 Registration Statement contains additional information about the Transaction.

Pursuant to Rule 12-g3(c) under the Securities Exchange Act of 1934 (the "Exchange Act"), the Class A ordinary shares and the Class B ordinary shares are registered under Section 12(b) of the Exchange Act. In accordance with Exchange Act Rule 12a-8, the Class A ADRs and the Class B ADRs are exempt from the operation of Section 12(a) of the Exchange Act and, consequently, need not be registered under Section 12(b) of the Exchange Act.

The Class A ADRs and the Class B ADRs have been authorized for listing on the New York Stock Exchange. Class A ADRs and Class B ADRs will trade under the ticker symbols "RDS.A" and "RDS.B", respectively. For technical purposes only, the Class A ordinary shares and the Class B ordinary shares have been authorized for listing on the New York Stock Exchange in connection with the listing of the Class A ADRs and the Class B ADRs but not for trading. The description of the Class A ordinary shares, Class B ordinary shares, Class A ADRs and Class B ADRs is set forth below under the headings "Description of Royal Dutch Shell Ordinary Shares" and "Description of Royal Dutch Shell American Depositary Receipts".

The Royal Dutch Ordinary Shares, the Shell Transport Ordinary Shares and the Shell Transport ADRs were registered under Section 12(b) of the Exchange Act. Royal Dutch Ordinary Shares and Shell Transport ADRs were listed on the New York Stock Exchange. Shell Transport is delisting the Shell Transport Ordinary Shares and the Shell Transport ADRs from the New York Stock Exchange and Shell Transport will today file a Form 15 with the SEC to terminate the registration thereof under the Exchange Act.

The Registrant intends to request that Royal Dutch seeks to delist the Royal Dutch Ordinary Shares in New York registry form from the New York Stock Exchange as soon as reasonably practicable in accordance with applicable rules.

#### DESCRIPTION OF ROYAL DUTCH SHELL ORDINARY SHARES

The following is a summary of the material terms of Royal Dutch Shell's ordinary shares, including brief descriptions of the provisions contained in our memorandum of association and articles of association and applicable laws of England in effect on the date of this document. This summary does not purport to include complete statements of these provisions. References to the provisions of our memorandum of association and articles of association are qualified in their entirety by reference to our full memorandum of association and articles of association which are filed as an exhibit to the F-4 Registration Statement. See "Description of Royal Dutch Shell American Depositary Receipts" section below for more information about the rights of holders of our ADRs. For the purposes of the discussion below, references to "we", "us" and "our" refer to Royal Dutch Shell.

## **Share Capital**

Once the shares accepted for tender in the Transaction as of July 20, 2005 have settled (and assuming full settlement), our authorized, issued and fully paid share capital will be as follows:

|                                       | Authorized<br>(number) | Authorized (amount) | Issued<br>(number) | Issued<br>(amount) |
|---------------------------------------|------------------------|---------------------|--------------------|--------------------|
| Class A ordinary shares of €0.07 each | 3,795,877,216          | €265,711,405        | 3,795,277,216      | €265,669,405       |
| Class B ordinary shares of €0.07 each | 2,759,360,000          | €193,155,200        | 2,759,360,000      | €193,155,200       |
| Sterling deferred shares of £1 each   | 50,000                 | £50,000             | 50,000             | £50,000            |
| Unclassified shares of €0.07 each     | 3,101,000,000          | €217,070,000        | Nil                | Nil                |

There are also approximately 344,000,000 euro deferred shares of €0.07 authorized and outstanding.

The unclassified shares can be issued as Class A ordinary shares or Class B ordinary shares at the discretion of our board of directors. Any future issue of additional Class B ordinary shares will only be made after prior consultation with the Dutch Revenue Service.

All Class A ordinary shares and Class B ordinary shares will be fully paid and free from all liens equities, charges, encumbrances and other interest and not subject to calls of any kind. All Class A ordinary shares and Class B ordinary shares will rank equally for all dividends and distributions on our ordinary share capital declared. Our Class A ordinary shares and Class B ordinary shares are admitted to the Official List of the UK Listing Authority and to trading on the market for listed securities of the London Stock Exchange. Our Class A ordinary shares and Class B ordinary shares are also listed on Euronext Amsterdam. In addition, for technical purposes only, the New York Stock Exchange, Inc. has authorized the listing of Class A ordinary shares and Class B ordinary shares, with respect to which there will be no trading privileges.

Our current authorized share capital (once the shares accepted for tender in the Transaction as of July 20, 2005 have settled, and assuming full settlement) consists of (i) £50,000 divided into 50,000 sterling deferred shares of £1 each and (ii) €700,000,000

divided into 343,762,784 euro deferred shares of €0.07 each, 3,795,877,216 Class A ordinary shares, 2,759,360,000 Class B ordinary shares and 3,101,000,000 unclassified shares of €0.07 each to be classified as Class A ordinary shares or Class B ordinary shares upon issue at the discretion of our directors. As of December 31, 2004, our issued share capital consisted of 20,000 ordinary shares of £1 each, 30,000 sterling deferred shares of £1 each and 4,148,800,000 euro deferred shares of €0.07 each. All ordinary shares, sterling deferred shares and euro deferred shares are fully paid and not subject to calls for additional payments of any kind. As of July 20, 2005, trusts holding shares for the benefit of employee plans of the Shell Group held 168 million ordinary shares of Royal Dutch Shell with a book value of \$4,549 million and a face value of €12 million.

## **Shareholders Meetings**

Under English law, we are required in each year to hold an annual general meeting of shareholders in addition to any other meeting of shareholders that may be held. Not more than 15 months may elapse between the date of one annual general meeting of shareholders and that of the next.

Our directors have the power to convene a general meeting of shareholders at any time. In addition, our directors must convene a meeting upon the request of shareholders holding not less than 10 percent of our paid-up capital carrying voting rights at general meetings of shareholders. A request for a general meeting of shareholders must state the objects of the meeting, and must be signed by the requesting shareholders and deposited at our registered office. If our directors fail to give notice of such meeting to shareholders with 21 days from receipt of notice, the shareholders that requested the general meeting, or any of them representing more than one-half of the total voting rights of all shareholders that requested the meeting, may themselves convene a meeting, but any meeting so convened shall not be held after the expiration of 3 months. Any such meeting must be convened in the same manner, as readily as possible, as that in which meetings are to be convened by our directors.

We are required to provide at least 21 clear days' notice of any annual general meeting, any general meeting where a special resolution is to be voted upon, or to pass a resolution of which special notice under the Companies Act of England and Wales 1985, as amended (the "Companies Act"), has been given. "Special resolutions" generally involve proposals to:

- · change the name of a company;
- alter a company's capital structure;
- change or amend the rights of shareholders;
- permit a company to issue new shares for cash without applying shareholders' pre-emptive rights;
- amend a company's objects clause in its memorandum of association;

- · amend a company's articles of association; and
- carry out other matters for which a company's articles of association or the Companies Act prescribe that a "special resolution" is required.

At least 14 clear days' notice is required for all other general meetings.

Our articles of association require that any notice of general meetings must be in writing and must specify where the meeting is to be held, the date and time of the meeting and the general nature of the business of the meeting. The listing rules ("the Listing Rules") of the UK Financial Services Authority acting in its capacity as the competent authority for the purposes of Part VI of the UK Financial Services and Markets Act 2000 require us to inform holders of our securities of the holding of meetings which they are entitled to attend.

A shareholder is entitled to appoint a proxy (which is not required to be another shareholder) to represent and vote on behalf of the shareholder at any general meeting of shareholders, including the annual general meeting.

Business may not be transacted at any general meeting, including the annual general meeting, unless a quorum is present. A quorum is two people who are entitled to vote at that general meeting. They can be shareholders who are personally present or proxies for shareholders entitled to vote at that general meeting or a combination of both.

If a quorum is not present within five minutes of the time fixed for a general meeting to start or within any longer period not exceeding one hour (as decided by the Chairman of the meeting), (i) if the meeting was called by shareholders it will be canceled and (ii) any other meeting will be adjourned to any day (being not less than three nor more than 28 days later), time and place stated in the notice of the meeting. If the notice does not provide for this, the meeting shall be adjourned to a day, time and place decided upon by the Chairman of the meeting. One shareholder present in person or by proxy and entitled to vote will constitute a quorum at any adjourned general meeting.

#### Record dates

In relation to shares in uncertificated form, the holders of those shares that are on the register of members on the record date have the right to attend and vote at meetings. In relation to shares in certificated form, holders of those shares that are on the register of members at the time of a meeting of shareholders are entitled to attend and vote at meetings.

#### Voting rights

The Class A ordinary shares and Class B ordinary shares have identical voting rights and vote together as a single class on all matters including the election of directors unless a matter affects the rights of one class as a separate class. If a resolution affects the rights attached to either class of shares as a separate class, it must be approved either in writing

by shareholders holding at least three-quarters of the issued shares of that class by amount, excluding any shares of that class held as treasury shares, or by an extraordinary resolution passed at a separate meeting of the registered holders of the relevant class of shares.

"Extraordinary resolutions" are confined to matters out of the ordinary course of business, such as a proposal to wind up the affairs of a company.

It is the intention that all voting at our general meetings will take place on a poll. On a poll, every holder of Class A ordinary shares or Class B ordinary shares present in person or by proxy has one vote for every share he holds.

This is subject to any rights or restrictions which are given to any class of shares. No shareholder is entitled to vote if he has been served with a restriction notice after failure to provide us with information concerning interests in his or her shares required to be provided under the Companies Act.

A "poll" is voting by means of a ballot where the number of shares held by each voting shareholder is counted, as opposed to voting by way of a show of hands where the actual number of shares held by voting shareholders is not taken into account.

Under the Companies Act, if a poll is demanded, the resolution conducted on a poll must be approved by holders of at least a majority of the votes cast at the meeting. Both special and extraordinary resolutions require the affirmative vote of at least 75% of the votes cast at the meeting to be approved.

#### **Dividends**

Under English law, dividends are payable on Class A ordinary shares and Class B ordinary shares only out of profits available for distribution, as determined in accordance with the Companies Act and under International Financial Reporting Standards.

Subject to the Companies Act, if our directors consider that our financial position justifies the declaration of a dividend, we can pay an interim dividend.

Our shareholders can declare dividends by passing an ordinary resolution. Dividends cannot exceed the amount recommended by our directors.

Dividends are payable to persons registered as shareholders on the record date relating to the relevant dividend.

All dividends will be divided and paid in proportions based on the amounts paid upon our shares during any period for which that dividend is paid.

Any dividend payable in cash relating to a share can be paid by sending a cheque, warrant or similar financial instrument payable to the shareholder entitled to the dividend

by post addressed to the shareholder's registered address or it can be made payable to someone else named in a written instruction from the shareholder and sent by post to the address specified in that instruction. A dividend can also be paid by inter-bank transfer or by other electronic means directly to an account with a bank or other financial institution named in a written instruction from the person entitled to receive the payment. Such bank or other financial institution must be in the United Kingdom other than in respect of our ordinary shares which are held within Euroclear Nederland and to which the Securities Giro Act (*Wet giraal effectenverkeer*) applies. Alternatively, a dividend can be paid in some other way requested in writing by a shareholder and agreed to by us. We will not be responsible for a payment which is lost or delayed.

Where any dividends or other amounts payable on a share have not been claimed, the directors can invest them or use them in any other way for our benefit until they are claimed. We will not be a trustee of the money and will not be liable to pay interest on it. If a dividend has not been claimed for 12 years after being declared or becoming due for payment, it will be forfeited and go back to us, unless the directors decide otherwise.

We expect that dividends on our outstanding Class B ordinary shares will be paid under the dividend access mechanism described below. As also noted below, any further issue of Class B ordinary shares is subject to advance consultation with the Dutch Revenue Service. See "Dividend Access Mechanism for Class B ordinary shares" below. Our articles of association provide that if any amount is paid by the issuer of the dividend access share by way of dividend on the dividend access share and paid by the dividend access trustee to any holder of Class B ordinary shares, the dividend that we would otherwise pay to such holder of Class B ordinary shares will be reduced by an amount equal to the amount paid to such holder of Class B ordinary shares by the dividend access trustee.

## Issuance of additional shares; other changes in share capital

Our articles of association provide that, subject to applicable law, Royal Dutch Shell can issue shares with any rights or restrictions attached to them as long as this is not restricted by any rights attached to existing shares. These rights or restrictions can be decided either by an ordinary resolution passed by the shareholders or by the directors as long as there is no conflict with any resolution passed by the shareholders. Accordingly, without further shareholder approval but subject to the limitations described above, including pre-emption rights, the directors could issue one or more series of preferred shares and establish the rights, preferences, redemption terms and other provisions of those shares.<sup>1</sup>

Subject to the provisions of applicable law and the provisions of our articles of association, shareholders can increase our share capital by passing an ordinary resolution. This resolution will fix the amount of the increase and the amount of new shares.

1 However, any further issue of Class B ordinary shares is subject to advance consultation with the Dutch Revenue Service.

Subject to applicable law and the provisions of our articles of association, shareholders can pass an ordinary resolution to do any of the following:

- (i) consolidate, or consolidate and then divide, all or any of our share capital into shares of a larger amount than the existing shares;
- (ii) divide some or all of our shares into shares of a smaller amount than the existing shares. The resolution can provide that holders of the divided shares will have different rights and restrictions if those rights or restrictions are of a kind which we can apply to new shares; and
- (iii) cancel any shares which have not been taken, or agreed to be taken, by anyone at the date of the resolution and reduce the amount of our share capital by the amount of the canceled shares.

Subject to applicable law and the provisions of our articles of association, shareholders can pass a special resolution to reduce our share capital, any capital redemption reserve, any share premium account or any other undistributable reserve in any way.

We may, subject to applicable law and existing shareholder rights, and to any requirements imposed by any relevant listing authority in respect of securities admitted to listing, purchase our own shares including redeemable shares.

#### Liquidation rights

If we are wound up (whether the liquidation is voluntary, under supervision of the court or by the court), the liquidator can, with the authority of an extraordinary resolution passed by our shareholders and any other sanction required by legislation, divide among the shareholders (excluding any shareholder holding shares as treasury shares) the whole or any part of our assets. For this purpose, the liquidator can set the value that the liquidator considers fair upon any property and decide how such division is carried out as between shareholders or different groups of shareholders.

#### Transfer of shares

Unless our articles of association provide otherwise, a shareholder may transfer some or all of his shares in certificated form to another person. A transfer of certificated shares must be either in the usual standard form or in any other form approved by the directors. The share transfer form for certificated shares must be signed or made effective in some other way by or on behalf of the person making the transfer.

In the case of a transfer of a certificated share, where the share is not fully paid, the share transfer form must also be signed or made effective in some other way by or on behalf of the person to whom the share is being transferred.

Unless our articles of association provide otherwise, a shareholder may transfer some or all of his shares in uncertificated form through CREST (the computerized settlement system to facilitate the transfer of title to shares in uncertificated form operated by CRESTCo Limited). Provisions of our articles of association do not apply to any uncertificated shares to the extent that those provisions are inconsistent with the holding of shares in uncertificated form or with the transfer of shares through CREST.

The person making a transfer will continue to be treated as a shareholder until the name of the person to whom the share is being transferred is put on the register for that share.

Our directors may, without giving any reasons, refuse to register the transfer of any shares which are not fully paid. Our directors may also refuse to register the transfer of any shares in the following circumstances:

#### Certificated shares

- (i) A share transfer form cannot be used to transfer more than one class of shares. Each class needs a separate form;
- (ii) Transfers may not be in favor of more than four joint holders; and
- (iii) The share transfer form must be properly stamped or certified or otherwise shown to our directors to be exempt from stamp duty and must be accompanied by the relevant share certificate and such other evidence of the right to transfer as our directors may reasonably require.

#### Uncertificated shares

- (i) Registration of a transfer of uncertificated shares can be refused in the circumstances set out in the Uncertificated Securities Regulations 2001 (SI 2001 No. 3755), as amended from time to time; and
- (ii) Transfers may not be in favor of more than four joint holders.

Title to certificated shares will be evidenced by entry in the register of our members and title to uncertificated shares will be evidenced by entry in the operator register maintained by CRESTCo (which forms part of the register of our members).

No share certificates will be issued in respect of our shares in uncertificated form. Since our shares will initially all be in uncertificated form, neither share certificates nor any temporary documents of title will initially be issued in respect of them. If any of our shares are converted to be held in certificated form, share certificates will be issued in respect of those shares in accordance with applicable legislation.

Our directors may refuse to register a transfer of any certificated shares by a person with a 0.25 per cent. or greater holding of the existing capital (calculated excluding any shares

held as treasury shares) if such a person has received a restriction notice (as defined in our articles of association) after failure to provide us with information concerning interests in these shares required to be provided under the legislation unless our directors are satisfied that they have been sold outright to an independent third party.

#### Dividend Access Mechanism for Class B ordinary shares

#### General

Class A ordinary shares and Class B ordinary shares are identical, except for the dividend access mechanism, which will only apply to the Class B ordinary shares.

Dividends paid on Class A ordinary shares have a Dutch source for tax purposes and are subject to Dutch withholding tax.

It is expected that holders of Class B ordinary shares will receive dividends through the dividend access mechanism. Dividends paid through the dividend access mechanism will have a UK source for UK and Dutch tax purposes and accordingly will not be subject to Dutch withholding tax insofar as these dividends are paid through the dividend access mechanism.

## Description of Dividend Access Mechanism

A dividend access share has been issued by Shell Transport to Hill Samuel Offshore Trust Company Limited as dividend access trustee. Pursuant to a declaration of trust, Hill Samuel Offshore Trust Company Limited will hold any dividends paid in respect of the dividend access share on trust for the holders of Class B ordinary shares from time to time and will arrange for prompt disbursement of such dividends to holders of Class B ordinary shares. Interest and other income earned on unclaimed dividends will be for the account of Shell Transport and any dividends which are unclaimed after 12 years will revert to Shell Transport. Holders of Class B ordinary shares will not have any interest in the dividend access share and will not have any rights against Shell Transport as issuer of the dividend access share. The only assets held on trust for the benefit of the holders of Class B ordinary shares will be dividends paid to the dividend access trustee in respect of the dividend access share. Any dividends paid on the dividend access share will have a UK source for Dutch and UK tax purposes; there will be no UK or Dutch withholding tax on such dividends and certain holders (not including U.S. holders) of Class B ordinary shares or Class B ADRs will be entitled to a UK tax credit in respect of their proportional share of such dividends.

The Shell Transport articles of association state that the maximum dividend that can be declared by Shell Transport on the dividend access share in respect of a specified period will be an amount equal to the aggregate dividend declared by us on the Class B ordinary shares in respect of such period. In addition, the dividends that Shell Transport may pay on the dividend access share in any year will be limited to a total of &3.3 billion. This limit can be varied by a resolution of the shareholders of Shell Transport from time to

time and will not be less than the aggregate dividends declared on the Class B ordinary shares in any year.

#### **Operation of the Dividend Access Mechanism**

Following the declaration of a dividend by us on the Class B ordinary shares, Shell Transport may declare a dividend on the dividend access share. Shell Transport will not declare a dividend on the dividend access share before we declare a dividend on the Class B ordinary shares, as Shell Transport will need to know what dividend we have declared on the Class B ordinary shares. This is to ensure that the dividend declared on the dividend access share does not exceed an amount equal to the total dividend declared by us on the Class B ordinary shares.

Before Shell Transport can declare any dividend, the Shell Transport directors will need to consider Shell Transport's financial condition and amount of distributable reserves. It is the expectation and the intention, although there can be no certainty, that holders of Class B ordinary shares will receive dividends via the dividend access mechanism.

To the extent that a dividend is declared by Shell Transport on the dividend access share and paid to the dividend access trustee, the holders of the Class B ordinary shares will be beneficially entitled to receive their share of that dividend pursuant to the declaration of trust (and arrangements will be made to ensure that the dividend is paid in the same currency in which they would have received a dividend from us). No dividend declared and paid by Shell Transport on the dividend access share will be paid to holders of Class A ordinary shares in respect of their Class A ordinary shares.

Our articles of association provide that if any amount is paid by Shell Transport by way of a dividend on the dividend access share and paid by the dividend access trustee to any holder of Class B ordinary shares, the dividend which we would otherwise pay on the Class B ordinary shares will be reduced by an amount equal to the amount paid to such holders of Class B ordinary shares by the dividend access trustee.

Royal Dutch Shell will have a full and unconditional obligation, in the event that the dividend access trustee does not pay an amount to holders of Class B ordinary shares on a cash dividend payment date (even if that amount has been paid to the dividend access trustee), to pay immediately the dividend declared on the Class B ordinary shares. The right of holders of Class B ordinary shares to receive distributions from the dividend access trustee will be reduced by an amount equal to the amount of any payment actually made by us on account of any dividend on Class B ordinary shares.

Any payment by Royal Dutch Shell will be subject to Dutch withholding tax (unless in any particular case an exemption is obtained under Dutch law or the provisions of an applicable tax treaty). If for any reason no dividend is paid on the dividend access share, holders of Class B ordinary shares will only receive dividends from Royal Dutch Shell directly.

The dividend access mechanism may be suspended or terminated at any time by our directors or the directors of Shell Transport, for any reason and without financial recompense. This might, for instance, occur in response to changes in relevant tax legislation.

The dividend access mechanism has been approved by the Dutch Revenue Service pursuant to an agreement (*vaststellingsovereenkomst*) with us and Royal Dutch dated October 26, 2004 as supplemented and amended by an agreement between the same parties dated April 25, 2005. The agreement states, among other things, that dividend distributions on the dividend access share by Shell Transport will not be subject to Dutch dividend withholding tax provided that the dividend access mechanism is structured and operated substantially as set out above. Royal Dutch Shell may not extend the dividend access mechanism to any future issuances of Class B ordinary shares without the approval of the Dutch Revenue Service. Accordingly, we would not expect to issue additional Class B ordinary shares unless we obtained that approval or determined that the continued operation of the dividend access mechanism was unnecessary. Any further issue of Class B ordinary shares is subject to advance consultation with the Dutch Revenue Service.

## Manner of holding shares

#### Holdings through Euroclear Nederland

We expect that the Admitted Institution (as defined below) or, if applicable, other bank or financial institution where a person who holds interests in our shares through Euroclear Nederland maintains a relevant securities account will send such person a statement detailing the interests in our shares such person holds through Euroclear Nederland. However, whether and, if so, how they do so, will depend on the individual arrangements between such Admitted Institution or other bank or financial institution and that person.

"Admitted Institution" means the institutions which hold Royal Dutch Shell ordinary shares on behalf of their clients through Euroclear Nederland as an admitted institution of Euroclear Nederland or, as the context so permits, which hold Royal Dutch Shell ordinary shares on behalf of their clients through an institution which is an admitted institution of Euroclear Nederland. References to Royal Dutch Shell shares shall, where the relevant shares are held by Euroclear Nederland in its capacity as central institute (*centraal instituut*) under the Dutch Securities Giro Act (*Wet giraal effectenverkeer*) and the context so permits, include references to interests held in such shares by other persons in accordance with the Dutch Securities Giro Act.

Euroclear Nederland has indicated that each person who holds interests in our shares through it will be able to exercise rights relating to those shares such that he will (subject to the individual arrangements between that person and the Admitted Institution or other bank or financial institution where that person maintains a relevant securities account):

- be able to attend and speak at, all of our general meetings;
- · be able to give directions as to voting at all of our general meetings; and

• be able to receive dividends via Euroclear Nederland and participate in capital events,

in each case, so far as is possible in accordance with the Securities Giro Act, other applicable law and the Euroclear Nederland rules and regulations issued pursuant to the Securities Giro Act and further subject to compliance by all concerned with any applicable policies and procedures.

#### Holdings through the Corporate Nominee Service

In order to allow the persons who hold our shares through the corporate nominee service provided by Lloyds TSB Bank plc (the "Corporate Nominee Service") to exercise rights relating to those shares, we have entered into an agreement with Lloyds TSB Bank plc (the "Corporate Nominee") requiring it to ensure that persons holding our shares through the Corporate Nominee Service will:

- receive notices of, and be able to attend and speak at, all of our general meetings;
- be able to give directions as to voting at all of our general meetings;
- have made available to them and be sent, on request, copies of our annual report and accounts and all the other documents issued to shareholders by us:
- be able to receive dividends via the Corporate Nominee Service;
- · be able to participate in capital events in the same manner as registered holders of the same class of our shares; and
- be treated in the same manner as registered holders of the same class of our shares in respect of all other rights attaching to those shares,

in each case, so far as is possible in accordance with the Uncertificated Securities Regulations 2001 and other applicable law. In particular, residents in, or citizens of, jurisdictions outside the United Kingdom should be aware that they will not be able to participate in capital events as registered holders of our shares unless the Corporate Nominee is satisfied that such participation or treatment would not breach any applicable laws or regulations in those jurisdictions.

It is the responsibility of persons resident in, or citizens of jurisdiction outside the United Kingdom to inform themselves of, and to satisfy themselves as to the full observance of, the laws of the relevant jurisdiction in connection with any applicable legal requirements in respect of holding our shares through the Corporate Nominee Service, including the obtaining of any governmental, exchange control or other consents which may be required, or the compliance with other necessary formalities that are required to be observed. If, due to applicable legal requirements, it is not permissible or practical to hold our shares through the Corporate Nominee Service, persons resident in, or citizen of, that jurisdiction should request that they be sent a share certificate for the Royal Dutch Shell ordinary shares to which they are entitled.

For so long as a person holds our shares through the Corporate Nominee Service, we will ensure that the Corporate Nominee sends each such person a statement of his holding of our shares at least once a year.

#### Change in the manner of holding our shares

Holders of our shares may, subject as set out below, change the manner in which they hold such shares so that they are held through Euroclear Nederland, through the Corporate Nominee Service or directly as the registered holder. The ability to change the manner of holding our shares is subject to, in each case, compliance with any relevant regulatory requirements and, in respect of holdings through the Corporate Nominee Service, the agreement of the Corporate Nominee and acceptance by the holder of our shares of the terms and conditions of the Corporate Nominee Service.

Holders of our shares who wish to change the manner in which they hold such shares are urged to consult their own legal, tax and financial advisers with respect to the legal, tax and cost consequences of any such change.

#### Repurchase of shares

Subject to applicable law and our articles of association, we may purchase our own shares if (a) in the case of an open-market purchase, authority to make the market purchase has been given by an ordinary resolution of our shareholders or; (b) in the case of an off-market purchase, authority has been given by a special resolution. However, we intend to comply with the guidance of the Association of British Insurers that authority to repurchase shares should be given by special resolution. We can only repurchase our own shares out of distributable reserves or the proceeds of a new issuance of shares made for the purposes of funding the repurchase.

Royal Dutch Shell has entered into agreements with the Dutch Revenue Service regarding the Dutch tax consequences of the repurchase of both Class A ordinary shares and Class B ordinary shares. Accordingly, Royal Dutch Shell will consider such Dutch tax consequences if and when it decides to repurchase ordinary shares.

## Shareholders' preemptive rights

Under the Companies Act, if we propose to issue for cash:

- · equity securities (which are securities carrying a right to participate in dividends or capital beyond a specified amount); or
- rights to subscribe for or convert into equity securities,

they must be offered first to each person who holds equity securities on the same or more favorable terms in proportion to those securities which is as nearly as practicable equal to the proportion in nominal value of the equity securities held by him or her to the aggregate issued equity securities. These pre-emption rights can be disapplied by a

special resolution passed by shareholders in a general meeting, either generally or specifically, for a maximum period not exceeding five years.

Subject to applicable law and our articles of association, any equity shares issued by us for cash must first be offered to existing shareholders in proportion to their existing holdings (the shareholders' pre-emption rights). Both the Companies Act and the Listing Rules allow for the disapplication of the shareholders' pre-emption rights. The pre-emption rights may be waived by a special resolution of the shareholders, either generally or specifically, for a maximum period not exceeding five years.

## Ability to pay commission on shares and to issue shares at a discount

In connection with any share issued, we can use all the powers given by applicable law to pay commissions or brokerage. Subject to the provisions of the Companies Act, we can pay the commissions in cash or by allotting shares or by a combination of both.

Subject to applicable law and our articles of association, we may also issue further shares of a class already issued at a discount to the market price. The Listing Rules limit the maximum discount under which shares may be issued in an open offer to 10 percent of the middle market price of those shares at the time of announcing the terms of the open offer. Furthermore, shares may not be allotted at less than their par value.

## Disputes between a shareholder or ADR holder and Royal Dutch Shell, any subsidiary, director or professional service provider

All disputes between a shareholder in its capacity as such and us or any of our subsidiaries or any of our or our subsidiaries' directors or former directors arising out of or in connection with our articles of association or otherwise and disputes between us or our subsidiaries and any of our or our subsidiaries' directors or former directors, including all claims made by us or any of our subsidiaries or on our behalf or on behalf of any of our subsidiaries against any such director, and disputes between a shareholder in its capacity as such and any of our professional service providers (which could include our auditors, legal counsel, bankers and ADR depositaries) that have agreed with us to be bound by the arbitration and exclusive jurisdiction provisions of our articles of association and between us and our professional service providers arising in connection with any such dispute between a shareholder and a professional service provider, shall be exclusively and finally resolved by arbitration in The Hague, The Netherlands under the Rules of Arbitration of the ICC. This would include all disputes arising under UK, Dutch or U.S. law (including securities laws), or under any other law, between parties covered by the arbitration provision.

The tribunal shall consist of three arbitrators to be appointed in accordance with the Rules of Arbitration of the ICC. The chairman must have at least 20 years experience as a lawyer qualified to practice in a common law jurisdiction which is within the Commonwealth and each other arbitrator must have at least 20 years experience as a qualified lawyer.

If a court or other competent authority in any jurisdiction determines that the arbitration requirement described above is invalid or unenforceable in any particular dispute in that jurisdiction, that dispute may only be brought in the courts of England and Wales.

The governing law of our articles of association is the substantive law of England.

We have incorporated arbitration clauses into all indemnities granted by us to our directors and into all service contracts between directors and our subsidiaries. We have incorporated an arbitration clause into the deposit agreements relating to the Class A ADRs and Class B ADRs which applies as between us and holders of the Class A ADRs and Class B ADRs (but not the depositaries).

Disputes relating to our failure or alleged failure to pay all or part of a dividend which has been declared and which has fallen due for payment will not be the subject of the arbitration and exclusive jurisdiction provisions of our articles of association.

We believe that the arbitration provision contained in our articles of association provides a predictable framework in which Royal Dutch Shell can run its affairs and assess risk, which is in the interest of Royal Dutch Shell and its shareholders as a whole. We also believe that the arbitration provision can provide the benefits of swiftness and economy, when compared with litigation as a dispute resolution mechanism. As a company based in Europe, most of whose shareholders are expected to be located in Europe, we consider it appropriate that the proceedings for resolving disputes among Royal Dutch Shell, its subsidiaries, its directors, its professional service providers (to the extent they have agreed to do so), and its shareholders and ADR holders, in their capacities as such, should be settled in Europe.

## History of our share capital

Our current authorized share capital (assuming the shares accepted for tender in the Transaction as of July 20, 2005 have settled, and assuming full settlement) consists of (i) £50,000 divided into 50,000 sterling deferred shares of £1 each and (ii) £700,000,000 divided into 343,762,784 euro deferred shares of £0.07 each, 3,795,877,216 Class A ordinary shares of £0.07 each, 2,759,360,000 Class B ordinary shares of £0.07 each and 3,101,000,000 unclassified shares of £0.07 each to be classified as Class A ordinary shares or Class B ordinary shares upon issue at the discretion of our directors. As of December 31, 2004, our issued share capital consisted of 20,000 ordinary shares of £1 each, 30,000 sterling deferred shares of £1 each and 4,414,800,000 euro deferred shares of £0.07 each. All ordinary shares, sterling deferred shares and euro deferred shares are fully paid and not subject to calls for additional payments of any kind.

We were incorporated with an authorized share capital of £1,000, divided into 1,000 ordinary shares of £1 each, one of which was issued to Instant Companies Limited.

The following alterations to our authorized and issued share capital have taken place since our incorporation:

- (i) On March 21, 2002, 300 ordinary shares were allotted and issued;
- (ii) On February 25, 2003, the authorized share capital was increased to £20,000 by the creation of 19,000 ordinary shares of £1 each ranking pari passu for all purposes with the existing ordinary shares. The directors were authorized to allot these shares pursuant to section 80 of the Companies Act:
- (iii) Subsequently on that date, 13,000 ordinary shares were allotted and issued, fully paid up in cash at par;
- (iv) On October 21, 2004, the authorized share capital was increased to £50,000 and €315,000,000 by the creation of:
  - (a) 30,000 sterling deferred shares of £1 each; and
  - (b) 4,500,000,000 euro deferred shares of €0.07 each.

The directors were authorized to allot these shares pursuant to section 80 of the Companies Act; and

- (v) Subsequently on that date, 4,148,800,000 euro deferred shares, 30,000 sterling deferred shares and 6,699 sterling ordinary shares were allotted and issued, fully paid up in cash at par;
- (vi) On April 27, 2005 the Royal Dutch Shell directors resolved, with immediate effect, to redeem 9,760,000 euro deferred shares for €0.01 in total, in accordance with the rights attaching to those shares due to there being more euro deferred shares on issue than were necessary to meet a full acceptance of the offer;
- (vii) On May 12, 2005, our authorized share capital was increased to £50,000 and £700,000,000 by the creation of:
  - 600,000 Class A ordinary shares of €0.07 each;
  - 2,759,360,000 Class B ordinary shares of €0.07 each; and
  - 2,740,040,000 unclassified shares of €0.07 each (to be classified as Class A ordinary shares or Class B ordinary shares upon allotment at the discretion of our directors<sup>2</sup>);

and our directors were authorized to allot relevant securities (as defined in the Companies Act) up to an aggregate nominal amount of €193,155,200 in connection with the Scheme of Arrangement;

<sup>2</sup> However, any future issue of Class B ordinary shares will only be made after prior consultation with the Dutch Revenue Service.

- (viii) on May 12, 2005, the 360,960,000 unissued euro deferred shares were re-classified as unclassified shares (to be classified as Class A ordinary shares or Class B ordinary shares upon allotment at the discretion of our directors);
- (ix) on May 13, 2005, our directors resolved to allot, conditional upon the Scheme of Arrangement becoming effective, Class B ordinary shares up to an aggregate nominal value of €193,155,200 to Relevant Holders (as that term is defined in the Scheme of Arrangement) in accordance with the terms of the Scheme of Arrangement;
- (x) on May 13, 2005, a special resolution was passed conditional on the offer of Royal Dutch Shell Class A ordinary shares for Royal Dutch ordinary shares becoming unconditional (*gestand wordt gedaan*) in all respects:
  - (A) re-classifying as Class A ordinary shares, immediately upon the offer being declared unconditional (*gestand wordt gedaan*) in all respects, such number of issued euro deferred shares as is equal to the number of Royal Dutch ordinary shares validly tendered in the offer acceptance period multiplied by two;
  - (B) re-classifying as Class A ordinary shares, on each occasion that Royal Dutch ordinary shares are validly tendered to the offer in the subsequent acceptance period (if any), such number of issued euro deferred shares as is equal to that number of Royal Dutch ordinary shares so tendered multiplied by two; and
  - (C) re-classifying as Class A ordinary shares, on each occasion that Royal Dutch ordinary shares are offered to Royal Dutch Shell for exchange into Class A ordinary shares after the later of the expiry of the offer acceptance period and the expiry of the subsequent acceptance period (if any) but at the absolute discretion of the Royal Dutch Shell directors (and subject to applicable law), such number of issued euro deferred shares as is equal to that number of Royal Dutch ordinary shares so offered multiplied by two;
- (xi) on May 13, 2005, a special resolution was passed, conditional upon the Scheme of Arrangement becoming effective, reclassifying our sterling ordinary shares as sterling deferred shares;
- (xii) on July 18, 2005, the offer to exchange Royal Dutch ordinary shares for Royal Dutch Shell Class A ordinary shares expired;
- (xiii) on July 20, 2005, the order sanctioning the Scheme of Arrangement by the Registrar of Companies in England and Wales was registered;
- (xiv) on July 20, 2005, the offer was declared unconditional (*gestanddoening*) and 1,897,638,608 Royal Dutch ordinary shares were accepted in exchange for 3,795,277,216 Royal Dutch Shell Class A ordinary shares; and

(xv) on July 20, 2005, the Transaction was completed and Royal Dutch Shell ordinary shares commenced trading on Euronext Amsterdam and the London Stock Exchange.

#### **Objects and Purposes**

We are incorporated under the name Royal Dutch Shell plc, and are registered at Companies House, Cardiff with company number 04366849, and the Chamber of Commerce, The Hague under number 34179503. Our registered office is at Shell Centre, London, SE1 7NA, UK and our headquarters are at Carel van Bylandtlaan 30, 2596 HR The Hague, The Netherlands. Our memorandum of association provides that our primary objective is to carry on the business of a holding company.

## Headquarters

We will, as required by our articles of association, have a single corporate headquarters in The Netherlands. The meaning of "headquarters" under our articles of association is established by our board of directors and can only be amended by a resolution of our board of directors in respect of which two-thirds of those Royal Dutch Shell directors who are present and voting vote in favor.

Our board of directors has resolved that "headquarters" shall mean the place of our effective management where:

- · substantially all members of our executive committee will have their main offices and carry out their managerial activities;
- a majority of the heads of the key functions will have their main office and carry out a substantial majority of their activities;
- a corporate secretariat will be located, which will provide all secretarial services to our executive committee, to our board of directors and to the committees of our board of directors;
- · in principle, all meetings of our board of directors, our executive committee and the committees of our board of directors will be held; and
- the majority of the main business units will have their effective place of management.

We are considered a resident of The Netherlands for Dutch and UK tax purposes.

## **Board of Directors**

*General*. Under English law, we are required to have a single-tier board of directors. Our board is headed by a non-executive Chairman and has a majority of independent non-executive directors. Our board currently comprises ten non-executive directors (including

the Chairman) and five executive directors. The directors can delegate any of their powers or discretions to an individual director or committees of one or more persons. All committees must comply with any regulations laid down by the directors.

The directors have delegated some responsibilities to the executive committee and others to the non-executive committees referred to below. The directors can exercise all of our powers, except where our memorandum of association, the articles of association or applicable law limit the use of powers to the shareholders voting to do so at a general meeting, for example to increase the total fees payable to all of the directors. The directors can give a director any of the powers which they have jointly as directors (with the power to sub-delegate). These powers can be given on terms and conditions decided on by the directors either in parallel with, or in place of, the powers of the directors acting jointly.

The directors can appoint anyone as our attorney by granting a power of attorney. Attorneys can either be appointed directly by the directors or the directors can give someone else the power to select attorneys. The directors or the persons who are authorized by them to select attorneys can decide on the purposes, powers, authorities and discretions of attorneys. However, they cannot give an attorney any power, authority or discretion which the directors do not have under our articles of association.

*Size.* Under our articles of association, we must have a minimum of three directors and a maximum of 20 directors (disregarding alternate directors (described in greater detail below). However, these restrictions can be amended by resolution of our board of directors.

Appointment and Election of Directors. Our shareholders may, by passing an ordinary resolution, elect any eligible willing person to be a director, either as an additional director or to fill a vacancy.

Subject to our articles of association, our directors can appoint any willing person to be a director, either as an additional director or as a replacement for another director. Any director appointed by our directors must retire from office at the first annual general meeting after his or her appointment and is then eligible for election.

Any director can appoint any person (including another director) to act in his place (called an "alternate director"). That appointment requires the approval of the directors, unless previously approved by the directors or unless the appointee is another director.

Retirement Our articles of association provide that at every annual general meeting any director who was in office at the time of the two previous annual general meetings and who did not retire at either of them must retire. Additional provisions in respect of retirement apply to our 2006 and 2007 annual general meetings. At the general meeting at which a director retires, shareholders can pass an ordinary resolution to re-elect the director or to elect another eligible person in his or her place.

A director who would not otherwise be required to retire must also retire if he is aged 70 or more at the date of the meeting or if he has been in office, other than as a director holding an executive position, for a continuous period of nine years or more at the date of the meeting. Any such director will be eligible to stand for re-election

*Removal of Directors* Under the Companies Act, our shareholders may remove any director without cause by ordinary resolution, irrespective of any provision of our articles of association or any service contract a director may have with us, provided that we are given 28 clear days' notice of the resolution. In these circumstances, we may be required by a service contract to pay compensation to the removed director.

In addition to any power to remove directors conferred by legislation, the company can pass a special resolution to remove a director from office even though his time in office has not ended.

*Committees*. The directors can delegate any of their powers or discretions to committees of one or more persons. Unless the directors decide not to allow this, any committee can sub-delegate any of its powers or discretions to sub-committees.

Our directors have delegated powers to the Executive Committee, Audit Committee, Nomination and Succession Committee, Remuneration Committee and Social Responsibility Committee.

*Executive Committee.* We have an executive committee which comprises the Chief Executive, the Chief Financial Officer and the other Executive Directors. The executive committee is responsible for our overall business and affairs and has the final authority in all matters of management that are not within the duties and authorities of our board of directors or our shareholders' meeting. The executive committee implements all resolutions of our Board of Directors and supervises the management of our businesses.

*Directors' Liability.* Under English law, each of our directors has a fiduciary duty to act in our best interest. This duty includes an obligation not to create an actual or potential conflict between the director's duty to us and duties to any other person or his personal interests as well as an obligation to exercise his or her powers only in accordance with our memorandum of association and articles of association and any applicable legislation. In addition, each of our directors is obligated under English law to exercise reasonable care and skill.

Limitation on Liability and Indemnification. Our articles of association provide that, as far as legislation allows, we can indemnify any director of the company, of an associated company or of any affiliate against any liability and that we can purchase and maintain insurance against any liability for any director of the company, of any associated company or of any affiliate.

English law provides that a company may indemnify a director against any liability except for: (i) any indemnity against any liability incurred by the director to the company or any associated company, (ii) any indemnity against any liability incurred by the director to pay a fine imposed in criminal proceedings or a sum payable to a regulatory authority by way of a penalty in respect of non-compliance with any requirement of a regulatory nature, and (iii) any indemnity against any liability incurred by the director in defending criminal proceedings in which he is convicted, or in defending civil proceedings brought by the company or an associated company in which judgment is given against him or in connection with an application under certain sections of the

Companies Act (acquisition of shares by an innocent nominee and relief in the case of honest and reasonable conduct) in relation to which the court refuses to grant him relief.

Class Action Suits and Shareholder Derivative Suits. The following provisions would only apply in circumstances where the arbitration provisions of our articles of association would be invalid or inapplicable. While English law permits a shareholder to initiate a lawsuit on behalf of the company only in limited circumstances, the Companies Act permits a shareholder whose name is on the register of shareholders of the company to apply for a court order:

- (i) when the company's affairs are being or have been conducted in a manner unfairly prejudicial to the interests of all or some shareholders, including the shareholder making the claim; or
- (ii) when any act or omission of the company is or would be so prejudicial. A court has wide discretion in granting relief, and may authorize civil proceedings to be brought in the name of the company by a shareholder on terms that the court directs.

Except in these limited circumstances, English law does not generally permit class action lawsuits by shareholders on behalf of the company or on behalf of other shareholders.

Transactions with Interested Directors. Under the Listing Rules, we must obtain shareholder approval for certain transactions with related parties (which includes certain transactions with directors). The Listing Rules provide that an announcement, a circular and prior approval of the shareholders in a general meeting will be required before such a transaction is entered into. The related party will not be allowed to vote on the resolution. Our articles of association state that, if legislation allows and provided that a director discloses the nature and extent of his or her interest to the other directors, he is permitted to:
(i) have an interest in any contract with, or involving, us or any other company in which we have an interest; (ii) hold any other position (other than as an auditor) with us as well as being a director; (iii) acting alone, or through a firm with which he is associated, do paid professional work for us or another company in which we have an interest (other than as an auditor); and (iv) hold any position within, or be a shareholder of, or have any other kind of interest in any company in which we have an interest. Except as provided for in our articles of association, a director cannot vote on, or be counted in a quorum in relation to, any resolution of the board of directors on any contract in which he has an interest which the director knows is material. Interests purely as a result of an interest in our securities will be disregarded for these purposes. Our articles of association provide that a director can vote, and be counted for purposes of a quorum, on those conflict of interest transactions specified in article 105(E) if the only material interest that director has in the transaction is one of those specified in article 105(E) of our articles of association.

Our articles of association provide that holders of our shares may by ordinary resolution suspend or relax the list contained in article 105 to any extent or to ratify any contract which has not been properly authorized in accordance with our articles of association.

Under our articles of association:

- (1) a director may not vote or be counted in the quorum in respect of any matter in which he is materially interested including any matter related to his own compensation;
- (2) the directors may exercise Royal Dutch Shell's power to borrow money provided that the borrowings of the Shell Group shall not, without the consent of an ordinary resolution of shareholders of Royal Dutch Shell, exceed two times Royal Dutch Shell's adjusted capital and reserves (these powers relating to borrowing may only be varied by special resolution of shareholders);
  - (3) directors over age 70 must retire at each Annual General Meeting, but are eligible for re-election; and
  - (4) directors are not required to hold shares of Royal Dutch Shell to be qualified to be a director.

#### Shareholders' Information Rights.

Except when closed under the provisions of the Companies Act (i.e. where a company, on giving notice by advertisement in a newspaper circulating in the district in which the company's registered office is situated, closes the register of members for any time or times not exceeding in the whole 30 days in each year), the register and index of names of our shareholders may be inspected during business hours:

- (i) for free, by our shareholders; and
- (ii) for a fee by any other person.

In both cases, the documents may be copied for a fee. Our shareholders may also, without charge, during business hours:

- (i) inspect minutes of shareholders' meetings and obtain copies of the minutes for a fee; and
- (ii) inspect service contracts of the company's directors, if the contracts have an unexpired term of more than 12 months or require more than 12 months' notice to terminate.

In addition, our published annual accounts are required to be available for shareholders at a general meeting, and a shareholder is entitled to a copy of these accounts.

#### **Disclosure of Shareholder Interests**

Section 198 of the Companies Act imposes an obligation upon a person who acquires or ceases to have notifiable interest in the relevant share capital of a public company to notify the company of that fact within 2 days (excluding weekends and bank holidays) of his knowing of its occurrence. The disclosure threshold is 3%.

Section 212 of the Companies Act provides a public company with the statutory means to ascertain the persons who are or have within the last 3 years been interested in its relevant share capital and the nature of such interests.

The Royal Dutch Shell articles of association provide that in any statutory notice under section 212, Royal Dutch Shell will ask for details of those who have an interest and the extent of their interest in a particular holding. The Royal Dutch Shell articles of association also provide that when a person receives a statutory notice, he has 14 days to comply with it. If he does not do so or if he makes a statement in response to the notice which is false or inadequate in some important way, Royal Dutch Shell may restrict the rights relating to the identified shares, following notice. The restriction notice will state that the identified shares no longer give the shareholder any right to attend or vote either personally or by proxy at a shareholders' meeting or to exercise any right in relation to the shareholders' meetings. Where the identified shares make up 0.25% or more (in amount or in number) of the existing shares of a class at the date of delivery of the restriction notice, the restriction notice can also contain the following further restrictions: (i) the directors can withhold any dividend or part of a dividend or other money otherwise payable in respect of the identified shares without any liability to pay interest when such money is finally paid to the shareholder; and (ii) the directors can refuse to register a transfer of any of the identified shares which are certificated shares unless the directors are satisfied that they have been sold outright to an independent third party. Once a restriction notice has been given, the directors are free to cancel it or exclude any shares from it at any time they think fit. In addition, they must cancel the restriction notice within seven days of being satisfied that they were sold outright to an independent third party, they must cancel the restriction notice within seven days of receipt of the notification of the sale. The Royal Dutch Shell articles of association do not restrict in any way the provision of section 212 of the Companies Act.

The UK City Code on Takeovers and Mergers imposes rigorous disclosure requirements affecting parties to a proposed takeover, their "associates" and persons acting "in concert" in relation to the shares of a company. These requirements also extend to dealings by persons who directly or indirectly own or control (either before or as a result of the dealing) 1% or more of the equity shares in an offeror or offeree company or of any other class of shares relevant to the offer in question.

The UK Rules Governing Substantial Acquisitions of Shares require accelerated disclosure of acquisitions of shares or rights over shares where a person holds, or as a result of an acquisition, comes to hold shares or rights over shares representing 15 percent or more of the voting rights of a company whose shares are listed on the London Stock Exchange.

#### **Amendment of Articles of Association**

Under the Companies Act, our shareholders have power to amend the objects, or purpose, clause in our memorandum of association and any provision of our articles of association by special resolution, subject to, in the case of amendments to the objects clause of the memorandum of association, the right of dissenting shareholders to apply to the courts to cancel the amendments.

Under the Companies Act, our board of directors is not authorized to change the memorandum of association or the articles of association. Our articles of association provide that if permitted by legislation, the rights attached to any class of our shares can be changed if this is approved either in writing by shareholders holding at least three-quarters of the issued shares of that class by amount (excluding any shares of that class held as treasury shares) or by an extraordinary resolution passed at a separate meeting of the holders of the relevant class of shares.

## DESCRIPTION OF ROYAL DUTCH SHELL AMERICAN DEPOSITARY RECEIPTS

#### General

JPMorgan Chase Bank, N.A., as depositary for our Class A ADRs and The Bank of New York as depositary for our Class B ADRs, will execute and deliver the Class A ADRs and Class B ADRs, respectively (collectively, the "ADRs"). Each Class A ADR and Class B ADR is a certificate evidencing a specific number of Class A or Class B American depositary shares ("Class A ADSs" and "Class B ADSs" and, collectively, "ADSs"), respectively. Each Class A ADS will represent two shares (or a right to receive two shares) deposited with the custodian of JPMorgan Chase Bank, N.A. Each Class B ADS will represent two shares (or a right to receive two shares) deposited with The Bank of New York. Each ADS will also represent any other securities, cash or other property which may be held by the depositary. The depositary's office at which the Class A ADRs will be administered is located at 4 New York Plaza, New York, New York 10004. The depositary's office at which the Class B ADRs will be administered is located at 101 Barclay Street, New York, New York 10286.

You may hold ADSs either directly (by having an ADR registered in your name) or indirectly through your broker or other financial institution. If you hold ADSs directly, you are an ADR holder. This description assumes you hold your ADSs directly. If you hold the ADSs indirectly, you must rely on the procedures of your broker or other financial institution to assert the rights of ADR holders described in this section. You should consult with your broker or financial institution to find out what those procedures are.

As an ADR holder, we will not treat you as one of our shareholders and you will not have shareholder rights. English law generally governs shareholder rights. The depositary or its nominee will be the holder of the shares underlying your ADSs. As a holder of ADRs, you will have ADR holder rights. A deposit agreement among us, the respective depositary and you, as an ADR holder, and the beneficial owners of ADRs sets out ADR holder rights as well as the rights and obligations of the respective depositary. New York law governs the deposit agreements and the ADRs except that the arbitration and exclusive jurisdiction provisions are governed by English law.

The following is a summary of the material provisions of the deposit agreement. For more complete information, you should read the entire deposit agreement and the form of ADRs. The deposit agreement relating to the Class A ADRs and the form of Class A

ADR relating thereto are attached hereto as Exhibits 99.2 and 99.3 and incorporated by reference herein. The deposit agreement relating to the Class B ADRs and the form of Class B ADR relating thereto are attached hereto as Exhibits 99.4 and 99.5 and incorporated by reference herein.

#### **Dividends and Other Distributions**

#### How will you receive dividends and other distributions on the shares?

The depositary has agreed to pay to you the cash dividends or other distributions it or the custodian receives on shares or other deposited securities, after deducting its fees and expenses. You will receive these distributions in proportion to the number of shares your ADSs represent.

• *Cash.* While the depositary may receive cash dividends and other distributions from us in U.S. dollars (in which case no conversion will be required) to the extent the depositary receives a cash dividend or other cash distribution in a currency other than U.S. dollars, the depositary will convert such cash dividend or other distribution we pay on the shares into U.S. dollars, if it can do so on a reasonable basis and can transfer the U.S. dollars to the United States. If that is not possible or if any government approval is needed and cannot be obtained, the deposit agreement allows the depositary to distribute the foreign currency only to those ADR holders to whom it is possible to do so. It will hold the foreign currency it cannot distribute for the account of the ADR holders who have not been paid. It will not invest the foreign currency and it will not be liable for any interest.

Before making a distribution, the depositary will deduct any withholding taxes that must be paid. It will distribute only whole U.S. dollars and cents and will round down fractional cents to the nearest whole cent. If the exchange rates fluctuate during a time when the depositary cannot convert the foreign currency, you may lose some or all of the value of the distribution.

- *Ordinary shares.* The depositary may distribute additional ADSs representing any shares we distribute as a dividend or free distribution. The depositary will only distribute whole ADSs. It will use its reasonable efforts to sell shares which would require it to deliver a fractional ADS and distribute the net proceeds in the same way as it does with cash. If the depositary does not distribute additional ADRs, the outstanding ADSs will also represent the new shares.
- *Rights to purchase additional shares.* If we offer holders of our securities any rights to subscribe for additional shares or any other rights, the depositary may make these rights available to you. If the depositary decides, after consultation with us, it is not legal or feasible to make the rights available but that it is practical to sell the rights, the depositary may sell the rights and distribute the proceeds in the same way as it does with cash. The depositary will allow rights that are not distributed or sold to lapse. *In that case, you will receive no value for them.*

If the depositary makes rights available to you, and you elect to exercise such rights, it will exercise the rights and purchase the shares on your behalf. The depositary will then deposit the shares and deliver ADSs to you. It will only exercise rights if you pay it the exercise price and any other charges the rights require you to pay.

U.S. securities laws may restrict transfers and cancellation of the ADSs represented by shares purchased upon exercise of rights. For example, you may not be able to trade these ADSs freely in the United States. In this case, the depositary may deliver restricted depositary shares that have the same terms as the ADSs described in this section except for changes needed to put the necessary restrictions in place.

• *Other Distributions.* The depositary will send to you anything else we distribute on deposited securities by any means it thinks is equitable and practical. If it cannot make the distribution in that way, the depositary has a choice, after consulting with us to the extent practical. It may decide to sell what we distributed and distribute the net proceeds, in the same way as it does with cash. However, the depositary is not required to distribute any securities (other than ADSs) to you unless it receives satisfactory assurance from us that it is legal to make that distribution.

The depositary is not responsible if it decides that it is unlawful or impractical to make a distribution available to any ADR holders. We have no obligation to register ADSs, shares, rights or other securities under the Securities Act. We also have no obligation to take any other action to permit the distribution of ADRs, shares, rights or anything else to ADR holders. This means that you may not receive the distributions we make on our shares or any value for them if it is deemed illegal or impractical for us to make them available to you.

## **Deposit and Withdrawal**

#### How are ADSs issued?

The depositary will deliver ADSs if you or your broker deposits shares or evidence of rights to receive shares with the relevant custodian. Upon payment of its fees and expenses and of any taxes or charges, such as stamp taxes or stock transfer taxes or fees, the depositary will register the appropriate number of ADSs in the names you request and will deliver the ADRs at its office to the persons you request.

#### How do ADS holders cancel an ADR and obtain shares?

You may surrender your ADRs at the respective depositary's office. Upon payment of its fees and expenses and of any taxes or charges, such as stamp taxes or stock transfer taxes or fees, the respective depositary will deliver (i) to the extent applicable, the shares, (ii) shares to an account designated by such owner with Euroclear Nederland or an Admitted Institution and (iii) and any other deposited securities underlying the ADR to you or a

person you designate at the office of the respective custodian. Or, in the case of certificated shares, at your request, risk and expense, the respective depositary will deliver the deposited securities at its office, if feasible.

#### **Voting Rights**

## How do you vote?

In the deposit agreements, upon the written request of a registered holder of Class A ADSs, the respective depositary endeavors to cause the appointment of such holder as each registered holder of ADSs its proxy with power to vote the number of shares its ADSs represent. This means that, subject to the procedures described below, if you are a registered holder of ADSs, you will have a right to attend and vote directly at shareholders' meetings. You also have a right to appoint the respective depositary your substitute and instruct it how to vote the number of shares your ADSs represent. The respective depositary will notify you of shareholders' meetings and arrange to deliver our voting materials to you if we ask it to. Those materials will describe the matters to be voted on and explain how you may vote directly or instruct the respective depositary how to vote. For instructions to be valid, they must reach the respective depositary by a date set by the respective depositary. In order for you to vote, the depositary must receive your request to be a proxy prior to the date specified for each meeting.

The respective depositary will try, as far as practical, subject to English law and the provisions of our articles of association, to vote the number of shares or other deposited securities represented by your ADSs as you instruct. The respective depositary will only vote or attempt to vote as you instruct.

We cannot ensure that you will receive voting materials or otherwise learn of an upcoming shareholders' meeting in time to ensure that you can instruct the respective depositary to vote your shares.

The respective depositary and its agents are not responsible for failing to carry out voting instructions or for the manner of carrying out voting instructions. This means that you may not be able to vote and there may be nothing you can do if your shares are not voted as you requested.

## Fees and Expenses

| Persons depositing shares or ADR holders must pay:     | For:  |
|--|---|
| \$5.00 (or less) per 100 ADSs (or portion of 100 ADSs) | <ul> <li>Issuance of ADSs, including issuances resulting from a distribution of shares or rights or other property</li> <li>Cancellation of ADSs for the purpose of withdrawal, including if the relevant deposit agreement terminates</li> </ul> |

## Persons depositing shares or ADR holders must pay:

A fee equivalent to the fee that would be payable if securities distributed to you had been shares and the shares had been deposited for issuance of ADSs

Registration or transfer fees

Expenses of the depositary in converting foreign currency to U.S. dollars

Expenses of the depositary

Taxes and other governmental charges payable on any ADR or share underlying an ADR, for example, stock transfer taxes, stamp duty or withholding taxes

#### For:

- Distribution of securities distributed to holders of deposited securities which are distributed by the respective depositaries to ADR holders
- Transfer and registration of shares on our share register to or from the name of the respective depositary or its agent when you deposit or withdraw shares
- Cable, telex and facsimile transmissions (when expressly provided in the deposit agreement)

## **Payment of Taxes**

The respective depositary may deduct the amount of any taxes owed from any payments to you. It may also sell deposited securities, by public or private sale, to pay any taxes owed. You will remain liable if the proceeds of the sale are not enough to pay the taxes. If the respective depositary sells deposited securities, it will, if appropriate, reduce the number of ADSs to reflect the sale and pay to you any proceeds, or send to you any property, remaining after it has paid the

## Reclassifications, Recapitalizations and Mergers

If we: \_\_\_\_\_\_ Then:

- Change the nominal or par value of our shares
- Reclassify, split up or consolidate any of the deposited securities
- Distribute securities on the shares that are not distributed to you
- Recapitalize, reorganize, merge, liquidate, sell all or substantially all of our assets, or take any similar action

The cash, shares or other securities received for the account of the respective depositary will become deposited securities. Each ADS will automatically represent its equal share of the new deposited securities.

The respective depositary may distribute some or all of the securities it received. It may also deliver new ADRs or ask you to surrender your outstanding ADRs in exchange for new ADRs identifying the new deposited securities.

#### **Amendment and Termination**

#### How may the deposit agreement be amended?

We may agree with the respective depositary to amend the respective deposit agreement and the ADRs without your consent for any reason. If an amendment adds or increases fees or charges, except for taxes and other governmental charges or expenses of the respective depositary for registration fees, facsimile costs, delivery charges or similar items, or prejudices a substantial right of ADR holders, it will not become effective for outstanding ADRs until 30 days after the respective depositary notifies ADR holders of the amendment. At the time an amendment becomes effective, you are considered, by continuing to hold your ADRs, to agree to the amendment and to be bound by the ADRs and the deposit agreement as amended.

#### How may the deposit agreement be terminated?

The respective depositary will terminate the respective deposit agreement if we ask it to do so. The respective depositary may also terminate the respective deposit agreement if it has told us that it would like to resign and we have not appointed a new depositary bank within 60 days. In either case, the respective depositary must notify you at least 30 days before termination.

After termination, the respective depositary and its agents will do the following under the respective deposit agreement but nothing else: (1) advise you that the deposit agreement is terminated, (2) collect distributions on the deposited securities (3) sell rights and other property, and (4) deliver shares and other deposited securities upon surrender of ADRs. Six months or more after termination, the respective depositary may sell any remaining deposited securities by public or private sale. After that, the depositary will hold the money it received on the sale, as well as any other cash it is holding under the deposit agreement for the *pro rata* benefit of the ADR holders of that class that have not surrendered their ADRs. It will invest the money in direct obligations of the federal government of the U.S. and has no liability for interest. The respective depositary's only obligations will be to account for the money and other cash. After termination our only obligations will be to indemnify the respective depositary and to pay fees and expenses of the depositary that we agreed to pay.

## Limitations on Obligations and Liability

## Limits on our Obligations and the Obligations of the Depositaries; Limits on Liability to Holders of ADRs

The deposit agreements expressly limit our obligations and the obligations of the depositaries. They also limit our liability and the liability of the depositaries. We and the depositaries:

- are only obligated to take the actions specifically set forth in the deposit agreements without negligence or bad faith;
- are not liable if any of us is prevented or delayed by law or circumstances beyond our control from performing our obligations under the deposit
  agreements;
- are not liable if any of us exercises discretion permitted under the deposit agreements;
- have no obligation to become involved in a lawsuit or other proceeding related to the ADRs or the deposit agreements on your behalf or on behalf of any other person;
- · may rely upon any documents we believe in good faith to be genuine and to have been signed or presented by the proper party; and
- are not liable for the depositaries' or any of their agents' reliance upon the authority of any information in, or for the depositaries' or any of their agents' compliance with directions from, any DTC participants in connection with the Direct Registration System.

By holding an ADR or an interest therein you will be agreeing that the respective depositary and its agents may fully respond to any and all demands or requests for information maintained by or on its behalf to the extent such information is requested or required by or pursuant to any lawful authority, including without limitation laws, rules, regulations, administrative or judicial process, banking, securities or other regulators.

Neither we nor the respective depositary nor any of our or their respective agents shall be liable to registered or other holders of ADSs or any other third party or parties for any indirect, special, punitive or consequential damages.

In the deposit agreements, we agree to indemnify the depositaries for acting as depositary, except for losses caused by the depositary's own negligence or bad faith, and the depositaries agree to indemnify us for losses resulting from their negligence or bad faith and in connection with pre-released ADRs.

#### **Requirements for Depositary Actions**

Before the depositaries will deliver or register a transfer of an ADR, make a distribution on an ADR, or permit withdrawal of shares or other property, the respective depositary may require:

• payment of stock transfer or other taxes or other governmental charges and transfer or registration fees charged by third parties for the transfer of any shares or other deposited securities;

- satisfactory proof of the identity and genuineness of any signature or other information it deems necessary; and
- compliance with regulations it may establish, from time to time, consistent with the respective deposit agreement, including presentation of transfer documents.

The respective depositary may refuse to deliver ADRs or register transfers of ADRs generally when the transfer books of the respective depositary or our transfer books are closed or at any time if the respective depositary or we think it advisable to do so.

## Your Right to Receive the Ordinary Shares Underlying your ADRs

You have the right to cancel your ADRs and withdraw the underlying shares or have shares credited to an account with Euroclear Nederland or an Admitted Institution at any time except:

- When temporary delays arise because: (i) the respective depositary has closed its transfer books or we have closed our transfer books; (ii) the transfer of shares is blocked to permit voting at a shareholders' meeting; or (iii) we are paying a dividend on our shares.
- · When you or other ADR holders seeking to withdraw shares owe money to pay fees, taxes and similar charges.
- When it is necessary to prohibit withdrawals in order to comply with any laws or governmental regulations that apply to relevant class of ADRs or to the withdrawal of shares or other deposited securities.

This right of withdrawal may not be limited by any other provision of the deposit agreements.

#### Pre-release of ADRs

The deposit agreements permit the depositaries to deliver ADRs before deposit of the underlying shares. This is called a pre-release of the ADRs. Subject to the terms and conditions of the deposit agreements, the pre-release of ADRs may occur only if (i) pre-released ADRs are fully collateralized (marked to market daily) with cash or U.S. government securities in an amount equal to not less than 100% of the market value of the pre-released ADRs held by the respective depositary for the benefit of owners of the applicable shares (but such collateral shall not constitute deposited securities), (ii) each recipient of pre-released ADRs agrees in writing with the respective depositary that such recipient (a) owns such shares, (b) assigns all beneficial right, title and interest therein to the respective depositary, (c) holds such shares for the account of the respective depositary and (d) will deliver such shares to the respective custodian as soon as practicable and promptly upon demand therefor and (iii) all pre-released ADRs evidence not more than 20% of all ADRs (excluding those evidenced by pre-released ADRs) or

such other percentage as we and the respective depositary may from time to time agree in writing, of the total number of shares represented by ADRs except to the extent, if any, that such limitation is exceeded solely because of the withdrawal of ADSs subsequent to the execution and delivery of pre-released ADRs in compliance with such limitation. The U.S. Treasury has expressed concerns regarding certain transactions involving the pre-release of ADRs.

#### Arbitration

Under the deposit agreements, each holder of ADSs is bound by the arbitration and exclusive jurisdiction provisions of our articles of association as if the applicable ADS holder was our shareholder. For a description of the arbitration and exclusive jurisdiction provisions of our articles of association see "Description of Royal Dutch Shell Ordinary Shares — Disputes between a shareholder or ADR holder and Royal Dutch Shell, any subsidiary, director or professional service provider".

# **Table of Contents**

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

ROYAL DUTCH SHELL PLC, (Registrant)

by <u>/s/ Jeroen van der Veer</u>

Name: Jeroen van der Veer Title: Chief Executive

by /s/ Peter Voser

Name: Peter Voser Title: CFO

Date: July 20, 2005

# **Table of Contents**

# **EXHIBITS**

| Exhibit No. | Description  |
|-------------|--|
| 99.1        | Press Announcement of Royal Dutch Shell, Royal Dutch and Shell Transport dated July 20, 2005                                       |
| 99.2        | Deposit Agreement among Royal Dutch Shell, JPMorgan Chase Bank, N.A., and Owners and Beneficial Owners of Class A American         |
|             | Depositary Receipts.   |
| 99.3        | Form of Class A American Depositary Receipts representing Royal Dutch Shell Class A American Depositary Shares each evidencing the |
|             | right to receive two Class A Shares of Royal Dutch Shell (included as Exhibit A to Exhibit 99.2 hereof).                           |
| 99.4        | Deposit Agreement among Royal Dutch Shell, The Bank of New York, and Owners and Beneficial Owners of Class B American Depositary   |
|             | Receipts.  |
| 99.5        | Form of Class B American Depositary Receipts representing Royal Dutch Shell Class B American Depositary Shares each evidencing the |
|             | right to receive two Class B Shares of Royal Dutch Shell (included as Exhibit A to Exhibit 99.4 herein).                           |

### NOT FOR RELEASE, PUBLICATION OR DISTRIBUTION IN WHOLE OR IN PART INTO ITALY OR JAPAN

20 July 2005

Joint Announcement by Royal Dutch Shell plc ('Royal Dutch Shell'), N.V. Koninklijke Nederlandsche Petroleum Maatschappij ('Royal Dutch') and The "Shell" Transport and Trading Company, p.l.c. ('Shell Transport')

### **Royal Dutch Shell unification completed**

All conditions to the public exchange offer for Royal Dutch (the 'Royal Dutch Offer') and the scheme of arrangement of Shell Transport (the 'Scheme') have now been satisfied or waived and the unification will proceed with trading in Royal Dutch Shell shares and ADRs beginning today.

The order of the High Court of Justice in England and Wales sanctioning the Scheme was registered by the Registrar of Companies in England and Wales this morning.

As announced yesterday, at the close of the acceptance period of the Royal Dutch Offer, Royal Dutch shareholders had tendered 1,897,638,608 ordinary shares in the share capital of Royal Dutch ('Royal Dutch Shares'), representing 91.69% of all issued and outstanding Royal Dutch Shares. Royal Dutch Shell exercised its right to waive down the minimum percentage of Royal Dutch Shares that must have been submitted for exchange from 95% to 75% (as announced on 8 July 2005).

Jeroen van der Veer, Chief Executive, said today: "We are now one company. The shares will trade today. Royal Dutch Shell moves forward."

### **Subsequent Offer Acceptance Period**

As announced yesterday, Royal Dutch Shell advises that a subsequent offer acceptance period will begin today, 20 July 2005 and will expire at 3:00 p.m. Amsterdam time on 9 August 2005 (the 'Subsequent Offer Acceptance Period'). During this period, holders of Royal Dutch Shares will be permitted – subject to the terms of the Royal Dutch Offer documents – to tender any remaining Royal Dutch Shares.

The Subsequent Offer Acceptance Period is not an extension of the Royal Dutch Offer and it will not affect the timing of the acceptance and delivery of Royal Dutch Shares previously tendered and accepted for exchange in the Royal Dutch Offer. During the Subsequent Offer Acceptance Period there will be no withdrawal rights either for Royal Dutch Shares tendered for exchange in the Royal Dutch Offer or for those tendered during the subsequent offer acceptance period. Royal Dutch Shell will promptly accept for exchange any Royal Dutch Shares tendered during the Subsequent Offer Acceptance Period at the same exchange ratio as in the Royal Dutch Offer period (that is, two Royal Dutch Shell Class A ordinary shares for every Royal Dutch Share held in bearer or Hague registry form tendered and one Royal Dutch Shell Class A ADR (representing two Royal Dutch Shell Class A Shares) for every Royal Dutch Share held in New York registry form tendered).

### **Remaining Royal Dutch Shares**

Royal Dutch Shell reserves the right to use any legally permitted method to obtain 100%. of the Royal Dutch Shares. This could include a squeeze out procedure, engaging in one or more corporate restructuring transactions, such as a merger, liquidation, transfer of assets or conversion of Royal Dutch into another form or corporate entity, or changing the Royal Dutch articles of association to alter the corporate or capital structure in a manner beneficial to Royal Dutch Shell. Further, Royal Dutch Shell could engage in one or more transactions with minority holders of Royal Dutch Shares which may include public or private exchanges, tender offers or purchases for consideration consisting of Royal Dutch Shell Shares, other securities or cash.

#### Settlement

Delivery of the Royal Dutch Shell "A" shares and ADRs to those holders of Royal Dutch Shares who accepted the Royal Dutch Offer prior to its expiry is expected to occur no later than 25 July 2005. However, as per usual practice, trading in Royal Dutch Shell "A" shares and ADRs can begin today.

The Royal Dutch Shell "B" shares and ADRs issued pursuant to the Scheme will be credited to shareholder accounts today and trading in Royal Dutch Shell "B" shares and ADRs can begin today.

Useful information on the Royal Dutch Shell "A" and "B" Shares can be found in the table below:

Royal Dutch Shell "B" shares Royal Dutch Shell "A" shares London Stock Exchange ticker code **RDSA RDSB** Euronext Amsterdam ticker code **RDSA RDSB** GB00B03MLX29 GB00B03MM408 **ISIN CUSIP** G7690A100 G7690A118 SEDOL - London B03MLX2 B03MM40 SEDOL - Euronext B09CBL4 B09CBN6

Useful information on the Royal Dutch Shell "A" and "B" ADRs can be found in the table below:

Royal Dutch Shell "A" ADRs
New York Stock Exchange ticker code
ISIN
CUSIP
SEDOL

Royal Dutch Shell "B" ADRs
Royal Dutch Shell "B" ADRs
RDS.B
US7802592060
US7802591070
US780259206
780259206
780259107
B03MM62
B03MM73

Further details on the manner of settlement of Royal Dutch Shell shares can be found in the Royal Dutch Offer and Scheme documents dated 19 May 2005.

This announcement is made pursuant to article 9b paragraph 1 of the 1995 Decree on the Supervision of Securities Trade.

# **Enguiries:**

# Media

UK/USA/International:

Stuart Bruseth +44 20 7934 6238 Andy Corrigan +44 20 7934 5963 

 Simon Buerk
 +44 20 7934 3453

 Bianca Ruakere
 +44 20 7934 4323

 Lisa Givert
 +44 20 7934 2914

 Susan Shannon
 +44 20 7934 3277

 Bernadette Cunnane
 +44 20 7934 2713

**Netherlands:** 

Herman Kievits +31 70 377 8750

### Institutional Investors

UK:

David Lawrence +44 20 7934 3855 Gerard Paulides +44 20 7934 6287

Europe:

Bart van der Steenstraten +33

+31 70 377 3996

USA:

Harold Hatchett +1 212 218 3112

### **Private Investor Call Centres**

UK Call Centre: Freephone 0800 169 1679 (+44 1903 276323 from outside the UK) Monday to Friday 8:30 a.m. to 5:30 p.m., London time

#### **US Call Centres:**

Holders of Royal Dutch New York Registered Shares: Within the US call toll free (877) 278 4235, outside of the US call +1 212 440 9800 Monday to Friday 9.00 a.m. to 11.00 p.m., New York time

*Holders of Shell Transport ADRs:* Within the US call toll free (877) 278 6357, outside of the US call +1 212 440 9800 Monday to Friday 9.00 a.m. to 11.00 p.m., New York time

## Legal notices

Holders of ordinary shares of Royal Dutch are urged to carefully review the registration statement on Form F-4 (including the prospectus) and other documents relating to the Royal Dutch Offer that have been filed with, or furnished to, the US Securities and Exchange Commission (the 'SEC') by Royal Dutch Shell and the related solicitation/recommendation statement on Schedule 14D-9 that has been filed with the SEC by Royal Dutch and, as the case may be, the Royal Dutch Offer document and the prospectus which have been filed with, or furnished to, Euronext Amsterdam N.V. and the Dutch Authority for the Financial Markets by Royal Dutch Shell, regarding the Royal Dutch Offer, because each of these documents will contain important information relating to the Royal Dutch Offer. A free copy of any of these documents filed with the SEC (once filed) and other documents made public or filed with, or furnished to, the SEC by Royal Dutch Shell, Royal Dutch and Shell Transport can be obtained at the SEC's website at www.sec.gov. These documents may also be obtained free of charge at the Royal Dutch Shell website at www.shell.com/unification or free of charge by contacting Investor Relations, Shell International B.V., FSK, PO Box 162, 2501 AN The Hague, The Netherlands.

The Royal Dutch Offer and the Scheme are not being and will not be made, directly or indirectly, in or into or by the use of the mails or any other means or instrumentality (including, without limitation, facsimile transmission, telex, telephone or internet) of interstate or foreign commerce of, or any such facilities of a national securities exchange of, Japan, and are not and will not be capable of acceptance by any such use, means, instrumentality or facilities from or within Japan. The Royal Dutch Offer and the Scheme are not being made to residents of Japan or in Japan. This announcement and other documents related to the transaction must not be electronically provided to, nor accessed by, residents of Japan or persons who are in Japan. Copies of this announcement any other documents related to the transaction are not being, and must not be, mailed or otherwise distributed or sent to any person or company in or from Japan. Persons receiving this announcement (including custodians, nominees and trustees) or other documents related to the transaction must not distribute or send them to any person or company in or from Japan.

The Royal Dutch Offer and the Scheme have not been notified to the Commissione Nazionale per le Società e la Borsa pursuant to applicable Italian securities laws and implementing regulations. Absent such notification, no public offer can be carried out in the Republic of Italy. Consequently, this announcement and other documents relating to the transaction have not been, and cannot bee, disclosed to any Italian residents or person or entity in the Republic of Italy and no other form of solicitation has been and can be, carried out in the Republic of Italy. This announcement and any document relating to the transaction may not be mailed, distributed, disseminated or otherwise disclosed to any Italian residents or persons or entities in the Republic of Italy.

The Royal Dutch Offer is not to be made in New Zealand and may not be accepted by persons in New Zealand except as set out below. No prospectus has been registered with the New Zealand Registrar of Companies in accordance with the Securities Act 1978 (New Zealand) (the 'New Zealand Securities Act'). Accordingly, neither the Royal Dutch Offer documents nor any other offering materials or advertisement in relation to the Royal Dutch Offer may be received by a person in New Zealand nor may Royal Dutch Shell shares be offered directly or indirectly in New Zealand except in circumstances where there is no contravention of the New Zealand Securities Act (or any statutory modification or re-enactment of, or statutory substitution for, the New Zealand Securities Act).

The distribution of this announcement in jurisdictions other than The Netherlands, England or the US may be affected by the laws of the relevant jurisdiction. Interested persons should inform themselves about and observe all applicable requirements. Any failure to comply with these restrictions may constitute a violation of the securities laws of any such jurisdiction.

All investment is subject to risk. The value of the Royal Dutch Shell shares may go down as well as up. Past performance is no guarantee for future returns. Investors are advised to seek expert financial advice before making any decisions as regards the proposals, including the Royal Dutch Offer referred to in this announcement.

# ROYAL DUTCH SHELL plc

and

# JPMORGAN CHASE BANK, N.A.

As Depositary

and

## OWNERS AND BENEFICIAL OWNERS OF AMERICAN DEPOSITARY RECEIPTS

**Deposit Agreement** 

Dated as of May 19, 2005

# TABLE OF CONTENTS

|                                   |   | Page     |
|-----------------------------------|---|----------|
| ARTICLE 1.                        | DEFINITIONS   | rage 1   |
| SECTION<br>1.01.<br>SECTION       | American Depositary Shares.   | 1        |
| 1.02.<br>SECTION                  | Beneficial Owner.   | 1        |
| 1.03.<br>SECTION                  | Commission.   | 2        |
| 1.04.<br>SECTION                  | Company.  | 2        |
| 1.05.<br>SECTION                  | Custodian.  | 2        |
| 1.06.<br>SECTION                  | Deliver; Surrender.   | 2        |
| 1.07.<br>SECTION                  | Deposit Agreement.  | 2        |
| 1.08.<br>SECTION                  | Depositary; Principal Office.   | 2        |
| 1.09.<br>SECTION                  | Deposited Securities.   | 3        |
| 1.10.<br>SECTION 1.11.<br>SECTION | Direct Registration System. Dollars; GBP.   | 3        |
| 1.12.<br>SECTION                  | Foreign Registrar.  | 3        |
| 1.13.<br>SECTION                  | Owner.  | 3        |
| 1.14.<br>SECTION                  | Receipts.   | 4        |
| 1.15.<br>SECTION                  | Registrar.  | 4        |
| 1.16.<br>SECTION                  | Restricted Securities.  | 4        |
| 1.17.<br>SECTION                  | Securities Act of 1933.   | 4        |
| 1.18.<br>ARTICLE 2.               | Shares. FORM OF RECEIPTS, DEPOSIT OF SHARES, EXECUTION AND DELIVERY, TRANSFER AND SURRENDER OF RECEIPTS | 4<br>5   |
| SECTION                           |   |          |
| 2.01.<br>SECTION                  | Form and Transferability of Receipts.   | 5        |
| 2.02.<br>SECTION                  | Deposit of Shares.  | 5        |
| 2.03.<br>SECTION                  | Execution and Delivery of Receipts.   | 6        |
| 2.04.<br>SECTION                  | Registration of Transfer of Receipts; Combination and Split-up of Receipts.                             | 7        |
| 2.05.<br>SECTION                  | Surrender of Receipts and Withdrawal of Shares.   | 8        |
| 2.06.<br>SECTION                  | Limitations on Execution and Delivery, Transfer and Surrender of Receipts.                              | 9        |
| 2.07.<br>SECTION                  | Lost Receipts, etc.   | 10       |
| 2.08.<br>SECTION                  | Cancellation and Destruction of Surrendered Receipts.   | 10       |
| 2.09.<br>SECTION                  | Pre-Release of Receipts.  | 10       |
| 2.10.<br>ARTICLE 3.<br>SECTION    | Maintenance of Records. CERTAIN OBLIGATIONS OF OWNERS AND BENEFICIAL OWNERS OF RECEIPTS                 | 11<br>11 |
| 3.01.<br>SECTION                  | Filing Proofs, Certificates and Other Information.  | 11       |
| 3.02.<br>SECTION                  | Liability of Owner or Beneficial Owner for Taxes.   | 11       |
| 3.03.<br>SECTION                  | Warranties on Deposit of Shares.  | 12       |
| 3.04.<br>ARTICLE 4.               | Disclosure of Interests. THE DEPOSITED SECURITIES   | 12<br>12 |

12

SECTION

Cash Distributions.

| 4.01.<br>SECTION |  |    |
|------------------|--|----|
| 4.02.            | Distributions Other Than Cash, Shares or Rights. | 13 |
| SECTION          | Distributions Other Than Cash, Shares of Rights. | 15 |
| 4.03.            | Distributions in Shares.                         | 14 |
| SECTION          |  |    |
| 4.04.            | Rights.  | 14 |
| SECTION          |  |    |
| 4.05.            | Conversion of Foreign Currency.                  | 16 |
| SECTION 4.06.    | Fixing of Record Date.                           | 16 |
|                  |  |    |

|                  |  | Page |
|------------------|--|------|
| SECTION          | Value of Described Committee   | 17   |
| 4.07.<br>SECTION | Voting of Deposited Securities.                                      | 17   |
| 4.08.            | Changes Affecting Deposited Securities.                              | 18   |
| SECTION          | Changes Affecting Deposited Securities.                              | 10   |
| 4.09.            | Reports.   | 18   |
| SECTION          | reports.   | 10   |
| 4.10.            | Lists of Owners.   | 19   |
| SECTION 4.11.    | Withholding.   | 19   |
| ARTICLE 5.       | THE DEPOSITARY, THE CUSTODIANS AND THE COMPANY                       | 19   |
| SECTION          |  |      |
| 5.01.            | Maintenance of Office and Transfer Books by the Depositary.          | 19   |
| SECTION          |  |      |
| 5.02.            | Prevention or Delay in Performance by the Depositary or the Company. | 20   |
| SECTION          |  |      |
| 5.03.            | Obligations of the Depositary, the Custodian and the Company.        | 20   |
| SECTION          |  |      |
| 5.04.            | Resignation and Removal of the Depositary.                           | 22   |
| SECTION          |  |      |
| 5.05.            | The Custodians.  | 22   |
| SECTION          |  |      |
| 5.06.            | Notices and Reports.   | 23   |
| SECTION          | Distribution of Additional Chance Districts                          | 22   |
| 5.07.<br>SECTION | Distribution of Additional Shares, Rights, etc.                      | 23   |
| 5.08.            | Indemnification.   | 24   |
| SECTION          | muemmineauon.  | 24   |
| 5.09.            | Charges of Depositary.   | 25   |
| SECTION          | ominges of 2 epositing)  | _5   |
| 5.10.            | Retention of Depositary Documents.                                   | 25   |
| SECTION 5.11.    | Exclusivity.   | 26   |
| SECTION          |  |      |
| 5.12.            | List of Restricted Securities Owners.                                | 26   |
| ARTICLE 6.       | AMENDMENT AND TERMINATION  | 26   |
| SECTION 6.01.    | Amendment.   | 26   |
| SECTION          |  |      |
| 6.02.            | Termination.   | 26   |
| ARTICLE 7.       | MISCELLANEOUS  | 27   |
| SECTION          |  | 27   |
| 7.01.<br>SECTION | Counterparts.  | 27   |
| 7.02.            | No Third Party Beneficiaries.  | 28   |
| SECTION          | 110 Tilliu Party Belleficiaries.                                     | 20   |
| 7.03.            | Severability.  | 28   |
| SECTION          | ocveraomey.  | 20   |
| 7.04.            | Owners and Beneficial Owners as Parties; Binding Effect.             | 28   |
| SECTION          | o where and Denement of where as I takes, Dinamy Direct              |      |
| 7.05.            | Notices.   | 28   |
| SECTION          |  |      |
| 7.06.            | Compliance with U.S. Securities Laws.                                | 29   |
| SECTION          |  |      |
| 7.07.            | Governing Law.   | 29   |
|                  |  |      |

Form of Receipt

### DEPOSIT AGREEMENT

**DEPOSIT AGREEMENT** dated as of May 19, 2005, among ROYAL DUTCH SHELL plc, incorporated under the laws of England and Wales (herein called the Company), JPMORGAN CHASE BANK, N.A., a national banking association organized under the laws of the United States (herein called the Depositary), and all Owners and Beneficial Owners from time to time of American Depositary Receipts issued hereunder.

### WITNESSETH:

**WHEREAS**, the Company desires to provide, as hereinafter set forth in this Deposit Agreement, for the deposit of Shares (as hereinafter defined) of the Company from time to time with the Depositary or with the Custodian (as hereinafter defined) as agent of the Depositary for the purposes set forth in this Deposit Agreement, for the creation of American Depositary Shares representing the Shares so deposited and for the execution and delivery of American Depositary Receipts evidencing the American Depositary Shares; and

**WHEREAS**, the American Depositary Receipts are to be substantially in the form of Exhibit A annexed hereto, with appropriate insertions, modifications and omissions, as hereinafter provided in this Deposit Agreement;

**NOW**, **THEREFORE**, in consideration of the premises, it is agreed by and among the parties hereto as follows:

## **ARTICLE 1. DEFINITIONS**

The following definitions shall for all purposes, unless otherwise clearly indicated, apply to the respective terms used in this Deposit Agreement:

SECTION 1.01. <u>American Depositary Shares.</u>

The term "American Depositary Shares" shall mean the securities representing the interests in the Deposited Securities and evidenced by the Receipts issued hereunder. Each American Depositary Share shall represent the number of Shares specified in Exhibit A annexed hereto, until there shall occur a distribution upon Deposited Securities covered by Section 4.03 or a change in Deposited Securities covered by Section 4.08 with respect to which additional Receipts are not executed and delivered, and thereafter American Depositary Shares shall represent the amount of Shares or Deposited Securities specified in such Sections.

### SECTION 1.02. Beneficial Owner.

The term "Beneficial Owner" shall mean each person owning from time to time any beneficial interest in the American Depositary Shares evidenced by any Receipt.

### SECTION 1.03. Commission.

The term "Commission" shall mean the Securities and Exchange Commission of the United States or any successor governmental agency in the United States.

## SECTION 1.04. Company.

The term "Company" shall mean Royal Dutch Shell plc, incorporated under the laws of England and Wales, and its successors.

### SECTION 1.05. Custodian.

The term "Custodian" shall mean the principal Amsterdam office of ING Groep NV, as agent of the Depositary for the purposes of this Deposit Agreement, and any other firm or corporation which may hereafter be appointed by the Depositary pursuant to the terms of Section 5.05, as substitute or additional custodian or custodians hereunder, as the context shall require and shall also mean all of them collectively.

### SECTION 1.06. Deliver; Surrender.

- (a) The term "deliver", or its noun form, when used with respect to Shares shall mean (i) one or more book-entry transfers to an account or accounts maintained with a depository institution authorized under the laws of England and Wales to effect book-entry transfers of such securities or (ii) the physical transfer of certificates representing Shares.
- (b) The term "deliver", or its noun form, when used with respect to Receipts, shall mean (i) one or more book-entry transfers of American Depositary Shares on the Direct Registration System or to an account or accounts at The Depository Trust Company ("DTC") designated by the person entitled to such delivery or (ii) if requested by the person entitled to such delivery, to delivery at the Principal Office of the Depositary of one or more Receipts.
- (c) The term "surrender", when used with respect to Receipts, shall mean (i) one or more book-entry transfers of American Depositary Shares on the Direct Registration System or to the DTC account of the Depositary or (ii) surrender to the Depositary at its Principal Office of one or more Receipts.

### SECTION 1.07. Deposit Agreement.

The term "Deposit Agreement" shall mean this Deposit Agreement, as the same may be amended from time to time in accordance with the provisions of this Deposit Agreement.

## SECTION 1.08. <u>Depositary; Principal Office.</u>

The term "Depositary" shall mean JPMorgan Chase Bank, N.A., a national banking association organized under the laws of the United State, and any successor as depositary hereunder. The term "Principal Office", when used with respect to the Depositary,

shall mean the office of the Depositary which at the date of this Agreement is Four New York Plaza, New York, New York 10004.

### SECTION 1.09. Deposited Securities.

The term "Deposited Securities" as of any time shall mean Shares at such time deposited or deemed to be deposited under this Deposit Agreement and any and all other securities, property and cash received by the Depositary or the Custodian in respect thereof and at such time held hereunder, subject as to cash to the provisions of Section 4.05.

### SECTION 1.10. <u>Direct Registration System.</u>

The term "Direct Registration System" means the system for the uncertificated registration of ownership of securities established by DTC and utilized by the Depositary pursuant to which the Depositary may record the ownership of Receipts without the issuance of a certificate, which ownership shall be evidenced by periodic statements issued by the Depositary to the Holders entitled thereto. For purposes hereof, the Direct Registration System shall include access to the Profile Modification System maintained by DTC which provides for automated transfer of ownership between DTC and the Depositary ("DRS/Profile").

## SECTION 1.11. Dollars; GBP; Euro.

The term "Dollars" shall mean United States dollars. The term "GBP" shall mean United Kingdom pounds. The term "euro" shall mean the single currency of the participating Member States in the Third Stage of European Economic and Monetary Union of the Treaty Establishing the European Community, as amended from time to time.

## SECTION 1.12. Foreign Registrar.

The term "Foreign Registrar" shall mean the entity that presently carries out the duties of registrar for the Shares or any successor as registrar for the Shares and any other appointed agent of the Company for the transfer and registration of Shares.

### SECTION 1.13. Owner.

The term "Owner" shall mean the person in whose name a Receipt is registered on the books of the Depositary maintained for such purpose.

## SECTION 1.14. Receipts; Direct Registration Receipts.

The term "Receipts" shall mean the American Depositary Receipts issued hereunder, including Pre-Released Receipts, evidencing American Depositary Shares. The term "Direct Registration Receipts" shall mean book entry notations recorded on the Direct Registration System which are evidenced by periodic statements from the Depositary. References to "Receipts" shall include Direct Registration Receipts, unless the context otherwise requires.

## SECTION 1.15. Registrar.

The term "Registrar" shall mean any bank or trust company having an office in the Borough of Manhattan, The City of New York, which shall be appointed, after consultation with the Company, to register Receipts and transfers of Receipts as herein provided.

### SECTION 1.16. Restricted Securities.

The term "Restricted Securities" shall mean Shares, or Receipts representing such Shares, which are acquired directly or indirectly from the Company, or any affiliate (as defined in Rule 144 to the Securities Act of 1933) of the Company, in a transaction or chain of transactions not involving any public offering, or which are held by an officer, director (or persons performing similar functions) or other affiliate of the Company, or which would require registration under the Securities Act of 1933 in connection with the public offer and sale thereof in the United States, or which are subject to other restrictions on sale or deposit under the laws of the United States or England and Wales, or under a shareholder agreement or the Memorandum or Articles of Association of the Company.

## SECTION 1.17. Securities Act of 1933.

The term "Securities Act of 1933" shall mean the United States Securities Act of 1933, as from time to time amended.

### SECTION 1.18. Shares.

The term "Shares" shall mean Class A ordinary shares in registered form of the Company heretofore validly issued and outstanding and fully paid, nonassessable and that were not issued in violation of any pre-emptive or similar rights of the holders of outstanding Shares or hereafter validly issued and outstanding and fully paid, nonassessable and that are not issued in violation of any pre-emptive or similar rights of the holders of outstanding Shares or interim certificates representing such Shares; provided, however, that, if there shall occur any change in nominal value, a split-up or consolidation or any other reclassification or, upon the occurrence of an event described in Section 4.08, an exchange or conversion in respect of the Shares of the Company, the term "Shares" shall thereafter also mean the successor securities resulting from such change in nominal value, split-up or consolidation or such other reclassification or such exchange or conversion. Shares may be certificated or uncertificated.

## ARTICLE 2. FORM OF RECEIPTS, DEPOSIT OF SHARES, EXECUTION AND DELIVERY, TRANSFER AND SURRENDER OF RECEIPTS

SECTION 2.01. Form and Transferability of Receipts.

Definitive Receipts shall be substantially in the form set forth in Exhibit A annexed to this Deposit Agreement, with appropriate insertions, modifications and omissions, as hereinafter provided. No Receipt shall be entitled to any benefits under this Deposit Agreement or be valid or obligatory for any purpose, unless such Receipt shall have been either issued through the Direct Registration System or executed by the Depositary by the manual signature of a duly authorized signatory of the Depositary; provided, however, that such signature may be a facsimile if a Registrar for the Receipts shall have been appointed and such Receipts are countersigned by the manual signature of a duly authorized officer of the Registrar. The Depositary shall maintain books on which each Receipt so issued, executed and/or delivered as hereinafter provided and the transfer of each such Receipt shall be registered. Receipts bearing the manual or facsimile signature of a duly authorized signatory of the Depositary who was at any time a proper signatory of the Depositary shall bind the Depositary, notwithstanding that such signatory has ceased to hold such office prior to the execution and delivery of such Receipts by the Registrar or did not hold such office on the date of issuance of such Receipts.

The Receipts may be endorsed with or have incorporated in the text thereof such legends or recitals or modifications not inconsistent with the provisions of this Deposit Agreement as may be required by the Depositary or required to comply with any applicable law or regulations thereunder or with the rules and regulations of any securities exchange upon which American Depositary Shares may be listed or to conform with any usage with respect thereto, or to indicate any special limitations or restrictions to which any particular Receipts are subject by reason of the date of issuance of the underlying Deposited Securities or otherwise.

Notwithstanding anything in this Deposit Agreement or in the Receipts to the contrary, American Depositary Shares shall be evidenced by Direct Registration Receipts, unless certificated Receipts are specifically requested by the Owner. Owners shall be bound by the terms and conditions of this Deposit Agreement and of the form of Receipt, regardless of whether their Receipts are Direct Registration Receipts or certificated Receipts.

Title to a Receipt (and to the American Depositary Shares evidenced thereby), when properly endorsed or accompanied by proper instruments of transfer, shall be transferable by delivery with the same effect as in the case of a negotiable instrument under the laws of New York; <u>provided</u>, <u>however</u>, that the Depositary, notwithstanding any notice to the contrary, may treat the Owner thereof as the absolute owner thereof for the purpose of determining the person entitled to distribution of dividends or other distributions or to any notice provided for in this Deposit Agreement and for all other purposes.

## SECTION 2.02. <u>Deposit of Shares.</u>

Subject to the terms and conditions of this Deposit Agreement, Shares or evidence of rights to receive Shares may be deposited by delivery thereof to any Custodian hereunder,

accompanied by any appropriate instrument or instruments of transfer, or endorsement, in form satisfactory to the Custodian, together with all such certifications as may be required by the Depositary or the Custodian in accordance with the provisions of this Deposit Agreement, and, if the Depositary requires, together with a written order directing the Depositary to execute and deliver to, or upon the written order of, the person or persons stated in such order, a Receipt or Receipts for the number of American Depositary Shares representing such deposit. No Share shall be accepted for deposit unless accompanied by evidence satisfactory to the Depositary that any necessary approval, where relevant, has been granted by any governmental body in England and Wales that is then performing the function of the regulation of currency exchange. If required by the Depositary, Shares presented for deposit at any time, whether or not the transfer books of the Company or the Foreign Registrar, if applicable, are closed, shall also be accompanied by an agreement or assignment, or other instrument satisfactory to the Depositary, which will provide for the prompt transfer to the Custodian of any dividend, or right to subscribe for additional Shares or to receive other property which any person in whose name the Shares are or have been recorded may thereafter receive upon or in respect of such deposited Shares, or in lieu thereof, such agreement of indemnity or other agreement as shall be satisfactory to the Depositary.

In the case of certificated Shares, at the request and risk and expense of any person proposing to deposit Shares, and for the account of such person, the Depositary may receive certificates for Shares to be deposited, together with the other instruments herein specified, for the purpose of forwarding such Share certificates to the Custodian for deposit hereunder.

In the case of Certificated Shares, upon each delivery to a Custodian of a certificate or certificates for Shares to be deposited hereunder, together with the other documents above specified, such Custodian shall be required by the Depositary, as soon as transfer and recordation can be accomplished, to present such certificate or certificates to the Company or the Foreign Registrar, if applicable, for transfer and recordation of the Shares being deposited in the name of the Depositary or its nominee or such Custodian or its nominee. To the extent that the provisions of or governing the Shares make delivery of certificates therefor impracticable, Shares may be deposited hereunder by such delivery thereof as the Depositary or the Custodian may reasonably accept, including, without limitation, by causing them to be credited to an account maintained by the Custodian for such purpose with the Company or an accredited intermediary, such as a bank, acting as a registrar for the Shares, together with delivery of the documents, payments and other delivery instructions referred to herein to the Custodian or the Depositary.

Deposited Securities shall be held by the Depositary or by a Custodian for the account and to the order of the Depositary or at such other place or places as the Depositary shall determine. The Depositary shall provide written notice informing the Company of any such other place or places.

## SECTION 2.03. Execution and Delivery of Receipts.

Upon receipt by any Custodian of any deposit pursuant to Section 2.02 hereunder (and in addition, if the transfer books of the Company or the Foreign Registrar, if applicable, are

open, the Depositary may in its sole discretion require a proper acknowledgment or other evidence from the Company that any Deposited Securities have been recorded upon the books of the Company or the Foreign Registrar, if applicable, in the name of the Depositary or its nominee or such Custodian or its nominee), together with the other documents required as above specified, such Custodian shall be required by the Depositary to notify the Depositary of such deposit and the person or persons to whom or upon whose written order a Receipt or Receipts are deliverable in respect thereof and the number of American Depositary Shares to be evidenced thereby. Such notification shall be required to be made by letter or, at the request, risk and expense of the person making the deposit, by cable, telex or facsimile transmission. Upon receiving such notice from such Custodian, or upon the receipt of Shares by the Depositary, the Depositary, subject to the terms and conditions of this Deposit Agreement, shall execute and deliver at its Principal Office, to or upon the order of the person or persons entitled thereto, a Receipt or Receipts, registered in the name or names and evidencing any authorized number of American Depositary Shares requested by such person or persons, but only upon payment to the Depositary of the fees and expenses of the Depositary for the execution and delivery of such Receipt or Receipts as provided in Section 5.09, and of all taxes and governmental charges and fees payable in connection with such deposit and the transfer of the Deposited Securities.

## SECTION 2.04. Registration of Transfer of Receipts; Combination and Split-up of Receipts.

The Depositary, subject to the terms and conditions of this Deposit Agreement, shall without unreasonable delay, register transfers of Receipts on its transfer books, upon any surrender of a Receipt, by the Owner in person or by a duly authorized attorney, properly endorsed or accompanied by proper instruments of transfer, and duly stamped as may be required by the laws of the State of New York and of the United States of America. Thereupon the Depositary shall execute a new Receipt or Receipts and deliver the same to or upon the order of the person entitled thereto.

The Depositary, subject to the terms and conditions of this Deposit Agreement, shall upon surrender of a Receipt or Receipts for the purpose of effecting a split-up or combination of such Receipt or Receipts, execute and deliver a new Receipt or Receipts for any authorized number of American Depositary Shares requested, evidencing the same aggregate number of American Depositary Shares as the Receipt or Receipts surrendered. At the request of an Owner, the Depositary shall, for the purpose of substituting a certificated Receipt with a Direct Registration Receipt, or vice versa, execute and deliver a certificated Receipt or a Direct Registration Receipt, as the case may be, for any authorized number of American Depositary Shares requested, evidencing the same aggregate number of American Depositary Shares as those evidenced by the certificated Receipt or Direct Registration Receipt, as the case may be, substituted.

The Depositary may appoint one or more co-transfer agents for the purpose of effecting transfers, combinations and split-ups of Receipts at designated transfer offices on behalf of the Depositary. In carrying out its functions, a co-transfer agent may require evidence of authority and compliance with applicable laws and other requirements by Owners or persons

entitled to Receipts and will be entitled to protection and indemnity to the same extent as the Depositary.

### SECTION 2.05. Surrender of Receipts and Withdrawal of Shares.

Upon surrender at the Principal Office of the Depositary of a Receipt for the purpose of withdrawal of the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, and upon payment of the fee of the Depositary for the surrender of Receipts as provided in Section 5.09 and payment of all taxes and governmental charges payable in connection with such surrender and withdrawal of the Deposited Securities, and subject to the terms and conditions of this Deposit Agreement and English law, the Owner of such Receipt shall be entitled to delivery, to him or upon his order, of the amount of Deposited Securities at the time represented by the American Depositary Shares evidenced by such Receipt. Delivery of such Deposited Securities may be made by the delivery of (a) to the extent available, certificates in the name of such Owner or as ordered by him or certificates properly endorsed or accompanied by proper instruments of transfer to such Owner or as ordered by him, (b) Deposited Securities to an account designated by such Owner with the Euroclear Nederland (including any successors thereto, "Euroclear Nederland") or an institution that maintains accounts with the Euroclear Nederland and (c) any other securities, property and cash to which such Owner is then entitled in respect of such Receipts to such Owner or as ordered by him. Such delivery shall be made, as hereinafter provided, without unreasonable delay.

To the extent applicable, the Owner requesting withdrawal of Shares shall have the sole responsibility for ensuring that such Owner, or its customer, has a valid account with Euroclear Nederland or an institution that maintains accounts with Euroclear Nederland and that the information required for the bookentry transfer to such account is accurately and promptly provided to the Depositary.

A Receipt surrendered for such purposes may be required by the Depositary to be properly endorsed in blank (in the case of a certificated Receipt) or accompanied by proper instruments of transfer in blank, and if the Depositary so requires, the Owner thereof shall execute and deliver to the Depositary a written order directing the Depositary to cause the Deposited Securities being withdrawn to be delivered to or upon the written order of a person or persons designated in such order. Thereupon the Depositary shall direct the Custodian to deliver, in the case of Shares, as provided for in the first paragraph of this Section 2.05, and otherwise at the office of such Custodian, subject to Sections 2.06, 3.01 and 3.02 and to the other terms and conditions of this Deposit Agreement and English law, to or upon the written order of the person or persons designated in the order delivered to the Depositary as above provided, the amount of Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, except that the Depositary may make delivery to such person or persons at the Principal Office of the Depositary of any dividends or distributions with respect to the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, or of any proceeds of sale of any dividends, distributions or rights, which may at the time be held by the Depositary.

At the request, risk and expense of any Owner so surrendering a Receipt, and for the account of such Owner, the Depositary shall direct the Custodian to forward any cash or

other property (other than rights) comprising, and forward a certificate or certificates and other proper documents of title for, the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt to the Depositary for delivery at the Principal Office of the Depositary. Such direction shall be given by letter or, at the request, risk and expense of such Owner, by cable, telex or facsimile transmission. Rights, if any, shall be delivered to such Owner pursuant to Section 4.04.

## SECTION 2.06. <u>Limitations on Execution and Delivery, Transfer and Surrender of Receipts.</u>

As a condition precedent to the execution and delivery, registration of transfer, split-up, combination or surrender of any Receipt or withdrawal of any Deposited Securities, the Depositary, Custodian or Registrar may require payment from the depositor of Shares or the presenter of the Receipt of a sum sufficient to reimburse it for any tax or other governmental charge and any stock transfer or registration fee with respect thereto (including any such tax or charge and fee with respect to Shares being deposited or withdrawn) and payment of any applicable fees as herein provided, may require the production of proof satisfactory to it as to the identity and genuineness of any signature and may also require compliance with any regulations the Depositary may establish consistent with the provisions of this Deposit Agreement, including, without limitation, this Section 2.06.

The delivery of Receipts against deposit of Shares generally or against deposit of particular Shares may be suspended, or the transfer of Receipts in particular instances may be refused, or the registration of transfer of outstanding Receipts generally may be suspended, during any period when the transfer books of the Depositary are closed, or if any such action is deemed necessary or advisable by the Depositary or the Company at any time or from time to time because of any requirement of law or of any government or governmental body or commission, or under any provision of this Deposit Agreement, or for any other reason, subject to the provisions of Section 7.07 hereof. The Depositary shall not knowingly accept for deposit under this Deposit Agreement any Shares which would be required to be registered under the provisions of the Securities Act of 1933 for the public offer and sale thereof in the United States unless a registration statement is in effect as to such Shares for such offer and sale. The Depositary will use reasonable efforts to comply with written instructions of the Company that the Depositary shall not accept for the deposit hereunder any Shares identified in such circumstances as may reasonably be specified in such instructions to facilitate the Company's compliance with the U.S. securities laws.

Notwithstanding anything to the contrary in this Deposit Agreement, Owners shall be entitled to surrender Receipts and withdraw Deposited Securities as provided in Section 2.05 at any time, subject only to (i) temporary delays caused by closing the transfer books of the Depositary or the Company or the deposit of Shares in connection with voting at a shareholders' meeting, or the payment of dividends, (ii) the payment of fees, taxes and similar charges, (iii) compliance with any U.S. or foreign laws or governmental regulations relating to the Receipts or to the withdrawal of the Deposited Securities and (iv) any other reason that may at any time be specified in paragraph I.(A)(1) of the General Instructions to Form F-6 under the Securities Act of 1933, from time to time in effect, or any successor provision thereto.

## SECTION 2.07. Lost Receipts, etc.

In case any Receipt shall be mutilated, destroyed, lost or stolen, the Depositary shall issue a new Receipt through the Direct Registration system or execute and deliver a new Receipt of like tenor in exchange and substitution for such mutilated Receipt upon cancellation thereof, or in lieu of and in substitution for such destroyed, lost or stolen Receipt. Before the Depositary shall execute and deliver a new Receipt in substitution for a destroyed, lost or stolen Receipt, the Owner thereof shall have (a) filed with the Depositary (i) a request for such execution and delivery before the Depositary has notice that the Receipt has been acquired by a bona fide purchaser and (ii) a sufficient indemnity bond and (b) satisfied any other reasonable requirements imposed by the Depositary.

## SECTION 2.08. Cancellation and Destruction of Surrendered Receipts.

All Receipts surrendered to the Depositary shall be canceled by the Depositary. The Depositary is authorized to destroy Receipts so canceled subject to Section 2.10.

## SECTION 2.09. <u>Pre-Release of Receipts.</u>

The Depositary will lend neither the Shares held under this Deposit Agreement nor the Receipts. The Depositary reserves the right to execute and deliver Receipts prior to the receipt of Shares pursuant to Section 2.02 on the terms and conditions set forth below. The Depositary may receive Receipts in lieu of Shares as settlement of the pre-release of a Receipt. Subject to the terms and conditions of this Deposit Agreement, the Pre-Release of Receipts may occur only if (i) Pre-released Receipts are fully collateralized (marked to market daily) with cash or U.S. government securities in an amount equal to not less than 100% of the market value of the Pre-Released Receipts held by the Depositary for the benefit of Owners (but such collateral shall not constitute Deposited Securities), (ii) each recipient of Pre-released Receipts agrees in writing with the Depositary that such recipient (a) owns such Shares, (b) assigns all beneficial right, title and interest therein to the Depositary, (c) holds such Shares for the account of the Depositary and (d) will deliver such Shares to the Custodian as soon as practicable and promptly upon demand therefor and (iii) all Pre-released Receipts evidence not more than 20% of all American Depositary Shares (excluding those evidenced by Pre-released Receipts) or such other percentage as the Company and the Depositary may from time to time agree in writing, of the total number of Shares represented by Receipts except to the extent, if any, that such limitation is exceeded solely because of the withdrawal of Deposited Securities subsequent to the execution and delivery of Pre-Released Receipts in compliance with such limitation. The Depositary will also set limits with respect to the number of Receipts and Shares involved in transactions to be done hereunder with anyone person on a case by case basis as it deems appropriate.

The Depositary may retain for its own account any compensation received by it in connection with the foregoing.

### SECTION 2.10. Maintenance of Records.

The Depositary agrees to maintain or cause its agents to maintain records of all Receipts surrendered and Deposited Securities withdrawn under Section 2.05, substitute Receipts delivered under Section 2.07, and of cancelled or destroyed Receipts under Section 2.08, in keeping with procedures ordinarily followed by stock transfer agents located in the City of New York or as required by laws or regulations governing the Depositary. The Depositary shall provide full access to such records to the Company and its agents from time to time during normal business hours upon the reasonable written request of the Company.

### ARTICLE 3. CERTAIN OBLIGATIONS OF OWNERS AND BENEFICIAL OWNERS OF RECEIPTS

## SECTION 3.01. Filing Proofs, Certificates and Other Information.

Any person presenting Shares for deposit or any Owner or Beneficial Owner of a Receipt may be required from time to time to file with the Depositary or the Custodian such proof of citizenship or residence, exchange control approval, or such information relating to the registration on the books of the Company or the Foreign Registrar, if applicable, to execute such certificates and to make such representations and warranties, as the Depositary may deem necessary or proper. The Depositary may, and shall if requested by the Company pursuant to the provisions of Section 7.07 hereof, withhold the delivery or registration of transfer of any Receipt or the distribution of any dividend or sale or distribution of rights or of the proceeds thereof or the delivery of any Deposited Securities until such proof or other information is filed or such certificates are executed or such representations and warranties made. The Depositary shall provide copies thereof to the Company as promptly as practicable upon written request by the Company, to the extent that disclosure is permitted under applicable law.

## SECTION 3.02. <u>Liability of Owner or Beneficial Owner for Taxes.</u>

If any tax or other governmental charge shall become payable by the Custodian or the Depositary with respect to any Receipt or any Deposited Securities represented by any Receipt, such tax or other governmental charge shall be payable by the Owner or Beneficial Owner of such Receipt to the Depositary. The Depositary may refuse to effect any transfer of such Receipt or any withdrawal of Deposited Securities represented by American Depositary Shares evidenced by such Receipt until such payment is made, and may withhold any dividends or other distributions, or may sell for the account of the Owner or Beneficial Owner thereof any part or all of the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, and may apply such dividends or other distributions or the proceeds of any such sale in payment of such tax or other governmental charge and the Owner or Beneficial Owner of such Receipt shall remain liable for any deficiency. Neither the Company nor the Depositary shall be liable for failure of an Owner to comply with applicable tax laws or governmental charges.

### SECTION 3.03. Warranties on Deposit of Shares.

Every person depositing Shares under this Deposit Agreement shall be deemed thereby to represent and warrant that such Shares and each certificate therefor are validly issued, fully paid, nonassessable and were not issued in violation of any preemptive or similar rights of the holders of outstanding Shares and that the person making such deposit is duly authorized so to do. Every such person shall also be deemed to represent that such Shares are not, and American Depositary Shares representing such Shares would not be, Restricted Securities. Such representations and warranties shall survive the deposit of Shares and delivery of Receipts.

## SECTION 3.04. <u>Disclosure of Interests.</u>

To the extent that provisions of or governing any Deposited Securities (including the Company's Memorandum and Articles of Association or applicable English law) may require the disclosure of beneficial or other ownership of Deposited Securities, other Shares and other securities to the Company and may provide for blocking transfer and voting or other rights to enforce such disclosure or limit such ownership, the Depositary shall, to the extent reasonably practicable, comply with the Company's instructions as to Receipts in respect of any such enforcement or limitation, and Owners and Beneficial Owners of Receipts shall comply with all such disclosure requirements and ownership limitations and shall cooperate with the Depositary's compliance with such Company instructions. The Company may from time to time request Owners to provide information (a) as to the capacity in which such Owners own or owned American Depositary Shares, (b) regarding the identity of any other persons then or previously interested in such American Depositary Shares and (c) regarding the nature of such interest and various other matters pursuant to applicable law or the Memorandum and Articles of Association of the Company or other such corporate document of the Company, all as if such American Depositary Shares were to the extent practicable the underlying Shares. Each Owner agrees to provide any information requested by the Company or the Depositary pursuant to this Section whether or not such person is still an Owner at the time of the request. The Depositary agrees to use reasonable efforts to comply with written instructions received from the Company requesting that the Depositary forward any such requests to Owners and to forward to the Company any responses to such requests received by the Depositary.

### ARTICLE 4. THE DEPOSITED SECURITIES

### SECTION 4.01. Cash Distributions.

Notwithstanding any rights under the Company's articles of association, dividends paid on the Deposited Securities that are not paid to the Depositary or its nominee or Custodian in US Dollars will be paid by the Company in euro. Whenever the Depositary shall receive any cash dividend or other cash distribution in respect of any Deposited Securities, the Depositary shall, subject to the provisions of Section 4.05 in the case of a dividend and/or distribution received in a currency other than Dollars, convert the amounts so received into Dollars. Promptly after the settlement of such conversion or, in the in the case of any cash dividend or other cash distribution received by the Depositary in Dollars, the Depositary shall as promptly as practicable, distribute the amount thus

received (net of the fees and expenses of the Depositary as provided in Section 5.09) to the Owners entitled thereto, in proportion to the number of American Depositary Shares representing such Deposited Securities held by each of them; <u>provided, however</u>, that in the event that the Depositary shall be required to withhold and does withhold from such cash dividend or such other cash distribution an amount on account of taxes, the amount distributed to the Owner of the Receipts evidencing American Depositary Shares representing such Deposited Securities shall be reduced accordingly. The Depositary shall distribute only such amount, however, as can be distributed without attributing to any Owner a fraction of one cent. Any such fractional amounts shall be rounded down to the nearest whole cent and so distributed to Owners entitled thereto. The Company or its agent will remit to the appropriate governmental agency all amounts withheld and owing to such agency. The Depositary will forward to the Company or its agent such information from its records as the Company may reasonably request to enable the Company or its agent to file necessary reports with governmental agencies, and the Depositary or the Company or its agent may file any such reports necessary to obtain benefits under the applicable tax treaties for the Owners of Receipts.

Subject to the rules and regulations of any stock exchange upon which the American Depositary Shares may be traded, the Depositary shall endeavor to convert the funds as promptly as practicable and that distributions to Owners under this Section are made within five New York Stock Exchange trading days of the day on which the cash dividend or cash distribution on the Deposited Securities is received by the Depositary.

## SECTION 4.02. <u>Distributions Other Than Cash, Shares or Rights.</u>

Subject to the provisions of Sections 4.11 and 5.09, whenever the Depositary shall receive any distribution other than a distribution described in Section 4.01, 4.03 or 4.04, the Depositary shall cause the securities or property received by it to be distributed to the Owners entitled thereto, after deduction or upon payment of any fees and expenses of the Depositary or any taxes or other governmental charges, in proportion to the number of American Depositary Shares representing such Deposited Securities held by them respectively, in any manner that the Depositary may deem equitable and practicable for accomplishing such distribution; provided, however, that if in the reasonable opinion of the Depositary such distribution cannot be made proportionately among the Owners entitled thereto, or if for any other reason (including, but not limited to, any requirement that the Company or the Depositary withhold an amount on account of taxes or other governmental charges or that such securities must be registered under the Securities Act of 1933 in order to be distributed to Owners or Beneficial Owners) the Depositary deems such distribution not to be feasible, the Depositary may, after consultation with the Company, adopt such method as it may deem equitable and practicable for the purpose of effecting such distribution, including, but not limited to, the public or private sale of the securities or property thus received, or any part thereof, and the net proceeds of any such sale (net of the fees and expenses of the Depositary as provided in Section 5.09) shall be distributed by the Depositary to the Owners entitled thereto, all in the manner and subject to the conditions described in Section 4.01; provided, however, that no distribution to Owners pursuant to this Section 4.02 shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any distribution of securities under this Section 4.02 if

it has not received reasonably satisfactory assurances from the Company that the distribution does not require registration under the Securities Act of 1933.

### SECTION 4.03. Distributions in Shares.

If any distribution upon any Deposited Securities consists of a dividend in, or free distribution of, Shares, the Depositary may, and shall if the Company shall so request, distribute to the Owners of outstanding Receipts entitled thereto, in proportion to the number of American Depositary Shares representing such Deposited Securities held by them respectively, additional Receipts evidencing an aggregate number of American Depositary Shares representing the amount of Shares received as such dividend or free distribution, subject to the terms and conditions of the Deposit Agreement with respect to the deposit of Shares and the issuance of American Depositary Shares evidenced by Receipts, including the withholding of any tax or other governmental charge as provided in Section 4.11 and the payment of the fees and expenses of the Depositary as provided in Section 5.09; provided, however, that no distribution to Owners pursuant to this Section 4.03 shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any such distribution of Receipts if it has not received satisfactory assurances from the Company that such distribution does not require registration under the Securities Act of 1933. In lieu of delivering Receipts for fractional American Depositary Shares in any such case, the Depositary shall use reasonable efforts to sell the amount of Shares represented by the aggregate of such fractions and distribute the net proceeds, if any, all in the manner and subject to the conditions described in Section 4.01. If additional Receipts are not so distributed, each American Depositary Share shall thenceforth also represent the additional Shares distributed upon the Deposited Securities represented thereby.

## SECTION 4.04. Rights.

In the event that the Company shall offer or cause to be offered to the holders of any Deposited Securities any rights to subscribe for additional Shares or any rights of any other nature, the Depositary, after consultation with the Company, shall have discretion as to the procedure to be followed in making such rights available to any Owners or in disposing of such rights on behalf of any Owners and making the net proceeds available to such Owners or, if by the terms of such rights offering or for any other reason, the Depositary may not either make such rights available to any Owners or dispose of such rights and make the net proceeds available to such Owners, then the Depositary shall allow the rights to lapse. If at the time of the offering of any rights the Depositary, after consultation with the Company, determines in its reasonable discretion that it is lawful and feasible to make such rights available to all or certain Owners but not to other Owners, the Depositary may, and at the request of the Company shall, distribute to any Owner to whom it determines the distribution to be lawful and feasible, in proportion to the number of American Depositary Shares held by such Owner, warrants or other instruments therefor in such form as it deems appropriate.

In circumstances in which rights would otherwise not be distributed, if an Owner of Receipts requests the distribution of warrants or other instruments in order to exercise the rights allocable to the American Depositary Shares of such Owner hereunder, the Depositary will make such rights available to such Owner upon written notice from the Company to the

Depositary that (a) the Company has elected in its sole discretion to permit such rights to be exercised and (b) such Owner has executed such documents as the Company has determined in its sole discretion are reasonably required under applicable law.

If the Depositary has distributed warrants or other instruments for rights to all or certain Owners, then upon instruction from such an Owner pursuant to such warrants or other instruments to the Depositary from such Owner to exercise such rights, upon payment by such Owner to the Depositary for the account of such Owner of an amount equal to the purchase price of the Shares to be received upon the exercise of the rights, and upon payment of the fees and expenses of the Depositary and any other charges as set forth in such warrants or other instruments, the Depositary shall, on behalf of such Owner, exercise the rights and purchase the Shares, and the Company shall cause the Shares so purchased to be delivered to the Depositary on behalf of such Owner. As agent for such Owner, the Depositary will cause the Shares so purchased to be deposited pursuant to Section 2.02 of this Deposit Agreement, and shall, pursuant to Section 2.03 of this Deposit Agreement, execute and deliver Receipts to such Owner; <u>provided</u>, <u>however</u>, that in the case of a distribution pursuant to the second paragraph of this Section, such Receipts shall be legended in accordance with applicable U.S. laws, and shall be subject to the appropriate restrictions on sale, deposit, cancellation, and transfer under applicable United States laws.

If the Depositary determines that it is not lawful and feasible to make such rights available to all or certain Owners, it may sell the rights, warrants or other instruments in proportion to the number of American Depositary Shares held by the Owners to whom it has determined it may not lawfully or feasibly make such rights available, and allocate the net proceeds of such sales (net of the fees and expenses of the Depositary as provided in Section 5.09 and all taxes and governmental charges payable in connection with such rights and subject to the terms and conditions of this Deposit Agreement) for the account of such Owners otherwise entitled to such rights, warrants or other instruments, upon an averaged or other practical basis without regard to any distinctions among such Owners because of exchange restrictions or the date of delivery of any Receipt or otherwise.

The Depositary will not offer rights to Owners unless both the rights and the securities to which such rights relate are either exempt from registration under the Securities Act of 1933 with respect to a distribution to all Owners or are registered under the provisions of such Act; provided, that nothing in this Deposit Agreement shall create any obligation on the part of the Company to file a registration statement with respect to such rights or underlying securities or to endeavor to have such a registration statement declared effective. If an Owner of Receipts requests the distribution of warrants or other instruments, notwithstanding that there has been no such registration under the Securities Act of 1933, the Depositary shall not effect such distribution unless it has received an opinion from recognized counsel in the United States for the Company upon which the Depositary may rely that such distribution to such Owner does not require such registration.

The Depositary shall not be responsible for any failure to determine that it may be lawful or feasible to make such rights available to Owners in general or any Owner in particular.

## SECTION 4.05. Conversion of Foreign Currency.

Whenever the Depositary or the Custodian shall receive foreign currency, by way of dividends or other distributions or the net proceeds from the sale of securities, property or rights, and if at the time of the receipt thereof the foreign currency so received can in the reasonable judgment of the Depositary be converted on a reasonable basis into Dollars and the resulting Dollars transferred to the United States, the Depositary shall, as promptly as practicable, convert or cause to be converted, by sale or in any other manner that it may reasonably determine, such foreign currency into Dollars, and such Dollars shall be distributed to the Owners entitled thereto or, if the Depositary shall have distributed any warrants or other instruments which entitle the holders thereof to such Dollars, then to the holders of such warrants and/or instruments upon surrender thereof for cancellation. Such distribution may be made upon an averaged or other practicable basis without regard to any distinctions among Owners on account of exchange restrictions, the date of delivery of any Receipt or otherwise and shall be net of any expenses of conversion into Dollars incurred by the Depositary as provided in Section 5.09.

If such conversion or distribution can be effected only with the approval or license of any government or agency thereof, the Depositary shall file such application for approval or license, if any, as it may deem desirable.

If at any time the Depositary shall determine that in its reasonable judgment any foreign currency received by the Depositary or the Custodian is not convertible on a reasonable basis into Dollars transferable to the United States, or if any approval or license of any government or agency thereof which is required for such conversion is denied or in the reasonable opinion of the Depositary is not obtainable, or if any such approval or license is not obtained within a reasonable period as determined by the Depositary, the Depositary may distribute the foreign currency (or an appropriate document evidencing the right to receive such foreign currency) received by the Depositary to, or in its discretion may hold such foreign currency uninvested and without liability for interest thereon for the accounts of, the Owners entitled to receive the same.

If any such conversion of foreign currency, in whole or in part, cannot be effected for distribution to some of the Owners entitled thereto, the Depositary may in its discretion make such conversion and distribution in Dollars to the extent permissible to the Owners entitled thereto and may distribute the balance of the foreign currency received by the Depositary to, or hold such balance uninvested and without liability for interest thereon for the accounts of, the Owners entitled thereto.

## SECTION 4.06. Fixing of Record Date.

Whenever any cash dividend or other cash distribution shall become payable or any distribution other than cash shall be made, or whenever rights shall be issued with respect to the Deposited Securities, or whenever the Depositary shall receive notice of any meeting of holders of Shares or other Deposited Securities, or whenever for any reason the Depositary causes a change in the number of Shares that are represented by each American Depositary

Share, or whenever the Depositary shall find it necessary, the Depositary shall fix a record date (a) for the determination of the Owners who shall be (i) entitled to receive such dividend, distribution or rights or the net proceeds of the sale thereof or (ii) entitled to give instructions for the exercise of voting rights at any such meeting or (iii) responsible for any fee or charges assessed by the Depositary pursuant to this Deposit Agreement, or (b) on or after which each American Depositary Share will represent the changed number of Shares. Subject to the provisions of Sections 4.01 through 4.05 and to the other terms and conditions of this Deposit Agreement, the Owners on such record date shall be entitled, as the case may be, to receive the amount distributable by the Depositary with respect to such dividend or other distribution or such rights or the net proceeds of sale thereof in proportion to the number of American Depositary Shares held by each of them, to give voting instructions and to act in respect of any other such matter.

## SECTION 4.07. Voting of Deposited Securities.

Upon the written request of an Owner of record of a Receipt as of the record date (the "Voting Record Date") received on or before the date established by the Depositary for such purpose (the "Instruction Date"), the Depositary will endeavor to cause the appointment (or, if the Deposited Securities are registered in the name of or held by its Custodian or a nominee, the Depositary hereby agrees to procure that the Custodian or its nominee shall cause the appointment), subject to the Articles of Association of the Company, such Owner as of the Voting Record Date fixed by the Depositary in accordance with Section 4.06 as a proxy in respect of any meeting (including any adjourned meeting) at which such Owner will be entitled to attend and vote at in respect of the Deposited Securities represented by the American Depositary Shares evidenced by the Receipts held by such Owner on the Voting Record Date. In respect of any such meeting each such Owner may appoint a person nominated by the Depositary, its Custodian or nominee, to vote on behalf of the Owner subject to and in accordance with the provisions of this Section 4.07 and the Articles of Association of the Company. Upon receipt of notice of any meeting of holders of Shares or other Deposited Securities, if requested in writing by the Company the Depositary shall, as soon as practicable thereafter, mail to the Owners of record as of the Voting Record Date a notice, the form of which notice shall be approved by the Company which shall contain (a) such information as is contained in such notice of meeting, (b) a voting instruction card in the form prepared by the Depositary after consultation with the Company, (c) a statement that the Owners of record as of the close of business on the Voting Record Date will be entitled, subject to any applicable provision of English law and of the Memorandum and Articles of Association of the Company and the provisions of or governing the Deposited Securities, either (i) to use such voting instruction card to inform the Depositary that the Owner intends to attend such meeting as the proxy of the Depositary, the Custodian or its nominee (as appropriate) solely with respect to the Shares or other Deposited Securities represented by American Depositary Shares evidenced by such Owner's Receipts or (ii) to instruct such person nominated by the Depositary, the Custodian or nominee as to the exercise of the voting rights pertaining to the number of Shares or other Deposited Securities represented by the American Depositary Shares evidenced by such Owner's Receipts, and (d) a brief statement as to the manner in which voting instructions may be given to the person nominated by the Depositary. Upon the written request of an Owner of record of a Receipt on the Voting Record Date, received on or before the Instruction Date the Depositary

shall endeavor, in so far as practicable, to vote or cause to be voted the number of Shares or other Deposited Securities represented by the American Depositary Shares evidenced by such Receipt in accordance with the instructions set forth in such request. Neither the Depositary, nor the Custodian nor the nominee of either of them shall vote or attempt to exercise the right to vote that attaches to the Shares or other Deposited Securities, other than in accordance with such written instructions from Owners given in accordance with this Section 4.07. If no valid written instructions are received by the Depositary from an Owner with respect to any of the number of Deposited Securities represented by the American Depositary Shares evidenced by such Owner's Receipts on or before the Instruction Date, that number of Deposited Securities shall not be voted by the Depositary, the Custodian or the nominee of either of them.

There can be no assurance that Owners generally or any Owner in particular will receive the notice described in the preceding paragraph sufficiently prior to the Instruction Date to ensure that the Depositary will appoint the Owner as proxy or vote the Shares or other Deposited Securities in accordance with the provisions set forth in the preceding paragraph.

## SECTION 4.08. Changes Affecting Deposited Securities.

In circumstances where the provisions of Section 4.03 do not apply, upon any change in nominal value, change in par value, split-up, consolidation or any other reclassification of Deposited Securities, or upon any recapitalization, reorganization, merger or consolidation or sale of assets affecting the Company or to which it is a party, any securities which shall be received by the Depositary or a Custodian in exchange for or in conversion of or in respect of Deposited Securities, shall be treated as new Deposited Securities under this Deposit Agreement, and American Depositary Shares shall thenceforth represent, in addition to the existing Deposited Securities, the right to receive the new Deposited Securities so received in exchange or conversion, unless additional Receipts are delivered pursuant to the following sentence. In any such case the Depositary may, and shall if the Company shall so reasonably request, execute and deliver additional Receipts as in the case of a dividend in Shares, or call for the surrender of outstanding Receipts to be exchanged for new Receipts specifically describing such new Deposited Securities.

## SECTION 4.09. Reports.

The Company is subject to the periodic reporting requirements of the Securities Exchange Act of 1934 and accordingly files certain reports with the United States Securities and Exchange Commission (hereinafter called the "Commission"). Such reports and other information may be inspected and copied at public reference facilities maintained by the Commission located at the date hereof at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549. The Depositary shall make available for inspection by Owners at its Principal Office any reports and communications, including any proxy soliciting material, received from the Company which are both (a) received by the Depositary as the holder of the Deposited Securities and (b) made generally available to the holders of such Deposited Securities by the Company. The Depositary shall also, upon the written request of the Company, send to the Owners copies of such reports when furnished by the Company pursuant to Section 5.06.

### SECTION 4.10. Lists of Owners.

Upon the written request of the Company, the Depositary shall, as promptly as practical, at the expense of the Company, furnish to it a list, as of a recent date, of the names, addresses and holdings of American Depositary Shares by all persons in whose names Receipts are registered on the books of the Depositary.

## SECTION 4.11. Withholding.

In the event that the Depositary determines that any distribution in property (including Shares and rights to subscribe therefor) is subject to any tax or other governmental charge which the Depositary is obligated to withhold, the Depositary may by public or private sale dispose of all or a portion of such property (including Shares and rights to subscribe therefor) in such amounts and in such manner as the Depositary reasonably deems necessary and practicable to pay such taxes or charges and the Depositary shall distribute the net proceeds of any such sale after deduction of such taxes or charges to the Owners entitled thereto in proportion to the number of American Depositary Shares held by them.

## ARTICLE 5. THE DEPOSITARY, THE CUSTODIANS AND THE COMPANY

## SECTION 5.01. Maintenance of Office and Transfer Books by the Depositary.

Until termination of this Deposit Agreement in accordance with its terms, the Depositary shall maintain in the Borough of Manhattan, The City of New York, facilities for the execution and delivery, registration, registration of transfers and surrender of Receipts in accordance with the provisions of this Deposit Agreement.

The Depositary shall keep books, at its Principal Office, for the registration of Receipts and transfers of Receipts which at all reasonable times shall be open for inspection by the Owners, provided that such inspection shall not be for the purpose of communicating with Owners in the interest of a business or object other than the business of the Company or a matter related to this Deposit Agreement or the Receipts.

The Depositary may close the transfer books, at any time or from time to time, when reasonably deemed expedient by it in connection with the performance of its duties hereunder or at the written reasonable request of the Company, provided that any such closing of the transfer books shall be subject to the provisions of Section 2.06 which limit the suspension of withdrawals of Shares.

If any Receipts or the American Depositary Shares evidenced thereby are listed on one or more stock exchanges in the United States, the Depositary shall act as Registrar or appoint a Registrar or one or more co-registrars for registry of such Receipts in accordance with any requirements of such exchange or exchanges. The Company shall have the right, upon

reasonable written request, to inspect the transfer and registration records of the Depositary relating to the Receipts, and to take copies thereof.

### SECTION 5.02. Prevention or Delay in Performance by the Depositary or the Company.

Neither the Depositary nor the Company nor any of their directors, employees, agents or affiliates shall incur any liability to any Owner or Beneficial Owner if, by reason of any provision of any present or future law or regulation of the United States or any other country, or of any governmental or regulatory authority or stock exchange, or by reason of any provision, present or future, of the Memorandum or Articles of Association of the Company, or by reason of any provision of any securities issued or distributed by the Company, or any offering or distribution thereof, or by reason of any act of God or war or terrorism or other circumstances beyond its control, the Depositary or the Company shall be prevented, delayed or forbidden from, or be subject to any civil or criminal penalty on account of, doing or performing any act or thing which by the terms of this Deposit Agreement or Deposited Securities it is provided shall be done or performed; nor shall the Depositary or the Company or any of their directors, employees, agents or affiliates incur any liability to any Owner or Beneficial Owner of any Receipt by reason of any nonperformance or delay, caused as aforesaid, in the performance of any act or thing which by the terms of this Deposit Agreement it is provided shall or may be done or performed, or by reason of any exercise of, or failure to exercise, any discretion provided for in this Deposit Agreement. Where, by the terms of a distribution pursuant to Section 4.01, 4.02, or 4.03, or an offering or distribution pursuant to Section 4.04, or for any other reason, such distribution or offering may not be made available to Owners, and the Depositary may not dispose of such distribution or offering on behalf of such Owners and make the net proceeds available to Such Owners, then the Depositary shall not make such distribution or offering, and shall allow any rights, if applicable, to lapse.

## SECTION 5.03. Obligations of the Depositary, the Custodian and the Company.

Neither the Company nor any of its directors, officers, employees or agents assumes any obligation nor shall any of them be subject to any liability under this Deposit Agreement to Owners or Beneficial Owners, except that the Company agrees to perform its obligations specifically set forth in this Deposit Agreement without negligence or bad faith.

Neither the Depositary nor any of its directors, officers, employees or agents assumes any obligation nor shall any of them be subject to any liability under this Deposit Agreement to any Owner or Beneficial Owner (including, without limitation, liability with respect to the validity or worth of the Deposited Securities), except that the Depositary agrees to perform its obligations specifically set forth in this Deposit Agreement without negligence or bad faith.

Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be under any obligation to appear in or prosecute or defend any action, suit or other proceeding in respect of any Deposited Securities or in respect of the

Receipts on behalf of any Owner, Beneficial Owner or other person. The parties understand that the Custodian is not a party to this Deposit Agreement and, accordingly, has no obligations whatsoever with respect to such proceedings, the responsibility of the Custodian being solely to the Depositary.

In connection with and in accordance with the arrangements and procedures relating to DRS/Profile, the parties understand that the Depositary will not verify, determine or otherwise ascertain that the DTC participant which is claiming to be acting on behalf of an Owner in requesting registration of transfer and delivery described in subsection (a) has the actual authority to act on behalf of the Owner. Each Owner agrees that neither the Depositary nor the Company shall have any liability for the Depositary's or any of its agents' reliance upon the authority of any information in, nor for the Depositary's or any of its agents' compliance with directions from, any DTC participants as set forth above. Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be liable for any action or nonaction by it in reliance upon the advice of or information from legal counsel, accountants, any person presenting Shares for deposit, any Owner or any other person believed by it in good faith to be competent to give such advice or information. Each of the Depositary, the Company and their respective directors, officers, employees and agents may rely and shall be protected in acting upon any written notice, request, direction or other document believed by such person to be genuine and to have been signed or presented by the proper party or parties.

The Depositary shall not be liable for any acts or omissions made by a successor depositary whether in connection with a previous act or omission of the Depositary or in connection with any matter arising wholly after the removal or resignation of the Depositary, provided that in connection with the issue out of which such potential liability arises the Depositary performed its obligations without negligence or bad faith while it acted as Depositary.

The Depositary and its agents will not be responsible for (i) any failure to carry out any instructions to vote any of the Deposited Securities or for the manner in which any such vote is cast, in each case to the extent the Depositary or its agents act without gross negligence or willful misconduct or (ii) for the effect of any such vote.

Notwithstanding anything to the contrary set forth in the Deposit Agreement or any Receipt, the Depositary and its agents may fully respond to any and all demands or requests for information maintained by or on its behalf in connection with the Deposit Agreement, any Owner or Owners, any Receipt or Receipts or otherwise related hereto to the extent such information is requested or required by or pursuant to any lawful authority, including without limitation laws, rules, regulations, administrative or judicial process, banking, securities or other regulators. Neither the Company nor the Depositary nor any of their respective agents shall be liable to Owners or holders of interests in American Depositary Shares or any other third party or parties for any indirect, special, punitive or consequential damages.

No disclaimer of liability under the Securities Act of 1933 is intended by any provision of this Deposit Agreement.

## SECTION 5.04. Resignation and Removal of the Depositary.

The Depositary may at any time resign as Depositary hereunder by written notice of its election so to do delivered to the Company, such resignation to take effect upon the appointment of a successor depositary and its acceptance of such appointment as hereinafter provided.

The Depositary may at any time be removed by the Company by written notice to the Depositary, such removal to become effective upon the appointment of a successor depositary and its acceptance of such appointment as hereinafter provided.

In case at any time the Depositary acting hereunder shall resign or be removed, the Company shall use its reasonable efforts to appoint a successor depositary, which shall be a bank or trust company having an office in the Borough of Manhattan, The City of New York. Every successor depositary shall execute and deliver to its predecessor and to the Company an instrument in writing accepting its appointment hereunder, and thereupon such successor depositary, without any further act or deed, shall become fully vested with all the rights, powers, duties and obligations of its predecessor; but such predecessor, nevertheless, upon payment of all sums due it and on the written request of the Company shall execute and deliver an instrument transferring to such successor all rights and powers of such predecessor hereunder, shall duly assign, transfer and deliver all right, title and interest in the Deposited Securities to such successor, and shall deliver to such successor a list of the Owners of all outstanding Receipts. Any such successor depositary shall promptly mail notice of its appointment to the Owners.

Any corporation into or with which the Depositary may be merged or consolidated shall be the successor of the Depositary without the execution or filing of any document or any further act.

### SECTION 5.05. The Custodians.

The Custodian shall be subject at all times and in all respects to the directions of the Depositary and shall be responsible solely to it. If the Depositary receives notice of the resignation of a Custodian and, upon the effectiveness of such resignation, there would be no Custodian acting hereunder, the Depositary shall, promptly after receiving such notice and upon consultation with the Company if practicable, appoint a substitute custodian or custodians, each of which shall thereafter be a Custodian hereunder. Whenever the Depositary in its reasonable discretion determines that it is in the best interest of the Owners to do so, it may, after consultation with the Company if practicable, appoint, a substitute or additional custodian or custodians, each of which shall thereafter be one of the Custodians hereunder. The Depositary shall require each such substitute or additional custodian to deliver to the Depositary, forthwith upon its appointment, an acceptance of such appointment satisfactory in form and substance to the Depositary.

Upon the appointment of any successor depositary hereunder, each Custodian then acting hereunder shall forthwith become, without any further act or writing, the agent

hereunder of such successor depositary and the appointment of such successor depositary shall in no way impair the authority of each Custodian hereunder; but the successor depositary so appointed shall, nevertheless, on the written request of any Custodian, execute and deliver to such Custodian all such instruments as may be proper to give to such Custodian full and complete power and authority as agent hereunder of such successor depositary.

### SECTION 5.06. Notices and Reports.

On or before the first date on which the Company gives notice, by publication or otherwise, of any meeting of holders of Shares or other Deposited Securities, or of any adjourned meeting of such holders, or of the taking of any action in respect of any cash or other distributions or the offering of any rights, the Company agrees to transmit to the Depositary and the Custodian a copy of the notice thereof in English but otherwise in the form given or to be given to holders of Shares or other Deposited Securities.

The Company will arrange for the translation into English, if not already in English, to the extent required pursuant to any regulations of the Commission, and the prompt transmittal by the Company to the Depositary and the Custodian of such notices and any other reports and communications which are made generally available by the Company to holders of its Shares. If requested by the Company, the Depositary will arrange for the mailing, at the Company's expense, of copies of such notices, reports and communications to all Owners. The Company will timely provide the Depositary with the quantity of such notices, reports, and communications, as reasonably requested by the Depositary from time to time, in order for the Depositary to effect such mailings.

# SECTION 5.07. <u>Distribution of Additional Shares, Rights, etc.</u>

If the Company or any affiliate of the Company determines to make any issuance or distribution of (1) additional Shares, (2) rights to subscribe for Shares, (3) securities convertible into Shares, or (4) rights to subscribe for such securities (each a "Distribution"), the Company shall notify the Depositary in writing in English as promptly as practicable and in any event before the Distribution starts and, if requested in writing by the Depositary, the Company shall promptly furnish to the Depositary a written opinion from U.S. counsel for the Company that is reasonably satisfactory to the Depositary, stating whether or not the Distribution requires, or, if made in the United States, would require, registration under the Securities Act of 1933. If, in the opinion of that counsel, the Depositary a written opinion as to whether or not there is a registration statement under the Securities Act of 1933 in effect that will cover that Distribution.

The Company agrees with the Depositary that neither the Company nor any company controlled by, controlling or under common control with the Company will at any time deposit any Shares, either originally issued or previously issued and reacquired by the Company or any such affiliate, unless a registration statement is in effect as to such Shares under the Securities Act of 1933.

The Company reserves full discretion as to whether in the future it may or may not register under said Act for purposes of offering and selling in the United States any Shares or any other securities, including any Shares or other securities which may be the subject of subscription or purchase rights pertaining to Deposited Securities at the time deposited under this Deposit Agreement.

### SECTION 5.08. Indemnification.

The Company agrees to indemnify the Depositary, its directors, employees, agents and affiliates and any Custodian against, and hold each of them harmless from, any liability or expense (including, but not limited to, the reasonable fees and expenses of counsel) which may arise out of or in connection with (a) any registration with the Commission of Receipts, American Depositary Shares or Deposited Securities or the offer or sale thereof in the United States or (b) acts performed or omitted, pursuant to the provisions of this Deposit Agreement and of the Receipts, as the same may be amended, modified or supplemented from time to time, (i) by either the Depositary or a Custodian or their respective directors, employees, agents and affiliates, or (ii) by the Company or any of its directors, employees, agents and affiliates excepting, however, any liability arising out of the negligence or bad faith of the Depositary or the Custodian or any of their respective directors, employees, agents or affiliates or the Registrar or any co-transfer agent.

The Depositary agrees to indemnify the Company, its directors, employees, agents and affiliates and hold them harmless from any liability or expense which may arise out of acts performed or omitted by the Depositary or its Custodian or their respective directors, employees, agents and affiliates in connection with the issuance of Pre-Released Receipts and the transactions contemplated by any Pre-Release Agreement or due to their negligence or bad faith.

Notwithstanding any other provision of this Deposit Agreement or the Receipts to the contrary, neither the Company nor the Depositary, nor any of their agents, shall be liable to the other for any indirect, special, punitive or consequential damages (collectively "Special Damages") except (i) to the extent such Special Damages arise from the gross negligence or willful misconduct of the party from whom indemnification is sought or (ii) to the extent Special Damages arise from or out of a claim brought by a third party (including, without limitation, Owners) against the Depositary or its agents, except to the extent such Special Damages arise out of the gross negligence or willful misconduct of the party seeking indemnification hereunder.

The Company and the Depositary agree that the foregoing indemnification shall apply to the Depositary's implementation of DRS/Profile and to the extent the relevant transfer is performed in connection with and accordance with the arrangements and procedures related to Profile generally in effect that reliance by the Depositary on the authority of a DTC participant, claiming to act on behalf of an Owner of Direct Registration Receipts, to direct the Depositary to register a transfer of American Depositary Shares to DTC or its nominee or to deliver American Depositary Shares to the DTC account of that DTC participant, without receipt by the Depositary of prior authorization from the Owner to register such transfer or make such delivery (unless such prior authorization is required by DRS/Profile), shall not be deemed negligence, gross

negligence, bad faith or willful misconduct by the Depositary within the meaning of this Section 5.08, unless the Depositary had actual knowledge, or had reason to know (despite the absence of an investigation), that such direction was not authorized or was otherwise invalid.

### SECTION 5.09. Charges of Depositary.

The Company agrees to pay the fees, reasonable expenses and out-of-pocket charges of the Depositary and those of any Registrar only in accordance with agreements in writing entered into between the Depositary and the Company from time to time. The Depositary shall present its statement for such charges and expenses to the Company once every three months. The charges and expenses of the Custodian are for the sole account of the Depositary.

The following charges shall be incurred by any party depositing or withdrawing Shares or by any party surrendering Receipts or to whom Receipts are issued (including, without limitation, issuance pursuant to a stock dividend or stock split declared by the Company or an exchange of stock regarding the Receipts or Deposited Securities or a distribution of Receipts pursuant to Section 4.03), or by Owners, as applicable: (1) taxes and other governmental charges, (2) such registration fees as may from time to time be in effect for the registration of transfers of Shares generally on the Share register of the Company or Foreign Registrar and applicable to transfers of Shares to or from the name of the Depositary or its nominee or the Custodian or its nominee on the making of deposits or withdrawals hereunder, (3) such cable, telex and facsimile transmission expenses as are expressly provided in this Deposit Agreement, (4) such expenses as are incurred by the Depositary in the conversion of foreign currency pursuant to Section 4.05, (5) a fee of \$5.00 or less per 100 American Depositary Shares (or portion thereof) for the execution and delivery of Receipts pursuant to Section 2.03, 4.03 or 4.04 and the surrender of Receipts pursuant to Section 2.05 or 6.02, and (6) a fee for the distribution of securities pursuant to Section 4.02, such fee being in an amount equal to the fee for the execution and delivery of American Depositary Shares referred to above which would have been charged as a result of the deposit of such securities (for purposes of this clause 6 treating all such securities as if they were Shares) but which securities are instead distributed by the Depositary to Owners.

The Depositary, subject to Section 2.09, may own and deal in any class of securities of the Company and its affiliates and in Receipts.

## SECTION 5.10. Retention of Depositary Documents.

The Depositary is authorized to destroy those documents, records, bills and other data compiled during the term of this Deposit Agreement at the times permitted by the laws or regulations governing the Depositary unless the Company requests that such papers be retained for a longer period.

### SECTION 5.11. Exclusivity.

Subject to Section 5.04, the Company agrees not to appoint any other depositary for issuance of American or global depositary receipts for the Shares so long as JPMorgan Chase Bank, N.A. is acting as Depositary hereunder.

## SECTION 5.12. List of Restricted Securities Owners.

The Company shall provide to the Depositary a list setting forth, to the actual knowledge of the Company, those persons or entities who beneficially own Restricted Securities as of the date hereof and the Company shall update that list on a regular basis as changes occur. The Company agrees to advise in writing each of the persons or entities so listed that such Restricted Securities, so long as they remain such, are ineligible for deposit hereunder. The Depositary (i) may rely on the list provided under this Section 5.12, as most recently updated, but shall not be liable for any action or omission made in reliance thereon and (ii) shall keep such a list strictly confidential, except as required by applicable law, legal process, regulation, judicial or administrative proceeding.

### ARTICLE 6. AMENDMENT AND TERMINATION

### SECTION 6.01. Amendment.

The form of the Receipts and any provisions of this Deposit Agreement may at any time and from time to time be amended by agreement between the Company and the Depositary without the consent of Owners or Beneficial Owners in any respect which they may deem necessary or desirable. Any amendment which shall impose or increase any fees or charges (other than taxes and other governmental charges, registration fees, cable, telex or facsimile transmission costs, delivery costs or other such expenses), or which shall otherwise prejudice any substantial existing right of Owners, shall, however, not become effective as to outstanding Receipts until the expiration of 30 days after notice of such amendment shall have been given to the Owners of outstanding Receipts. Every Owner and Beneficial Owner, at the time any amendment so becomes effective, shall be deemed, by continuing to hold such Receipt or any interest therein, to consent and agree to such amendment and to be bound by the Deposit Agreement as amended thereby. In no event shall any amendment impair the right of the Owner of any Receipt to surrender such Receipt and receive therefor the Deposited Securities represented thereby, except in order to comply with mandatory provisions of applicable law.

The Company and the Depositary shall each use their reasonable efforts to amend this Deposit Agreement as necessary to reflect changes in English or U.S. law and in the Company's Memorandum and Articles of Association.

### SECTION 6.02. Termination.

The Depositary shall, at any time at the direction of the Company, terminate this Deposit Agreement by mailing notice of such termination to the Owners of all Receipts then outstanding at least 30 days prior to the date fixed in such notice for such termination. The Depositary may likewise terminate this Deposit Agreement by mailing notice of such termination

to the Company and the Owners of all Receipts then outstanding, if at any time 60 days shall have expired after the Depositary shall have delivered to the Company a written notice of its election to resign and a successor depositary shall not have been appointed and accepted its appointment as provided in Section 5.04. On and after the date of termination, the Owner of a Receipt will, upon (a) surrender of such Receipt at the Principal Office of the Depositary, and (b) payment of the fee of the Depositary for the surrender of Receipts referred to in section 2.05, and (c) payment of any applicable taxes or governmental charges, be entitled to delivery, to him or upon his order, of the amount of Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, If any Receipts shall remain outstanding after the date of termination, the Depositary thereafter shall discontinue the registration of transfers of Receipts, shall suspend the distribution of dividends to the Owners thereof, and shall not give any further notices or perform any further acts under this Deposit Agreement, except that the Depositary shall continue to collect dividends and other distributions pertaining to Deposited Securities, shall sell rights and other property as provided in this Deposit Agreement, and shall continue to deliver Deposited Securities, together with any dividends or other distributions received with respect thereto and the net proceeds of the sale of any rights or other property, in exchange for Receipts surrendered to the Depositary (after deducting, in each case, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of this Deposit Agreement, and any applicable taxes or governmental charges) and for its obligations under Section 5.08 hereof. At any time after the expiration of six months from the date of termination, the Depositary may sell the Deposited Securities then held hereunder and may thereafter hold the net proceeds of any such sale, together with any other cash then held by it hereunder, without liability for interest, for the pro rata benefit of the Owners of Receipts which have not theretofore been surrendered, such Owners thereupon becoming general creditors of the Depositary with respect to such net proceeds. All such proceeds and cash shall be invested in direct obligations of the federal government of the United States. After making such sale, the Depositary shall be discharged from all obligations under this Deposit Agreement, except to account for such net proceeds and other cash (after deducting, in each case, the fee of the Depositary for the surrender of a Receipt, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of this Deposit Agreement, and any applicable taxes or governmental charges) and for its obligations under Section 5.08 hereof. Upon the termination of this Deposit Agreement, the Company shall be discharged from all obligations under this Deposit Agreement except for its obligations to the Depositary under Sections 5.08 and 5.09.

## **ARTICLE 7. MISCELLANEOUS**

SECTION 7.01. Counterparts.

This Deposit Agreement may be executed in any number of counterparts, each of which shall be deemed an original and all of such counterparts shall constitute one and the same instrument. Copies of this Deposit Agreement shall be filed with the Depositary and the Custodian and shall be open to inspection by any Owner or Beneficial Owner during business hours.

## SECTION 7.02. No Third Party Beneficiaries.

This Deposit Agreement is for the exclusive benefit of the parties hereto and, except with respect to indemnification of the Custodian as set forth in Section 5.08, shall not be deemed to give any legal or equitable right, remedy or claim whatsoever to any other person.

#### SECTION 7.03. Severability.

In case any one or more of the provisions contained in this Deposit Agreement or in the Receipts should be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein or therein shall in no way be affected, prejudiced or disturbed thereby.

## SECTION 7.04. Owners and Beneficial Owners as Parties; Binding Effect.

The Owners and Beneficial Owners of Receipts from time to time shall be parties to this Deposit Agreement and shall be bound by all of the terms and conditions hereof and of the Receipts by acceptance thereof or any interest therein.

## SECTION 7.05. Notices.

Any and all notices to be given to the Company shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex or facsimile transmission confirmed by letter, addressed to Royal Dutch Shell plc, Shell Centre, London SE1 7NA, England, Attn: Company Secretary, or any other place to which the Company may have transferred its registered office with notice to the Depositary.

Any and all notices to be given to the Depositary shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex or facsimile transmission confirmed by letter, addressed to JPMorgan Chase Bank, N.A., Four New York Plaza, New York, New York 10286, Attention: American Depositary Receipt Administration, or any other place to which the Depositary may have transferred its Principal Office with notice to the Company.

Any and all notices to be given to any Owner shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex or facsimile transmission confirmed by letter, addressed to such Owner at the address of such Owner as it appears on the transfer books for Receipts of the Depositary, or, if such Owner shall have filed with the Depositary a written request that notices intended for such Owner be mailed to some other address, at the address designated in such request.

Delivery of a notice sent by mail or cable, telex or facsimile transmission shall be deemed to be effective at the time when a duly addressed letter containing the same (or a confirmation thereof in the case of a cable, telex or facsimile transmission) is deposited, postage prepaid, in a post-office letter box. The Depositary or the Company may, however, act upon any cable, telex or facsimile transmission received by it, notwithstanding that such cable, telex or facsimile transmission shall not subsequently be confirmed by letter as aforesaid.

## SECTION 7.06. Compliance with U.S. Securities Laws.

Notwithstanding anything in this Deposit Agreement to the contrary, the Company and the Depositary each agrees that it will not exercise any rights it has under this Deposit Agreement to permit the withdrawal or delivery of Deposited Securities in a manner which would violate the U.S. securities laws, including, but not limited to, Section I.A.(1) of the General Instructions to the Form F-6 registration statement, as amended from time to time, under the Securities Act of 1933.

# SECTION 7.07. Governing Law; Jurisdiction.

This Deposit Agreement and the Receipts shall be interpreted and all rights hereunder and thereunder and provisions hereof and thereof shall be governed by the laws of the State of New York.

It is hereby acknowledged and agreed that Owners and Beneficial Owners of ADRs, as such, are not shareholders of the Company and have no direct rights of a shareholder against the Company. The rights of holders of and of the Company with respect to the Shares are governed exclusively by the Company's Memorandum and Articles of Association and the laws of England and Wales.

The Company and, by holding a Receipt, each Owner, but not the Depositary or its agents hereunder, shall be bound by the arbitration and exclusive jurisdiction provisions set out in articles 152, 153 and 154 of the Articles of Association of the Company from time to time (the "Articles"), as if all references therein to "shareholder" were replaced with "ADR holder". Articles 152, 153 and 154 of the Articles shall accordingly be incorporated, *mutatis mutandis*, into the terms of this Deposit Agreement. A copy of articles 152, 153 and 154 of the Articles is available upon written request to the Company. All cross-references to the Articles in Article 24 of the Receipt and in this Section 7.07 of the Deposit Agreement will be updated and amended without further action of any party in the event the Articles themselves are renumbered. English law shall govern any such arbitration.

IN WITNESS WHEREOF, ROYAL DUTCH SHELL plc and JPMORGAN CHASE BANK, N.A. have duly executed this Deposit Agreement as of the day and year first set forth above and all Owners and Beneficial Owners shall become parties hereto upon acceptance by them of Receipts issued in accordance with the terms hereof or any interest therein.

# ROYAL DUTCH SHELL plc

By: /s/ Peter Voser

Name: Peter Voser

Title: Director and Chief Financial Officer

JPMORGAN CHASE BANK, N.A.,

as Depositary

By: /s/ Melinda Van Luit

Name: Melinda Van Luit Title: Vice President

#### **EXHIBIT A**

No.

## **AMERICAN DEPOSITARY SHARES**

(Each American Depositary Share represents two deposited Shares)

JPMORGAN CHASE BANK, N.A.

AMERICAN DEPOSITARY RECEIPT

FOR ORDINARY SHARES

NOMINAL VALUE 0,07 EURO EACH OF

ROYAL DUTCH SHELL plc

(INCORPORATED UNDER THE LAWS OF ENGLAND AND WALES)

| JPMorgan Chase Bank, N.A., as depositary (herein called the Depositary), hereby certifies that | , or |
|--|------|
| registered assigns IS THE OWNER OF   |      |

#### **AMERICAN DEPOSITARY SHARES**

representing deposited ordinary shares (herein called Shares) of Royal Dutch Shell plc, incorporated under the laws of England and Wales (herein called the Company). At the date hereof, each American Depositary Share represents two Shares deposited or subject to deposit under the Deposit Agreement (as such term is hereinafter defined) at the principal Amsterdam office of ING Groep NV.

# THE DEPOSITARY'S PRINCIPAL OFFICE ADDRESS IS FOUR NEW YORK PLAZA, NEW YORK, N.Y. 10004

## 1. THE DEPOSIT AGREEMENT.

This American Depositary Receipt is one of an issue (herein called Receipts), all issued and to be issued upon the terms and conditions set forth in the deposit agreement, dated as of May 19, 2005, as the same may be amended from time to time in accordance with its terms (the "Deposit Agreement"), by and among the Company, the Depositary, and all Owners and Beneficial Owners from time to time of Receipts issued thereunder, each of whom by accepting a Receipt or any interest therein agrees to become a party thereto and become bound by all the terms and conditions thereof. The Deposit Agreement sets forth the rights of Owners and Beneficial Owners of the Receipts and the rights and duties of the Depositary in respect of the Shares deposited thereunder and any and all other securities, property and cash from time to time received in respect of such Shares and held thereunder (such Shares, securities, property, and cash are herein called Deposited Securities). Copies of the Deposit Agreement are on file at the Depositary's Principal Office in New York City and at the office of the Custodian.

The statements made on the face and reverse of this Receipt are summaries of certain provisions of the Deposit Agreement and are qualified by and subject to the detailed

provisions of the Deposit Agreement, to which reference is hereby made. Capitalized terms defined in the Deposit Agreement and not defined herein shall have the meanings set forth in the Deposit Agreement.

## 2. SURRENDER OF RECEIPTS AND WITHDRAWAL OF SHARES.

Upon surrender at the Principal Office of the Depositary of a Receipt in certificated form or proper instructions and documentation, in the case of a Direct Registration Receipt, and upon payment of the fee of the Depositary provided in this Receipt, and subject to the terms and conditions of the Deposit Agreement and English law, the Owner hereof is entitled to delivery, to him or upon his order, of the Deposited Securities at the time represented by the American Depositary Shares for which this Receipt is issued. Delivery of such Deposited Securities may be made by the delivery of (a) to the extent applicable, certificates in the name of the Owner hereof or as ordered by him or certificates properly endorsed or accompanied by proper instruments of transfer, (b) Deposited Securities to an account designated by such Owner with the Euroclear Nederland (including any successors thereto, "Euroclear Nederland") or an institution that maintains accounts with the Euroclear Nederland and (c) any other securities, property and cash to which such Owner is then entitled in respect of this Receipt. In the case of certificated Deposited Securities, such delivery will be made at the option of the Owner hereof, either at the office of the Custodian or at the Principal Office of the Depositary, provided that the forwarding of certificates for Shares or other Deposited Securities for such delivery at the Principal Office of the Depositary shall be at the risk and expense of the Owner hereof.

To the extent applicable, the Owner requesting withdrawal of Shares shall have the sole responsibility for ensuring that such Owner, or its customer, has a valid account with Euroclear Nederland or an institution that maintains accounts with Euroclear Nederland and that the information required for the bookentry transfer to such account is accurately and promptly provided to the Depositary.

## 3. TRANSFERS, SPLIT-UPS, AND COMBINATIONS OF RECEIPTS.

The transfer of this Receipt is registrable on the books of the Depositary at its Principal Office by the Owner hereof in person or by a duly authorized attorney, upon surrender of this Receipt properly endorsed for transfer or accompanied by proper instruments of transfer and funds sufficient to pay any applicable transfer taxes and the expenses of the Depositary and upon compliance with such regulations, if any, as the Depositary may establish for such purpose. This Receipt may be split into other such Receipts, or may be combined with other such Receipts into one Receipt, evidencing the same aggregate number of American Depositary Shares as the Receipt or Receipts surrendered. As a condition precedent to the execution and delivery, registration of transfer, split-up, combination, or surrender of any Receipt or withdrawal of any Deposited Securities, the Depositary, the Custodian, or Registrar may require payment from the depositor of the Shares or the presenter of the Receipt of a sum sufficient to reimburse it for any tax or other governmental charge and any stock transfer or registration fee with respect thereto (including any such tax or charge and fee with respect to Shares being deposited or withdrawn)

and payment of any applicable fees as provided in this Receipt, may require the production of proof satisfactory to it as to the identity and genuineness of any signature and may also require compliance with any regulations the Depositary may establish consistent with the provisions of the Deposit Agreement or this Receipt, including, without limitation, this Article 3.

The delivery of Receipts against deposit of Shares generally or against deposit of particular Shares may be suspended, or the transfer of Receipts in particular instances may be refused, or the registration of transfer of outstanding Receipts generally may be suspended, during any period when the transfer books of the Depositary are closed, or if any such action is deemed necessary or advisable by the Depositary or the Company at any time or from time to time because of any requirement of law or of any government or governmental body or commission, or under any provision of the Deposit Agreement or this Receipt, or for any other reason, subject to the provisions of Section 7.07 of the Deposit Agreement. Notwithstanding any other provision of the Deposit Agreement or the Receipts, the surrender of outstanding Receipts and withdrawal of Deposited Securities may not be suspended. The Depositary shall not knowingly accept for deposit under the Deposit Agreement any Shares which would be required to be registered under the provisions of the Securities Act of 1933 for the public offer and sale thereof in the United States unless a registration statement is in effect as to such Shares for such offer and sale. The Depositary will use reasonable efforts to comply with written instructions of the Company that the Depositary shall not accept for the deposit hereunder any Shares identified in such circumstances as may reasonably be specified in such restrictions to facilitate the Company's compliance with the U.S. securities laws. Notwithstanding anything to the contrary in the Deposit Agreement, Owners shall be entitled to surrender Receipts and withdraw Deposited Securities as provided in Section 2.05 of the Deposit Agreement at any time, subject only to (i) temporary delays caused by closing the transfer books of the Depositary or the Company or the deposit of Shares in connection with voting at a shareholders' meeting, or the payment of dividends, (ii) the payment of fees, taxes and similar charges, (iii) compliance with any U.S. or foreign laws or governmental regulations relating to the Receipts or to the withdrawal of the Deposited Securities and (iv) any other reason that may at any time be specified in paragraph I.(A)(1) of the General Instructions to Form F-6 under the Securities Act of 1933, from time to time in effect, or any successor provision thereto.

#### 4. LIABILITY OF OWNER OR BENEFICIAL OWNER FOR TAXES.

If any tax or other governmental charge shall become payable with respect to any Receipt or any Deposited Securities represented hereby, such tax or other governmental charge shall be payable by the Owner or Beneficial Owner hereof to the Depositary. The Depositary may refuse to effect any transfer of this Receipt or any withdrawal of Deposited Securities represented by American Depositary Shares evidenced by such Receipt until such payment is made, and may withhold any dividends or other distributions, or may sell for the account of the Owner or Beneficial Owner hereof any part or all of the Deposited Securities represented by the American Depositary Shares evidenced by this Receipt, and may apply such dividends or other distributions or the proceeds of any such sale in payment of such tax or other governmental charge and the Owner or Beneficial Owner hereof shall remain liable for any deficiency. Neither the Company nor the Depositary shall be liable for failure of an Owner to comply with applicable tax laws or governmental charges.

#### 5. WARRANTIES ON DEPOSIT OF SHARES.

Every person depositing Shares under the Deposit Agreement shall be deemed thereby to represent and warrant that such Shares and each certificate therefor are validly issued, fully paid, non-assessable, and were not issued in violation of any preemptive or similar rights of the holders of outstanding Shares and that the person making such deposit is duly authorized so to do. Every such person shall also be deemed to represent that such Shares are not, and American Depositary Shares representing such Shares would not be, Restricted Securities. Such representations and warranties shall survive the deposit of Shares and delivery of Receipts.

## 6. FILING PROOFS, CERTIFICATES, AND OTHER INFORMATION.

Any person presenting Shares for deposit or any Owner or Beneficial Owner of a Receipt may be required from time to time to file with the Depositary or the Custodian such proof of citizenship or residence, exchange control approval, or such information relating to the registration on the books of the Company or the Foreign Registrar, if applicable, to execute such certificates and to make such representations and warranties, as the Depositary may deem necessary or proper.

The Depositary may, and shall if requested by the Company pursuant to the provisions of Section 7.07 of the Deposit Agreement, withhold the delivery or registration of transfer of any Receipt or the distribution of any dividend or sale or distribution of rights or of the proceeds thereof or the delivery of any Deposited Securities until such proof or other information is filed or such certificates are executed or such representations and warranties made. The Depositary shall notify the Company, upon its request, of the availability of any such proofs, certificates or other information and shall provide copies thereof to the Company as promptly as practicable upon request by the Company, unless such disclosure is prohibited by law. No Share shall be accepted for deposit unless accompanied by evidence satisfactory to the Depositary that any necessary approval, where relevant, has been granted by any governmental body in England and Wales that is then performing the function of the regulation of currency exchange.

#### 7. CHARGES OF DEPOSITARY.

The Company agrees to pay the fees, reasonable expenses and out-of-pocket charges of the Depositary and those of any Registrar only in accordance with agreements in writing entered into between the Depositary and the Company from time to time. The Depositary shall present its statement for such charges and expenses to the Company once every three months. The charges and expenses of the Custodian are for the sole account of the Depositary.

The following charges shall be incurred by any party depositing or withdrawing Shares or by any party surrendering Receipts or to whom Receipts are issued (including, without limitation, issuance pursuant to a stock dividend or stock split declared by the Company or an exchange of stock regarding the Receipts or Deposited Securities or a distribution of Receipts pursuant to Section 4.03 of the Deposit Agreement), or by Owners, as applicable: (1) taxes and

other governmental charges, (2) such registration fees as may from time to time be in effect for the registration of transfers of Shares generally on the Share register of the Company or Foreign Registrar and applicable to transfers of Shares to or from the name of the Depositary or its nominee or the Custodian or its nominee on the making of deposits or withdrawals under the terms of the Deposit Agreement, (3) such cable, telex and facsimile transmission expenses as are expressly provided in the Deposit Agreement, (4) such expenses as are incurred by the Depositary in the conversion of foreign currency pursuant to Section 4.05 of the Deposit Agreement, (5) a fee of \$5.00 or less per 100 American Depositary Shares (or portion thereof) for the execution and delivery of Receipts pursuant to Section 2.03, 4.03 or 4.04 of the Deposit Agreement and the surrender of Receipts pursuant to Section 2.05 or 6.02 of the Deposit Agreement and (6) a fee for the distribution of securities pursuant to Section 4.02 of the Deposit Agreement, such fee being in an amount equal to the fee for the execution and delivery of American Depositary Shares referred to above which would have been charged as a result of the deposit of such securities (for purposes of this clause 7 treating all such securities as if they were Shares), but which securities are instead distributed by the Depositary to Owners.

The Depositary, subject to Article 8 hereof, may own and deal in any class of securities of the Company and its affiliates and in Receipts.

#### 8. LOANS AND PRE-RELEASE OF SHARES AND RECEIPTS.

The Depositary will lend neither the Shares held under the Deposit Agreement nor the Receipts, provided, however, that the Depositary reserves the right to execute and deliver Receipts prior to the receipt of Shares pursuant to Section 2.02 of the Deposit Agreement on the terms and conditions set forth below and in the Deposit Agreement. The Depositary may receive Receipts in lieu of Shares as settlement of the pre-release of a Receipt. Subject to the terms and conditions of this Deposit Agreement, the Pre-Release of Receipts may occur only if (i) Pre-released Receipts are fully collateralized (marked to market daily) with cash or U.S. government securities in an amount equal to not less than 100% of the market value of the Pre-Released Receipts held by the Depositary for the benefit of Owners (but such collateral shall not constitute Deposited Securities), (ii) each recipient of Pre-released Receipts agrees in writing with the Depositary that such recipient (a) owns such Shares, (b) assigns all beneficial right, title and interest therein to the Depositary, (c) holds such Shares for the account of the Depositary and (d) will deliver such Shares to the Custodian as soon as practicable and promptly upon demand therefor and (iii) all Pre-released Receipts evidence not more than 20% of all American Depositary Shares (excluding those evidenced by Pre-released Receipts) or such other percentage as the Company and the Depositary may from time to time agree in writing, of the total number of Shares represented by Receipts except to the extent, if any, that such limitation is exceeded solely because of the withdrawal of Deposited Securities subsequent to the execution and delivery of Pre-Released Receipts in compliance with such limitation. The Depositary will also set limits with respect to the number of Receipts and Shares involved in transactions to be done under the Deposit Agreement with anyone person on a case by case basis as it deems appropriate. The Depositary may retain for its own account any compensation re

## 9. TITLE TO RECEIPTS.

It is a condition of this Receipt and every successive Owner and Beneficial Owner of this Receipt by accepting or holding the same consents and agrees, that title to this Receipt when properly endorsed or accompanied by proper instruments of transfer, is transferable by delivery with the same effect as in the case of a negotiable instrument under the laws of New York; <u>provided</u>, <u>however</u>, that the Depositary, notwithstanding any notice to the contrary, may treat the person in whose name this Receipt is registered on the books of the Depositary as the absolute owner hereof for the purpose of determining the person entitled to distribution of dividends or other distributions or to any notice provided for in the Deposit Agreement or for all other purposes.

## 10. VALIDITY OF RECEIPT.

This Receipt shall not be entitled to any benefits under the Deposit Agreement or be valid or obligatory for any purpose, unless this Receipt shall have been executed by the Depositary by the manual signature of a duly authorized signatory of the Depositary; provided, however, that such signature may be a facsimile if a Registrar for the Receipts shall have been appointed and such Receipts are countersigned by the manual signature of a duly authorized officer of the Registrar.

#### 11. REPORTS; INSPECTION OF TRANSFER BOOKS.

The Company is subject to the periodic reporting requirements of the Securities Exchange Act of 1934 and accordingly files certain reports with the United States Securities and Exchange Commission (hereinafter called the "Commission"). Such reports and other information may be inspected and copied at public reference facilities maintained by the Commission located at the date hereof at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549. The Depositary will make available for inspection by Owners of Receipts at its Principal Office, any reports and communications, including any proxy soliciting material, received from the Company which are both (a) received by the Depositary as the holder of the Deposited Securities and (b) made generally available to the holders of such Deposited Securities by the Company. The Depositary will also, upon written request, send to Owners of Receipts copies of such reports when furnished by the Company pursuant to the Deposit Agreement. Any such reports and communications, including any such proxy soliciting material, furnished to the Depositary by the Company shall be furnished in English to the extent such materials are required to be translated into English pursuant to any regulations of the Commission.

The Depositary will keep books, at its Principal Office, for the registration of Receipts and transfers of Receipts which at all reasonable times shall be open for inspection by the Owners of Receipts provided that such inspection shall not be for the purpose of communicating with Owners of Receipts in the interest of a business or object other than the business of the Company or a matter related to the Deposit Agreement or the Receipts.

## 12. <u>DIVIDENDS AND DISTRIBUTIONS</u>.

Notwithstanding any rights under the Company's articles of association, dividends paid on the Deposited Securities that are not paid to the

Depositary or its nominee or Custodian in US Dollars will be paid by the Company in euro. Whenever the Depositary receives any cash dividend or other cash distribution on any Deposited Securities distribution in respect of any Deposited Securities (including, without limitation, any dividend paid to holders of Shares in a foreign currency, the Depositary will, if at the time of receipt thereof any amounts received in a foreign currency can in the judgment of the Depositary be converted on a reasonable basis into United States dollars transferable to the United States, and subject to the Deposit Agreement, convert such dividend or distribution into Dollars. Promptly after the settlement of such conversion or, in the case of any cash dividend or other cash distribution received by the Depositary in Dollars, the Depositary shall as promptly as practicable, distribute the amount thus received (net of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.09 of the Deposit Agreement) to the Owners of Receipts entitled thereto; provided, however, that in the event that the Company or the Depositary is required to withhold and does withhold from any cash dividend or other cash distribution in respect of any Deposited Securities an amount on account of taxes, the amount distributed to the Owners of the Receipts evidencing American Depositary Shares representing such Deposited Securities shall be reduced accordingly.

Subject to the provisions of Section 4.11 and 5.09 of the Deposit Agreement, whenever the Depositary receives any distribution other than a distribution described in Section 4.01, 4.03 or 4.04 of the Deposit Agreement, the Depositary will cause the securities or property received by it to be distributed to the Owners entitled thereto, in any manner that the Depositary may deem equitable and practicable for accomplishing such distribution; provided, however, that if in the reasonable opinion of the Depositary such distribution cannot be made proportionately among the Owners of Receipts entitled thereto, or if for any other reason the Depositary deems such distribution not to be feasible, the Depositary may, after consultation with the Company, adopt such method as it may deem equitable and practicable for the purpose of effecting such distribution, including, but not limited to, the public or private sale of the securities or property thus received, or any part thereof, and the net proceeds of any such sale (net of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.09 of the Deposit Agreement) will be distributed by the Depositary to the Owners of Receipts entitled thereto all in the manner and subject to the conditions described in Section 4.01 of the Deposit Agreement; provided, however, that no distribution to Owners pursuant to Section 4.02 of the Deposit Agreement shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any distribution of securities under this Section 4.02 if it has not received satisfactory assurances from the Company that the distribution does not require registration under the Securities Act of 1933.

If any distribution consists of a dividend in, or free distribution of, Shares, the Depositary may, and shall if the Company shall so request, distribute to the Owners of outstanding Receipts entitled thereto, additional Receipts evidencing an aggregate number of American Depositary Shares representing the amount of Shares received as such dividend or free distribution subject to the terms and conditions of the Deposit Agreement with respect to the deposit of Shares and the issuance of American Depositary Shares evidenced by Receipts, including the withholding of any tax or other governmental charge as provided in Section 4.11 of the Deposit Agreement and the payment of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.09 of the Deposit Agreement; provided, however, that no

distribution to Owners pursuant to this Section 4.03 shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any such distribution of Receipts if it has not received reasonably satisfactory assurances from the Company that such distribution does not require registration under the Securities Act of 1933 or is exempt from registration under the provisions of such Act. In lieu of delivering Receipts for fractional American Depositary Shares in any such case, the Depositary shall sell the amount of Shares represented by the aggregate of such fractions and distribute the net proceeds, all in the manner and subject to the conditions described in Section 4.01 of the Deposit Agreement. If additional Receipts are not so distributed, each American Depositary Share shall thenceforth also represent the additional Shares distributed upon the Deposited Securities represented thereby.

In the event that the Depositary determines that any distribution in property (including Shares and rights to subscribe therefor) is subject to any tax or other governmental charge which the Depositary is obligated to withhold, the Depositary may by public or private sale dispose of all or a portion of such property (including Shares and rights to subscribe therefor) in such amounts and in such manner as the Depositary reasonably deems necessary and practicable to pay any such taxes or charges, and the Depositary shall distribute the net proceeds of any such sale after deduction of such taxes or charges to the Owners of Receipts entitled thereto.

Subject to the rules and regulations of any stock exchange upon which the American Depositary Shares may be traded, the Depositary shall endeavor to convert the funds as promptly as practicable and that distributions to Owners pursuant to Section 4.01 of the Deposit Agreement are made within five New York Stock Exchange trading days of the day on which the cash dividend or cash distribution on the Deposited Securities is received by the Depositary.

#### 13. RIGHTS.

In the event that the Company shall offer or cause to be offered to the holders of any Deposited Securities any rights to subscribe for additional Shares or any rights of any other nature, the Depositary, after consultation with the Company, shall have discretion as to the procedure to be followed in making such rights available to any Owners or in disposing of such rights on behalf of any Owners and making the net proceeds available to such Owners or, if by the terms of such rights offering or for any other reason, the Depositary may not either make such rights available to any Owners or dispose of such rights and make the net proceeds available to such Owners, then the Depositary shall allow the rights to lapse. If at the time of the offering of any rights the Depositary, after consultation with the Company, determines in its reasonable discretion that it is lawful and feasible to make such rights available to all or certain Owners but not to other Owners, the Depositary may, and at the request of the Company shall, distribute to any Owner to whom it determines the distribution to be lawful and feasible, in proportion to the number of American Depositary Shares held by such Owner, warrants or other instruments therefor in such form as it deems appropriate.

In circumstances in which rights would otherwise not be distributed, if an Owner of Receipts requests the distribution of warrants or other instruments in order to exercise the

rights allocable to the American Depositary Shares of such Owner hereunder, the Depositary will make such rights available to such Owner upon written notice from the Company to the Depositary that (a) the Company has elected in its sole discretion to permit such rights to be exercised and (b) such Owner has executed such documents as the Company has determined in its sole discretion are reasonably required under applicable law.

If the Depositary has distributed warrants or other instruments for rights to all or certain Owners, then upon instruction from such an Owner pursuant to such warrants or other instruments to the Depositary from such Owner to exercise such rights, upon payment by such Owner to the Depositary for the account of such Owner of an amount equal to the purchase price of the Shares to be received upon the exercise of the rights, and upon payment of the fees and expenses of the Depositary and any other charges as set forth in such warrants or other instruments, the Depositary shall, on behalf of such Owner, exercise the rights and purchase the Shares, and the Company shall cause the Shares so purchased to be delivered to the Depositary on behalf of such Owner. As agent for such Owner, the Depositary will cause the Shares so purchased to be deposited pursuant to Section 2.02 of the Deposit Agreement, and shall, pursuant to Section 2.03 of the Deposit Agreement, execute and deliver Receipts to such Owner; provided, however, that in the case of a distribution pursuant to the second paragraph of this Article 13, such Receipts shall be legended in accordance with applicable U.S. laws, and shall be subject to the appropriate restrictions on sale, deposit, cancellation, and transfer under applicable United States laws.

If the Depositary determines that it is not lawful and feasible to make such rights available to all or certain Owners, it may sell the rights, warrants or other instruments in proportion to the number of American Depositary Shares held by the Owners to whom it has determined it may not lawfully or feasibly make such rights available, and allocate the net proceeds of such sales (net of the fees and expenses of the Depositary as provided in Section 5.09 of the Deposit Agreement and all taxes and governmental charges payable in connection with such rights and subject to the terms and conditions of the Deposit Agreement) for the account of such Owners otherwise entitled to such rights, warrants or other instruments, upon an averaged or other practical basis without regard to any distinctions among such Owners because of exchange restrictions or the date of delivery of any Receipt or otherwise.

The Depositary will not offer rights to Owners unless both the rights and the securities to which such rights relate are either exempt from registration under the Securities Act of 1933 with respect to a distribution to all Owners or are registered under the provisions of the Securities Act of 1933, provided, that nothing in the Deposit Agreement shall create any obligation on the part of the Company to file a registration statement with respect to such rights or underlying securities or to endeavor to have such a registration statement declared effective. If an Owner of Receipts requests the distribution of warrants or other instruments, notwithstanding that there has been no such registration under such Act, the Depositary shall not effect such distribution unless it has received an opinion from recognized counsel in the United States for the Company upon which the Depositary may rely that such distribution to such Owner is exempt from such registration. The Company will have no obligation under the Deposit Agreement to register such rights under the Securities Act of 1933.

The Depositary shall not be responsible for any failure to determine that it may be lawful or feasible to make such rights available to Owners in general or any Owner in particular.

## 14. CONVERSION OF FOREIGN CURRENCY.

Whenever the Depositary or the Custodian shall receive foreign currency, by way of dividends or other distributions or the net proceeds from the sale of securities, property or rights the Depositary shall, as promptly as practicable, convert or cause to be converted, by sale or in any other manner that it may reasonably determine, such foreign currency into Dollars, and such Dollars shall be distributed to the Owners entitled thereto or, if the Depositary shall have distributed any warrants or other instruments which entitle the holders thereof to such Dollars, then to the holders of such warrants or instruments upon surrender thereof for cancellation. Such distribution may be made upon an averaged or other practicable basis without regard to any distinctions among Owners on account of exchange restrictions, the date of delivery of any Receipt or otherwise and shall be net of any expenses of conversion into Dollars incurred by the Depositary as provided in Section 5.09 of the Deposit Agreement.

If such conversion or distribution can be effected only with the approval or license of any government or agency thereof, the Depositary shall file such application for approval or license, if any, as it may deem desirable.

If at any time the Depositary shall determine that in its reasonable judgment any foreign currency received by the Depositary or the Custodian is not convertible on a reasonable basis into Dollars transferable to the United States, or if any approval or license of any government or agency thereof which is required for such conversion is denied or in the reasonable opinion of the Depositary is not obtainable, or if any such approval or license is not obtained within a reasonable period as determined by the Depositary, the Depositary may distribute the foreign currency (or an appropriate document evidencing the right to receive such foreign currency) received by the Depositary to, or in its reasonable discretion may hold such foreign currency uninvested and without liability for interest thereon for the accounts of, the Owners entitled to receive the same.

If any such conversion of foreign currency, in whole or in part, cannot be effected for distribution to some of the Owners entitled thereto, the Depositary may in its discretion make such conversion and distribution in Dollars to the extent permissible to the Owners entitled thereto and may distribute the balance of the foreign currency received by the Depositary to, or hold such balance uninvested and without liability for interest thereon for the accounts of, the Owners entitled thereto.

## 15. RECORD DATES.

Whenever any cash dividend or other cash distribution shall become payable or any distribution other than cash shall be made, or whenever rights shall be issued with respect to the Deposited Securities, or whenever the Depositary shall receive notice of any meeting of holders of Shares or other Deposited Securities, or whenever for any reason the Depositary causes a change in the number of Shares that are represented by each American Depositary

Share, or whenever the Depositary shall find it necessary, the Depositary shall fix a record date (a) for the determination of the Owners of Receipts who shall be (i) entitled to receive such dividend, distribution or rights or the net proceeds of the sale thereof, (ii) entitled to give instructions for the exercise of voting rights at any such meeting or (iii) responsible for any fees or charges assessed by the Depositary pursuant to the Deposit Agreement, or (b) on or after which each American Depositary Share will represent the changed number of Shares, subject to the provisions of the Deposit Agreement. The record date shall be fixed in accordance with any applicable rules of the New York Stock Exchange. The Depositary shall advise the Company and the New York Stock Exchange of any record date so fixed by the Depositary.

## 16. VOTING OF DEPOSITED SECURITIES.

Upon the written request of an Owner of record of a Receipt as of the record date (the "Voting Record Date") received on or before the date established by the Depositary for such purpose (the "Instruction Date"), the Depositary will endeavor to cause the appointment (or, if the Deposited Securities are registered in the name of or held by its Custodian or a nominee, the Depositary hereby agrees to procure that the Custodian or its nominee shall cause the appointment), subject to the Articles of Association of the Company, such Owner as of the Voting Record Date fixed by the Depositary in accordance with Section 4.06 of the Deposit Agreement as a proxy in respect of any meeting (including any adjourned meeting) at which such Owner will be entitled to attend and vote at in respect of the Deposited Securities represented by the American Depositary Shares evidenced by the Receipts held by such Owner on the Voting Record Date. In respect of any such meeting each such Owner may appoint a person nominated by the Depositary, its Custodian or nominee, to vote on behalf of the Owner subject to and in accordance with the provisions of Section 4.07 of the Deposit Agreement and the Articles of Association of the Company. Upon receipt of notice of any meeting of holders of Shares or other Deposited Securities, if requested in writing by the Company the Depositary shall, as soon as practicable thereafter, mail to the Owners of record as of the Voting Record Date a notice, the form of which notice shall be approved by the Company which shall contain (a) such information as is contained in such notice of meeting, (b) a voting instruction card in the form prepared by the Depositary after consultation with the Company, (c) a statement that the Owners of record as of the close of business on the Voting Record Date will be entitled, subject to any applicable provision of English law and of the Memorandum and Articles of Association of the Company and the provisions of or governing the Deposited Securities, either (i) to use such voting instruction card to inform the Depositary that the Owner intends to attend such meeting as the proxy of the Depositary, the Custodian or its nominee (as appropriate) solely with respect to the Shares or other Deposited Securities represented by American Depositary Shares evidenced by such Owner's Receipts or (ii) to instruct such person nominated by the Depositary, the Custodian or nominee as to the exercise of the voting rights pertaining to the number of Shares or other Deposited Securities represented by the American Depositary Shares evidenced by such Owner's Receipts, and (d) a brief statement as to the manner in which voting instructions may be given to the person nominated by the Depositary. Upon the written request of an Owner of record of a Receipt on the Voting Record Date, received on or before the Instruction Date the Depositary shall endeavor, in so far as practicable, to vote or cause to be voted the number of Shares or other Deposited Securities represented by the American Depositary Shares evidenced by such Receipt in accordance with the instructions set forth in

such request. Neither the Depositary, nor the Custodian nor the nominee of either of them shall vote or attempt to exercise the right to vote that attaches to the Shares or other Deposited Securities, other than in accordance with such written instructions from Owners given in accordance with Section 4.07 of the Deposit Agreement. If no valid written instructions are received by the Depositary from an Owner with respect to any of the number of Deposited Securities represented by the American Depositary Shares evidenced by such Owner's Receipts on or before the Instruction Date, that number of Deposited Securities shall not be voted by the Depositary, the Custodian or the nominee of either of them.

There can be no assurance that Owners generally or any Owner in particular will receive the notice described in the preceding paragraph sufficiently prior to the Instruction Date to ensure that the Depositary will appoint the Owner as proxy or vote the Shares or other Deposited Securities in accordance with the provisions set forth in the preceding paragraph.

#### 17. CHANGES AFFECTING DEPOSITED SECURITIES.

In circumstances where the provisions of Section 4.03 of the Deposit Agreement do not apply, upon any change in nominal value, change in par value, split-up, consolidation, or any other reclassification of Deposited Securities, or upon any recapitalization, reorganization, merger or consolidation, or sale of assets affecting the Company or to which it is a party, any securities which shall be received by the Depositary or a Custodian in exchange for or in conversion of or in respect of Deposited Securities shall be treated as new Deposited Securities under the Deposit Agreement, and American Depositary Shares shall thenceforth represent, in addition to the existing Deposited Securities, the right to receive the new Deposited Securities so received in exchange or conversion, unless additional Receipts are delivered pursuant to the following sentence. In any such case the Depositary may, and shall if the Company shall so reasonably request, execute and deliver additional Receipts as in the case of a dividend in Shares, or call for the surrender of outstanding Receipts to be exchanged for new Receipts specifically describing such new Deposited Securities.

#### 18. LIABILITY OF THE COMPANY AND DEPOSITARY.

Neither the Depositary nor the Company nor any of their directors, employees, agents or affiliates shall incur any liability to any Owner or Beneficial Owner if, by reason of any provision of any present or future law or regulation of the United States or any other country, or of any other governmental or regulatory authority, or by reason of any provision, present or future, of the Memorandum and Articles of Association of the Company, or by reason of any provision of any securities issued or distributed by the Company, or any offering or distribution thereof, or by reason of any act of God or war or terrorism or other circumstances beyond its control, the Depositary or the Company shall be prevented, delayed or forbidden from or be subject to any civil or criminal penalty on account of doing or performing any act or thing which by the terms of the Deposit Agreement or Deposited Securities it is provided shall be done or performed; nor shall the Depositary or the Company or any of their directors, employees, agents or affiliates incur any liability to any Owner or Beneficial Owner of a Receipt by reason of any non-performance or delay, caused as aforesaid, in the performance of any act or thing which by the terms of the Deposit Agreement it is provided shall or may be done or performed, or by

reason of any exercise of, or failure to exercise, any discretion provided for in the Deposit Agreement. Where, by the terms of a distribution pursuant to Section 4.01, 4.02 or 4.03 of the Deposit Agreement, or an offering or distribution pursuant to Section 4.04 of the Deposit Agreement, such distribution or offering may not be made available to Owners of Receipts, and the Depositary may not dispose of such distribution or offering on behalf of such Owners and make the net proceeds available to such Owners, then the Depositary shall not make such distribution or offering, and shall allow any rights, if applicable, to lapse. Neither the Company nor the Depositary assumes any obligation or shall be subject to any liability under the Deposit Agreement to Owners or Beneficial Owners of Receipts, except that they agree to perform their obligations specifically set forth in the Deposit Agreement without negligence or bad faith. The Depositary shall not be subject to any liability with respect to the validity or worth of the Deposited Securities. Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be under any obligation to appear in or prosecute or defend any action, suit, or other proceeding in respect of any Deposited Securities or in respect of the Receipts on behalf of any Owner, Beneficial Owner or other person. The parties to the Deposit Agreement understand that the Custodian is not a party to the Deposit Agreement and, accordingly, the Custodian has no obligations whatsoever with respect to such proceedings, the responsibility of the Custodian being solely to the Depositary. Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be liable for any action or nonaction by it in reliance upon the advice of or information from legal counsel, accountants, any person presenting Shares for deposit, any Owner or Beneficial Owner of a Receipt, or any other person believed by it in good faith to be competent to give such advice or information. Each of the Depositary, the Company and their respective directors, officers, employees and agents may rely and shall be protected in acting upon any written notice, request, direction or other document believed by such person to be genuine and to have been signed or presented by the proper party or parties. The Depositary and its agents will not be responsible for (i) any failure to carry out any instructions to vote any of the Deposited Securities or for the manner in which any such vote is cast, in each case to the extent the Depositary or its agents act without gross negligence or willful misconduct or (ii) for the effect of any such vote. Notwithstanding anything to the contrary set forth in the Deposit Agreement or any Receipt, the Depositary and its agents may fully respond to any and all demands or requests for information maintained by or on its behalf in connection with the Deposit Agreement, any Owner or Owners, any Receipt or Receipts or otherwise related hereto to the extent such information is requested or required by or pursuant to any lawful authority, including without limitation laws, rules, regulations, administrative or judicial process, banking, securities or other regulators. The Depositary shall not be liable for any acts or omissions made by a successor depositary whether in connection with a previous act or omission of the Depositary or in connection with a matter arising wholly after the removal or resignation of the Depositary, provided that in connection with the issue out of which such potential liability arises, the Depositary performed its obligations without negligence or bad faith while it acted as Depositary. In connection with and in accordance with the arrangements and procedures relating to DRS/Profile, the parties understand that the Depositary will not verify, determine or otherwise ascertain that the DTC participant which is claiming to be acting on behalf of an Owner in requesting registration of transfer and delivery through DRS/Profile has the actual authority to act on behalf of the Owner. Each Owner agrees that neither the Depositary nor the Company shall have any liability for the

Depositary's or any of its agents' reliance upon the authority of any information in, nor for any of the Depositary's or any of its agents' compliance with directions from, any DTC participants as set forth above. Neither the Company nor the Depositary nor any of their respective agents shall be liable to Owners or holders of interests in American Depositary Shares or any other third party or parties for any indirect, special, punitive or consequential damages. No disclaimer of liability under the Securities Act of 1933 is intended by any provision of the Deposit Agreement.

## 19. RESIGNATION AND REMOVAL OF THE DEPOSITARY; APPOINTMENT OF SUCCESSOR CUSTODIAN.

The Depositary may at any time resign as Depositary by written notice of its election so to do delivered to the Company, such resignation to take effect upon the appointment of a successor depositary and its acceptance of such appointment as provided in the Deposit Agreement. The Depositary may at any time be removed by the Company by written notice to the Depositary, such removal, to become effective upon the appointment of a successor depositary and its acceptance of such appointment as provided in the Deposit Agreement. Whenever the Depositary in its reasonable discretion determines that it is in the best interest of the Owners of Receipts to do so, it may appoint a substitute or additional custodian or custodians, after consultation with the Company.

#### 20. AMENDMENT.

The form of the Receipts and any provisions of the Deposit Agreement may at any time and from time to time be amended by agreement between the Company and the Depositary without the consent of Owners or Beneficial Owners in any respect which they may deem necessary or desirable. Any amendment which shall impose or increase any fees or charges (other than taxes and other governmental charges, registration fees and cable, telex or facsimile transmission costs, delivery costs or other such expenses), or which shall otherwise prejudice any substantial existing right of Owners of Receipts, shall, however, not become effective as to outstanding Receipts until the expiration of thirty days after notice of such amendment shall have been given to the Owners of outstanding Receipts. Every Owner and Beneficial Owner of a Receipt at the time any amendment so becomes effective shall be deemed, by continuing to hold such Receipt or any interest therein, to consent and agree to such amendment and to be bound by the Deposit Agreement as amended thereby. In no event shall any amendment impair the right of the Owner of any Receipt to surrender such Receipt and receive therefor the Deposited Securities represented thereby except in order to comply with mandatory provisions of applicable law.

The Company and the Depositary shall each use their reasonable efforts to amend the Deposit Agreement as necessary to reflect changes in English or U.S. law and in the Company's Memorandum and Articles of Association.

## 21. TERMINATION OF DEPOSIT AGREEMENT.

The Depositary at any time at the direction of the Company, shall terminate the Deposit Agreement by mailing notice of such termination to the Owners of all Receipts then

outstanding at least 30 days prior to the date fixed in such notice for such termination. The Depositary may likewise terminate the Deposit Agreement by mailing notice of such termination to the Company and the Owners of all Receipts then outstanding if at any time 60 days shall have expired after the Depositary shall have delivered to the Company a written notice of its election to resign and a successor depositary shall not have been appointed and accepted its appointment as provided in the Deposit Agreement. On and after the date of termination, the Owner of a Receipt will, upon (a) surrender of such Receipt at the Principal Office of the Depositary and (b) payment of the fee of the Depositary for the surrender of Receipts referred to in section 2.05, and (c) payment of any applicable taxes or governmental charges, be entitled to delivery, to him or upon his order, of the amount of Deposited Securities represented by the American Depositary Shares evidenced by such Receipt. If any Receipts shall remain outstanding after the date of termination, the Depositary thereafter shall discontinue the registration of transfers of Receipts, shall suspend the distribution of dividends to the Owners thereof, and shall not give any further notices or perform any further acts under the Deposit Agreement, except that the Depositary shall continue to collect dividends and other distributions pertaining to Deposited Securities, shall sell rights and other property as provided in the Deposit Agreement, and shall continue to deliver Deposited Securities, together with any dividends or other distributions received with respect thereto and the net proceeds of the sale of any rights or other property, in exchange for Receipts surrendered to the Depositary (after deducting, in each case, the fee of the Depositary for the surrender of Receipts, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of the Deposit Agreement, and any applicable taxes or governmental charges) and for its obligation under Section 5.08 of the Deposit Agreement. At any time after the expiration of six months from the date of termination, the Depositary may sell the Deposited Securities then held under the Deposit Agreement and may thereafter hold uninvested the net proceeds of any such sale, together with any other cash then held by it thereunder, unsegregated and without liability for interest, for the pro rata benefit of the Owners of Receipts which have not theretofore been surrendered, such Owners thereupon becoming general creditors of the Depositary with respect to such net proceeds. After making such sale, the Depositary shall be discharged from all obligations under the Deposit Agreement, except to account for such net proceeds and other cash (after deducting, in each case, the fee of the Depositary for the surrender of a Receipt, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of the Deposit Agreement, and any applicable taxes or governmental charges) and for its obligation under Section 5.08 of the Deposit Agreement. Upon the termination of the Deposit Agreement, the Company shall be discharged from all obligations under the Deposit Agreement except for its obligations to the Depositary with respect to indemnification, charges, and expenses.

#### 22. COMPLIANCE WITH U.S. SECURITIES LAWS.

Notwithstanding anything in the Deposit Agreement to the contrary, the Company and the Depositary each agrees that it will not exercise any rights it has under the Deposit Agreement to permit the withdrawal or delivery of Deposited Securities in a manner which would violate the U.S. securities laws, including, but not limited to, Section I.A.(1) of the General Instructions to the Form F-6 registration statement, as amended from time to time, under the Securities Act of 1933.

#### 23. DISCLOSURE OF INTERESTS.

To the extent that provisions of or governing any Deposited Securities (including the Company's Memorandum and Articles of Association or applicable English law) may require the disclosure of beneficial or other ownership of Deposited Securities, other Shares and other securities to the Company and may provide for blocking transfer and voting or other rights to enforce such disclosure or limit such ownership, the Depositary shall, to the extent reasonably practicable, comply with the Company's instructions as to Receipts in respect of any such enforcement or limitation, and Owners and Beneficial Owners of Receipts shall comply with all such disclosure requirements and ownership limitations and shall cooperate with the Depositary's compliance with such Company instructions. The Company may from time to time request Owners to provide information (a) as to the capacity in which such Owners own or owned American Depositary Shares, (b) regarding the identity of any other persons then or previously interested in such American Depositary Shares and (c) regarding the nature of such interest and various other matters pursuant to applicable law or the Memorandum and Articles of Association of the Company or other such corporate document of the Company, all as if such American Depositary Shares were to the extent practicable the underlying Shares. Each Owner agrees to provide any information requested by the Company or the Depositary pursuant to this Section whether or not such person is still an Owner at the time of the request. The Depositary agrees to use reasonable efforts to comply with written instructions received from the Company requesting that the Depositary forward any such requests to Owners and to forward to the Company any responses to such requests received by the Depositary.

#### 24. ARBITRATION REQUIREMENT

The Company and, by holding a Receipt, each Owner, but not the Depositary or its agents hereunder, shall be bound by the arbitration and exclusive jurisdiction provisions set out in articles 152, 153 and 154 of the Articles of Association of the Company from time to time (the "Articles"), as if all references therein to "shareholder" were replaced with "ADR holder". Articles 152, 153 and 154 of the Articles shall accordingly be incorporated, *mutatis mutandis*, into the terms of this Deposit Agreement. A copy of articles 152, 153 and 154 of the Articles is available upon written request to the Company. All cross-references to the Articles in this Article 24 and in Section 7.07 of the Deposit Agreement will be updated and amended without further action of any party in the event the Articles themselves are renumbered. English law shall govern any such arbitration.

ROYAL DUTCH SHELL plc

and

THE BANK OF NEW YORK

As Depositary

and

OWNERS AND BENEFICIAL OWNERS OF AMERICAN DEPOSITARY RECEIPTS

Deposit Agreement

Dated as of May 19, 2005

# TABLE OF CONTENTS

|                             |   | Page     |
|-----------------------------|---|----------|
| ARTICLE 1.<br>SECTION       | DEFINITIONS   | 1        |
| 1.01.<br>SECTION            | American Depositary Shares.   | 1        |
| 1.02.<br>SECTION            | Articles.   | 1        |
| 1.03.<br>SECTION            | Beneficial Owner.   | 2        |
| 1.04.<br>SECTION            | Commission.   | 2        |
| 1.05.<br>SECTION            | Company.  | 2        |
| 1.06.<br>SECTION            | Custodian.  | 2        |
| 1.07.<br>SECTION            | Deliver; Surrender.   | 2        |
| 1.08.<br>SECTION            | Deposit Agreement.  | 2        |
| 1.09.<br>SECTION            | Depositary; Principal Office.   | 3        |
| 1.10.<br>SECTION            | Deposited Securities.   | 3        |
| 1.11.<br>SECTION            | Dollars; GBP.   | 3        |
| 1.12.<br>SECTION            | DTC.  | 3        |
| 1.13.<br>SECTION            | Foreign Registrar.  | 3        |
| 1.14.<br>SECTION            | Owner.  | 3        |
| 1.15.<br>SECTION            | Receipts; Direct Registration Receipts.   | 4        |
| 1.16.<br>SECTION            | Registrar.  | 4        |
| 1.17.<br>SECTION            | Restricted Securities.  | 4        |
| 1.18.<br>SECTION            | Securities Act of 1933.   | 4        |
| 1.19.<br>ARTICLE 2.         | Shares.   | 4        |
| SECTION                     | FORM OF RECEIPTS, DEPOSIT OF SHARES, EXECUTION AND DELIVERY, TRANSFER AND SURRENDER OF RECEIPTS | 5        |
| 2.01.                       | Form and Transferability of Receipts.   | 5        |
| SECTION<br>2.02.            | Deposit of Shares.  | 6        |
| SECTION 2.03.               | Execution and Delivery of Receipts.   | 6        |
| SECTION<br>2.04.            | Registration of Transfer of Receipts; Combination and Split-up of Receipts.                     | 7        |
| SECTION<br>2.05.            | Surrender of Receipts and Withdrawal of Shares.   | 8        |
| SECTION<br>2.06.            | Limitations on Execution and Delivery, Transfer and Surrender of Receipts.                      | 9        |
| SECTION 2.07.               | Lost Receipts, etc.   | 9        |
| SECTION<br>2.08.            | Cancellation and Destruction of Surrendered Receipts.   | 10       |
| SECTION 2.09.               | Pre-Release of Receipts.  | 10       |
| SECTION<br>2.10.            | Maintenance of Records.   | 10       |
| SECTION 2.11.<br>ARTICLE 3. | Profile Modification System. CERTAIN OBLIGATIONS OF OWNERS AND BENEFICIAL OWNERS OF RECEIPTS    | 11<br>11 |
| SECTION<br>3.01.            | Filing Proofs, Certificates and Other Information.  | 11       |
| SECTION<br>3.02.            | Liability of Owner or Beneficial Owner for Taxes.   | 12       |
| SECTION<br>3.03             | Warranties on Denosit of Shares   | 12       |

| SECTION    | Disclosure of Interests.                         | 12 |
|------------|--|----|
| 3.04.      |  |    |
| ARTICLE 4. | THE DEPOSITED SECURITIES                         | 13 |
| SECTION    |  |    |
| 4.01.      | Cash Distributions                               | 13 |
| SECTION    |  |    |
| 4.02.      | Distributions Other Than Cash, Shares or Rights. | 13 |
| SECTION    |  |    |
| 4.03.      | Distributions in Shares.                         | 14 |
| SECTION    |  |    |
| 4.04.      | Rights.  | 14 |
|            |  |    |

| CE CETON.                |  | Page     |
|--------------------------|--|----------|
| SECTION<br>4.05.         | Conversion of Foreign Currency.                                      | 16       |
| SECTION                  | Conversion of Foreign Currency.                                      | 10       |
| 4.06.                    | Fixing of Record Date.   | 17       |
| SECTION                  |  |          |
| 4.07.                    | Voting of Deposited Securities.                                      | 17       |
| SECTION                  |  |          |
| 4.08.                    | Changes Affecting Deposited Securities.                              | 18       |
| SECTION                  | P  | 10       |
| 4.09.<br>SECTION         | Reports.   | 18       |
| 4.10.                    | Lists of Owners.   | 19       |
| SECTION 4.11.            | Withholding.   | 19       |
| ARTICLE 5.               | THE DEPOSITARY, THE CUSTODIANS AND THE COMPANY                       | 19       |
| SECTION                  |  |          |
| 5.01.                    | Maintenance of Office and Transfer Books by the Depositary.          | 19       |
| SECTION                  |  |          |
| 5.02.                    | Prevention or Delay in Performance by the Depositary or the Company. | 20       |
| SECTION<br>5.03.         | Obligations of the Depositary, the Custodian and the Company.        | 20       |
| SECTION                  | Obligations of the Depositary, the Custodian and the Company.        | 20       |
| 5.04.                    | Resignation and Removal of the Depositary.                           | 22       |
| SECTION                  | Resignation and Removal of the Depositary.                           |          |
| 5.05.                    | The Custodians.  | 22       |
| SECTION                  |  |          |
| 5.06.                    | Notices and Reports.   | 23       |
| SECTION                  |  |          |
| 5.07.                    | Distribution of Additional Shares, Rights, etc.                      | 23       |
| SECTION<br>5.08.         | Indemnification.   | 24       |
| SECTION                  | indenninication.   | 24       |
| 5.09.                    | Charges of Depositary.   | 25       |
| SECTION                  |  |          |
| 5.10.                    | Retention of Depositary Documents.                                   | 25       |
| SECTION 5.11.            | Exclusivity.   | 26       |
| SECTION                  |  |          |
| 5.12.                    | List of Restricted Securities Owners.                                | 26       |
| ARTICLE 6.               | AMENDMENT AND TERMINATION Amendment.                                 | 26<br>26 |
| SECTION 6.01.<br>SECTION | Amendment.   | 20       |
| 6.02.                    | Termination.   | 26       |
| ARTICLE 7.               | MISCELLANEOUS  | 27       |
| SECTION                  |  |          |
| 7.01.                    | Counterparts.  | 27       |
| SECTION                  |  |          |
| 7.02.                    | No Third Party Beneficiaries.  | 28       |
| SECTION<br>7.03.         | Correspility   | 28       |
| SECTION                  | Severability.  | 20       |
| 7.04.                    | Owners and Beneficial Owners as Parties; Binding Effect.             | 28       |
| SECTION                  | o whels and Beneficial o whels as Landes, Binding Birect             | _5       |
| 7.05.                    | Notices.   | 28       |
| SECTION                  |  |          |
| 7.06.                    | Compliance with U.S. Securities Laws.                                | 29       |
| SECTION                  | A Province of Drive to   | 22       |
| 7.07.<br>SECTION         | Arbitration of Disputes.   | 29       |
| 7.08.                    | Governing Law.   | 29       |
| 7.00.                    | GOVERNME LUW.  | 29       |

Form of Receipt

#### DEPOSIT AGREEMENT

**DEPOSIT AGREEMENT** dated as of May 19, 2005, among ROYAL DUTCH SHELL plc, incorporated under the laws of England and Wales (herein called the Company), THE BANK OF NEW YORK, a New York banking corporation (herein called the Depositary), and all Owners and Beneficial Owners from time to time of American Depositary Shares issued hereunder.

#### WITNESSETH:

**WHEREAS**, the Company desires to provide, as hereinafter set forth in this Deposit Agreement, for the deposit of Shares (as hereinafter defined) of the Company from time to time with the Depositary or with the Custodian (as hereinafter defined) as agent of the Depositary for the purposes set forth in this Deposit Agreement, for the creation of American Depositary Shares representing the Shares so deposited and for the execution and delivery of American Depositary Receipts evidencing the American Depositary Shares; and

**WHEREAS**, the American Depositary Receipts are to be substantially in the form of Exhibit A annexed hereto, with appropriate insertions, modifications and omissions, as hereinafter provided in this Deposit Agreement;

**NOW**, **THEREFORE**, in consideration of the premises, it is agreed by and among the parties hereto as follows:

## **ARTICLE 1. DEFINITIONS**

The following definitions shall for all purposes, unless otherwise clearly indicated, apply to the respective terms used in this Deposit Agreement:

SECTION 1.01. <u>American Depositary Shares.</u>

The term "American Depositary Shares" shall mean the securities representing the interests in the Deposited Securities and evidenced by the Receipts issued hereunder. Each American Depositary Share shall represent the number of Shares specified in Exhibit A annexed hereto, until there shall occur a distribution upon Deposited Securities covered by Section 4.03 or a change in Deposited Securities covered by Section 4.08 with respect to which additional Receipts are not executed and delivered, and thereafter American Depositary Shares shall represent the amount of Shares or Deposited Securities specified in such Sections.

SECTION 1.02. Articles.

The term "Articles" shall mean the articles of association from time to time of the Company.

#### SECTION 1.03. Beneficial Owner.

The term "Beneficial Owner" shall mean each person owning from time to time any beneficial interest in the American Depositary Shares evidenced by any Receipt.

## SECTION 1.04. Commission.

The term "Commission" shall mean the Securities and Exchange Commission of the United States or any successor governmental agency in the United States.

## SECTION 1.05. Company.

The term "Company" shall mean Royal Dutch Shell plc, incorporated under the laws of England and Wales, and its successors.

## SECTION 1.06. Custodian.

The term "Custodian" shall mean each of the principal London office of The Bank of New York, as agent of the Depositary for the purposes of this Deposit Agreement, and any other firm or corporation which may hereafter be appointed by the Depositary pursuant to the terms of Section 5.05, as substitute or additional custodian or custodians hereunder, as the context shall require and shall also mean all of them collectively.

#### SECTION 1.07. Deliver; Surrender.

- (a) The term "deliver", or its noun form, when used with respect to Shares shall mean (i) one or more book-entry transfers to an account or accounts maintained with an institution authorized under the laws of England and Wales to effect book-entry transfers of such securities or (ii) to the extent applicable, the physical transfer of certificates evidencing Shares.
- (b) The term "deliver", or its noun form, when used with respect to Receipts, shall mean (i) registration of American Depositary Shares on the Direct Registration System in the name of, or one or more book-entry transfers of American Depositary Shares to an account or accounts at DTC designated by, the person entitled to such delivery or (ii) if requested by the person entitled to such delivery at the Principal Office of the Depositary of one or more Receipts registered in the name requested by the person entitled to such delivery.
- (c) The term "surrender", when used with respect to Receipts, shall mean (i) delivery to the Depositary's Principal Office of an instruction from an Owner surrendering American Depositary Shares on the Direct Registration System to the Depositary, (ii) one or more book-entry transfers of American Depositary Shares to the DTC account of the Depositary or (ii) surrender to the Depositary at its Principal Office of one or more Receipts.

## SECTION 1.08. Deposit Agreement.

The term "Deposit Agreement" shall mean this Deposit Agreement, as the same may be amended from time to time in accordance with the provisions of this Deposit Agreement.

## SECTION 1.09. Depositary; Principal Office.

The term "Depositary" shall mean The Bank of New York, a New York banking corporation, and any successor as depositary hereunder. The term "Principal Office", when used with respect to the Depositary, shall mean the office of the Depositary which at the date of this Agreement is 101 Barclay Street, New York, New York 10286.

# SECTION 1.10. <u>Deposited Securities.</u>

The term "Deposited Securities" as of any time shall mean Shares at such time deposited or deemed to be deposited under this Deposit Agreement and any and all other securities, property and cash received by the Depositary or the Custodian in respect thereof and at such time held hereunder, subject as to cash to the provisions of Section 4.05.

## SECTION 1.10. <u>Direct Registration System.</u>

"<u>Direct Registration System</u>" means the system for the uncertificated registration of ownership of securities established by DTC and utilized by the Depositary pursuant to which the Depositary may record the ownership of American Depositary Shares without the issuance of a Receipt, which ownership shall be evidenced by periodic statements issued by the Depositary to the Owner. For purposes hereof, the Direct Registration System shall include access to the Profile Modification System maintained by DTC which provides for automated transfer of record ownership between DTC and other Owners.

## SECTION 1.11. Dollars; GBP.

The term "Dollars" shall mean United States dollars. The term "GBP" shall mean United Kingdom pounds.

#### SECTION 1.12. DTC.

The term "DTC" shall mean The Depository Trust Company, or its successor.

# SECTION 1.13. Foreign Registrar.

The term "Foreign Registrar" shall mean the entity that presently carries out the duties of registrar for the Shares or any successor as registrar for the Shares and any other appointed agent of the Company for the transfer and registration of Shares.

## SECTION 1.14. Owner.

The term "Owner" shall mean the person in whose name a Receipt is registered on the books of the Depositary maintained for such purpose.

## SECTION 1.15. Receipts; Direct Registration Receipts.

The term "Receipts" shall mean the American Depositary Receipts issued hereunder, including Pre-Released Receipts, evidencing American Depositary Shares. The term "<u>Direct Registration Receipts</u>" shall mean American Depositary Shares not evidenced by Receipts that are evidenced by book entry notations recorded on the Direct Registration System and by periodic statements from the Depositary. References to "Receipts" shall include Direct Registration Receipts, unless the context otherwise requires.

## SECTION 1.16. Registrar.

The term "Registrar" shall mean any bank or trust company having an office in the Borough of Manhattan, The City of New York, which shall be appointed, after consultation with the Company, to register Receipts and transfers of Receipts as herein provided.

#### SECTION 1.17. Restricted Securities.

The term "Restricted Securities" shall mean Shares, or Receipts representing such Shares, which are acquired directly or indirectly from the Company, or any affiliate (as defined in Rule 144 under the Securities Act of 1933) of the Company, in a transaction or chain of transactions not involving any public offering, or which are held by an officer, director (or persons performing similar functions) or other affiliate of the Company, or which would require registration under the Securities Act of 1933 in connection with the public offer and sale thereof in the United States, or which are subject to other restrictions on sale or deposit under the laws of the United States or England and Wales, or under a shareholder agreement or the Memorandum or Articles of Association of the Company.

## SECTION 1.18. Securities Act of 1933.

The term "Securities Act of 1933" shall mean the United States Securities Act of 1933, as from time to time amended.

#### SECTION 1.19. Shares.

The term "Shares" shall mean Class B ordinary shares in registered form of the Company heretofore validly issued and outstanding and fully paid, nonassessable and that were not issued in violation of any pre-emptive or similar rights of the holders of outstanding Shares or hereafter validly issued and outstanding and fully paid, nonassessable and that are not issued in violation of any pre-emptive or similar rights of the holders of outstanding Shares; provided, however, that, if there shall occur any change in nominal value, a split-up or consolidation or any other reclassification or, upon the occurrence of an event described in Section 4.08, an exchange or conversion in respect of the Shares of the Company, the term "Shares" shall thereafter also mean the successor securities resulting from such change in nominal value, split-up or consolidation or such other reclassification or such exchange or conversion. Shares may be certificated or uncertificated.

## ARTICLE 2. FORM OF RECEIPTS, DEPOSIT OF SHARES, EXECUTION AND DELIVERY, TRANSFER AND SURRENDER OF RECEIPTS

SECTION 2.01. Form and Transferability of Receipts.

Definitive Receipts shall be substantially in the form set forth in Exhibit A annexed to this Deposit Agreement, with appropriate insertions, modifications and omissions, as hereinafter provided. No Receipt shall be entitled to any benefits under this Deposit Agreement or be valid or obligatory for any purpose, unless such Receipt shall have been either issued through the Direct Registration System or executed by the Depositary by the manual signature of a duly authorized signatory of the Depositary; provided, however, that such signature may be a facsimile if a Registrar for the Receipts shall have been appointed and such Receipts are countersigned by the manual signature of a duly authorized officer of the Registrar. The Depositary shall maintain books on which each Receipt so issued, executed and/or delivered as hereinafter provided and the transfer of each such Receipt shall be registered. Receipts bearing the manual or facsimile signature of a duly authorized signatory of the Depositary who was at any time a proper signatory of the Depositary shall bind the Depositary, notwithstanding that such signatory has ceased to hold such office prior to the execution and delivery of such Receipts by the Registrar or did not hold such office on the date of issuance of such Receipts.

The Receipts may be endorsed with or have incorporated in the text thereof such legends or recitals or modifications not inconsistent with the provisions of this Deposit Agreement as may be required by the Depositary or required to comply with any applicable law or regulations thereunder or with the rules and regulations of any securities exchange upon which American Depositary Shares may be listed or to conform with any usage with respect thereto, or to indicate any special limitations or restrictions to which any particular Receipts are subject by reason of the date of issuance of the underlying Deposited Securities or otherwise.

Notwithstanding anything in this Deposit Agreement or in the Receipts to the contrary, American Depositary Shares shall be evidenced by Direct Registration Receipts or by a global Receipt registered in the name of a nominee of The Depository Trust Company, unless certificated Receipts are specifically requested by a person entitled to delivery of Receipts. Owners shall be bound by the terms and conditions of this Deposit Agreement and of the form of Receipt, regardless of whether their Receipts are Direct Registration Receipts or certificated Receipts.

American Depositary Shares evidenced by a Receipt, when properly endorsed or accompanied by proper instruments of transfer, shall be transferable as certificated securities in registered form under the laws of the State of New York. American Depositary Shares not evidenced by a Receipt (also referred to as Direct Registration Receipts in this Deposit Agreement) shall be transferable as uncertificated securities in registered form under the laws of the State of New York. The Depositary, notwithstanding any notice to the contrary, may treat the Owner of American Depositary Shares as the absolute owner thereof for the purpose of determining the person entitled to distribution of dividends or other distributions or to any notice provided for in this Deposit Agreement and for all other purposes.

#### SECTION 2.02. Deposit of Shares.

Subject to the terms and conditions of this Deposit Agreement, Shares or evidence of rights to receive Shares may be deposited by delivery thereof to any Custodian hereunder, accompanied by any appropriate instrument or instruments of transfer, or endorsement, in form satisfactory to the Custodian, together with all such certifications as may be required by the Depositary or the Custodian in accordance with the provisions of this Deposit Agreement, and, if the Depositary requires, together with a written order directing the Depositary to execute and deliver to, or upon the written order of, the person or persons stated in such order, a Receipt or Receipts for the number of American Depositary Shares representing such deposit. No Share shall be accepted for deposit unless accompanied by evidence satisfactory to the Depositary that any necessary approval, where relevant, has been granted by any governmental body in England and Wales that is then performing the function of the regulation of currency exchange. If required by the Depositary, Shares presented for deposit at any time, whether or not the transfer books of the Company or the Foreign Registrar, if applicable, are closed, shall also be accompanied by an agreement or assignment, or other instrument satisfactory to the Depositary, which will provide for the prompt transfer to the Custodian of any dividend, or right to subscribe for additional Shares or to receive other property which any person in whose name the Shares are or have been recorded may thereafter receive upon or in respect of such deposited Shares, or in lieu thereof, such agreement of indemnity or other agreement as shall be satisfactory to the Depositary.

In the case of certificated Shares, at the request and risk and expense of any person proposing to deposit Shares, and for the account of such person, the Depositary may receive certificates for Shares to be deposited, together with the other instruments herein specified, for the purpose of forwarding such Share certificates to the Custodian for deposit hereunder.

In the case of certificated Shares, upon each delivery to a Custodian of a certificate or certificates evidencing Shares to be deposited hereunder, together with the other documents above specified, such Custodian shall be required by the Depositary, as soon as transfer and recordation can be accomplished, to present such certificate or certificates to the Company or the Foreign Registrar, if applicable, for transfer and recordation of the Shares being deposited in the name of the Depositary or its nominee or such Custodian or its nominee.

Deposited Securities shall be held by the Depositary or by a Custodian for the account and to the order of the Depositary or at such other place or places as the Depositary shall determine. The Depositary shall provide written notice informing the Company of any such other place or places.

## SECTION 2.03. Execution and Delivery of Receipts.

Upon receipt by any Custodian of any deposit pursuant to Section 2.02 hereunder (and in addition, if the transfer books of the Company or the Foreign Registrar, if applicable, are open, the Depositary may in its sole discretion require a proper acknowledgment or other evidence from the Company that any Deposited Securities have been recorded upon the books of

the Company or the Foreign Registrar, if applicable, in the name of the Depositary or its nominee or such Custodian or its nominee), together with the other documents required as above specified, such Custodian shall be required by the Depositary to notify the Depositary of such deposit and the person or persons to whom or upon whose written order a Receipt or Receipts are deliverable in respect thereof and the number of American Depositary Shares to be evidenced thereby. Such notification shall be required to be made by letter or, at the request, risk and expense of the person making the deposit, by cable, telex or facsimile transmission. Upon receiving such notice from such Custodian, or upon the receipt of Shares by the Depositary, the Depositary, subject to the terms and conditions of this Deposit Agreement, shall execute and deliver at its Principal Office, to or upon the order of the person or persons entitled thereto, a Receipt or Receipts, registered in the name or names and evidencing any authorized number of American Depositary Shares requested by such person or persons, but only upon payment to the Depositary of the fees and expenses of the Depositary for the execution and delivery of such Receipt or Receipts as provided in Section 5.09, and of all taxes and governmental charges and fees payable in connection with such deposit and the transfer of the Deposited Securities.

## SECTION 2.04. Registration of Transfer of Receipts; Combination and Split-up of Receipts.

The Depositary, subject to the terms and conditions of this Deposit Agreement, shall without unreasonable delay, register transfers of Receipts on its transfer books, upon any surrender of a Receipt, by the Owner in person or by a duly authorized attorney, properly endorsed or accompanied by proper instruments of transfer or, in the case of Direct Registration Receipts, upon receipt of a proper instruction, and duly stamped as may be required by the laws of the State of New York and of the United States of America. Thereupon the Depositary shall execute a new Receipt or Receipts and deliver the same to or upon the order of the person entitled thereto.

The Depositary, subject to the terms and conditions of this Deposit Agreement, shall upon surrender of a Receipt or Receipts for the purpose of effecting a split-up or combination of such Receipt or Receipts, execute and deliver a new Receipt or Receipts for any authorized number of American Depositary Shares requested, evidencing the same aggregate number of American Depositary Shares as the Receipt or Receipts surrendered. At the request of an Owner, the Depositary shall, for the purpose of substituting a certificated Receipt with a Direct Registration Receipts, or vice versa, deliver a certificated Receipt or Direct Registration Receipts, as the case may be, for any authorized number of American Depositary Shares requested, evidencing the same aggregate number of American Depositary Shares as those evidenced by the certificated Receipt or Direct Registration Receipts, as the case may be, surrendered for substitution.

The Depositary may appoint one or more co-transfer agents for the purpose of effecting transfers, combinations and split-ups of Receipts at designated transfer offices on behalf of the Depositary. In carrying out its functions, a co-transfer agent may require evidence of authority and compliance with applicable laws and other requirements by Owners or persons

entitled to Receipts and will be entitled to protection and indemnity to the same extent as the Depositary.

#### SECTION 2.05. Surrender of Receipts and Withdrawal of Shares.

Upon surrender at the Principal Office of the Depositary of a Receipt for the purpose of withdrawal of the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, and upon payment of the fee of the Depositary for the surrender of Receipts as provided in Section 5.09 and payment of all taxes and governmental charges payable in connection with such surrender and withdrawal of the Deposited Securities, and subject to the terms and conditions of this Deposit Agreement and English law, the Owner of such Receipt shall be entitled to delivery, to him or upon his order, of the amount of Deposited Securities at the time represented by the American Depositary Shares evidenced by such Receipt. Delivery of such Deposited Securities may be made (a) in the case of certificated Shares, by delivery of certificates in the name of such Owner or as ordered by him or certificates properly endorsed or accompanied by proper instruments of transfer to such Owner or as ordered by him and (b) in the case of any securities cash or other property to which such Owner is then entitled in respect of such Receipt, by delivery of those securities or that cash or other property to such Owner or as ordered by him. Such delivery shall be made, as hereinafter provided, without unreasonable delay.

A Receipt surrendered for such purposes may be required by the Depositary to be properly endorsed in blank or accompanied by proper instruments of transfer in blank (in the case of a certificated Receipt), or surrendered pursuant to a proper instruction (in the case of Direct Registration Receipts), and if the Depositary so requires, the Owner thereof shall execute and deliver to the Depositary a written order directing the Depositary to cause the Deposited Securities being withdrawn to be delivered to or upon the written order of a person or persons designated in such order. Thereupon the Depositary shall direct the Custodian to deliver at the office of such Custodian, subject to Sections 2.06, 3.01 and 3.02 and to the other terms and conditions of this Deposit Agreement and English law, to or upon the written order of the person or persons designated in the order delivered to the Depositary as above provided, the amount of Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, except that the Depositery may make delivery to such person or persons at the Principal Office of the Depositary of any dividends or distributions with respect to the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, or of any proceeds of sale of any dividends, distributions or rights, which may at the time be held by the Depositary.

At the request, risk and expense of any Owner so surrendering a Receipt, and for the account of such Owner, the Depositary shall direct the Custodian to forward any cash or other property (other than rights) comprising, and forward a certificate or certificates and other proper documents of title, if any, for, the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt to the Depositary for delivery at the Principal Office of the Depositary. Such direction shall be given by letter or, at the request, risk and expense of such Owner, by cable, telex or facsimile transmission. Rights, if any, shall be delivered to such Owner pursuant to Section 4.04.

## SECTION 2.06. <u>Limitations on Execution and Delivery, Transfer and Surrender of Receipts.</u>

As a condition precedent to the execution and delivery, registration of transfer, split-up, combination or surrender of any Receipt or withdrawal of any Deposited Securities, the Depositary, Custodian or Registrar may require payment from the depositor of Shares or the presenter of the Receipt of a sum sufficient to reimburse it for any tax or other governmental charge and any stock transfer or registration fee with respect thereto (including any such tax or charge and fee with respect to Shares being deposited or withdrawn) and payment of any applicable fees as herein provided, may require the production of proof satisfactory to it as to the identity and genuineness of any signature and may also require compliance with any regulations the Depositary may establish consistent with the provisions of this Deposit Agreement, including, without limitation, this Section 2.06.

The delivery of Receipts against deposit of Shares generally or against deposit of particular Shares may be suspended, or the transfer of Receipts in particular instances may be refused, or the registration of transfer of outstanding Receipts generally may be suspended, during any period when the transfer books of the Depositary are closed, or if any such action is deemed necessary or advisable by the Depositary or the Company at any time or from time to time because of any requirement of law or of any government or governmental body or commission, or under any provision of this Deposit Agreement, or for any other reason, subject to the provisions of Section 7.07 hereof. The Depositary shall not knowingly accept for deposit under this Deposit Agreement any Shares which would be required to be registered under the provisions of the Securities Act of 1933 for the public offer and sale thereof in the United States unless a registration statement is in effect as to such Shares for such offer and sale. The Depositary will use reasonable efforts to comply with written instructions of the Company that the Depositary shall not accept for the deposit hereunder any Shares identified in such circumstances as may reasonably be specified in such instructions to facilitate the Company's compliance with the U.S. securities laws.

Notwithstanding anything to the contrary in this Deposit Agreement, Owners shall be entitled to surrender Receipts and withdraw Deposited Securities as provided in Section 2.05 at any time, subject only to (i) temporary delays caused by closing the transfer books of the Depositary or the Company or the deposit of Shares in connection with voting at a shareholders' meeting, or the payment of dividends, (ii) the payment of fees, taxes and similar charges, (iii) compliance with any U.S. or foreign laws or governmental regulations relating to the Receipts or to the withdrawal of the Deposited Securities and (iv) any other reason that may at any time be specified in paragraph I.(A)(1) of the General Instructions to Form F-6 under the Securities Act of 1933, from time to time in effect, or any successor provision thereto.

## SECTION 2.07. Lost Receipts, etc.

In case any Receipt shall be mutilated, destroyed, lost or stolen, the Depositary shall deliver a new Receipt through the Direct Registration system or execute and deliver a new Receipt of like tenor in exchange and substitution for such mutilated Receipt upon cancellation thereof, or in lieu of and in substitution for such destroyed, lost or stolen Receipt. Before the

Depositary shall execute and deliver a new Receipt in substitution for a destroyed, lost or stolen Receipt, the Owner thereof shall have (a) filed with the Depositary (i) a request for such execution and delivery before the Depositary has notice that the Receipt has been acquired by a bona fide purchaser and (ii) a sufficient indemnity bond and (b) satisfied any other reasonable requirements imposed by the Depositary.

#### SECTION 2.08. Cancellation and Destruction of Surrendered Receipts.

All Receipts surrendered to the Depositary shall be canceled by the Depositary. The Depositary is authorized to destroy Receipts so canceled subject to Section 2.10.

#### SECTION 2.09. Pre-Release of Receipts.

The Depositary will lend neither the Shares held under this Deposit Agreement nor the Receipts. The Depositary reserves the right to execute and deliver Receipts prior to the receipt of Shares pursuant to Section 2.02 on the terms and conditions set forth below (a "Pre-Release"). The Depositary may receive Receipts in lieu of Shares as settlement of the Pre-Release of a Receipt. Subject to the terms and conditions of this Deposit Agreement, the Pre-Release of Receipts may occur only if (i) Pre-released Receipts are fully collateralized (marked to market daily) with cash or U.S. government securities in an amount equal to not less than 100% of the market value of the Pre-Released Receipts held by the Depositary for the benefit of Owners (but such collateral shall not constitute Deposited Securities), (ii) each recipient of Pre-released Receipts agrees in writing with the Depositary that such recipient (a) owns such Shares, (b) assigns all beneficial right, title and interest therein to the Depositary, (c) holds such Shares for the account of the Depositary and (d) will deliver such Shares to the Custodian as soon as practicable and promptly upon demand therefor and (iii) all Pre-released Receipts evidence not more than 20% of all American Depositary Shares (excluding those evidenced by Pre-released Receipts) and all Pre-Released Receipts evidence American Depositary Shares representing not more than 1% of all Shares outstanding or such other percentage of American Depositary Shares or Shares, as the case may be, as the Company and the Deposited Securities subsequent to the execution and delivery of Pre-Released Receipts in compliance with such limitation. The Depositary will also set limits with respect to the number of American Depositary Shares and Shares involved in transactions to be done hereunder with any one person on a case by case basis as it deems appropriate.

The Depositary may retain for its own account any compensation received by it in connection with the foregoing.

#### SECTION 2.10. Maintenance of Records.

The Depositary agrees to maintain or cause its agents to maintain records of all Receipts surrendered and Deposited Securities withdrawn under Section 2.05, substitute Receipts delivered under Section 2.07, and of cancelled or destroyed Receipts under Section 2.08, in

keeping with procedures ordinarily followed by stock transfer agents located in the City of New York or as required by laws or regulations governing the Depositary. The Depositary shall provide full access to such records to the Company and its agents from time to time during normal business hours upon the reasonable written request of the Company.

## SECTION 2.11. Profile Modification System.

- (a) Notwithstanding the provisions of Section 2.04, the parties acknowledge the Profile Modification System ("Profile") shall apply to Direct Registration Receipts upon acceptance thereof by DTC. Profile is a required feature of the Direct Registration System that allows a DTC participant, claiming to act on behalf of the Owner of Direct Registration Receipts, to direct the Depositary to register a transfer of the American Depositary Shares to DTC or its nominee and to deliver those American Depositary Shares to the DTC account of that DTC participant without receipt by the Depositary of prior authorization from the Owner to register such transfer.
- (b) In connection with and in accordance with the arrangements and procedures relating to Profile, the parties understand that the Depositary will not verify, determine or otherwise ascertain that the DTC participant that is claiming to be acting on behalf of an Owner in requesting a registration of transfer and delivery described in subsection (a) has the actual authority to act on behalf of the Owner. Each Owner agrees that neither the Company nor the Depositary shall have any liability for the Depositary's reliance upon information, or compliance with directions, it receives from a DTC participant as set forth in subsection (a) above.

## ARTICLE 3. CERTAIN OBLIGATIONS OF OWNERS AND BENEFICIAL OWNERS OF RECEIPTS

#### SECTION 3.01. Filing Proofs, Certificates and Other Information.

Any person presenting Shares for deposit or any Owner or Beneficial Owner of a Receipt may be required from time to time to file with the Depositary or the Custodian such proof of citizenship or residence, exchange control approval, or such information relating to the registration on the books of the Company or the Foreign Registrar, if applicable, to execute such certificates and to make such representations and warranties, as the Depositary may deem necessary or proper. The Depositary may withhold the delivery or registration of transfer of any Receipt or the distribution of any dividend or sale or distribution of rights or of the proceeds thereof or the delivery of any Deposited Securities until such proof or other information is filed or such certificates are executed or such representations and warranties made. The Depositary shall provide copies thereof to the Company as promptly as practicable upon written request by the Company, to the extent that disclosure is permitted under applicable law.

## SECTION 3.02. Liability of Owner or Beneficial Owner for Taxes.

If any tax or other governmental charge shall become payable by the Custodian or the Depositary with respect to any Receipt or any Deposited Securities represented by any Receipt, such tax or other governmental charge shall be payable by the Owner or Beneficial Owner of such Receipt to the Depositary. The Depositary may refuse to effect any transfer of such Receipt or any withdrawal of Deposited Securities represented by American Depositary Shares evidenced by such Receipt until such payment is made, and may withhold any dividends or other distributions, or may sell for the account of the Owner or Beneficial Owner thereof any part or all of the Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, and may apply such dividends or other distributions or the proceeds of any such sale in payment of such tax or other governmental charge and the Owner or Beneficial Owner of such Receipt shall remain liable for any deficiency. Neither the Company nor the Depositary shall be liable for failure of an Owner to comply with applicable tax laws or governmental charges.

## SECTION 3.03. Warranties on Deposit of Shares.

Every person depositing Shares under this Deposit Agreement shall be deemed thereby to represent and warrant that such Shares and each certificate therefor are validly issued, fully paid, nonassessable and were not issued in violation of any preemptive or similar rights of the holders of outstanding Shares and that the person making such deposit is duly authorized so to do. Every such person shall also be deemed to represent that such Shares are not, and American Depositary Shares representing such Shares would not be, Restricted Securities. Such representations and warranties shall survive the deposit of Shares and delivery of Receipts.

#### SECTION 3.04. Disclosure of Interests.

To the extent that provisions of or governing any Deposited Securities (including the Company's Memorandum and Articles of Association or applicable English law) may require the disclosure of beneficial or other ownership of Deposited Securities, other Shares and other securities to the Company and may provide for blocking transfer and voting or other rights to enforce such disclosure or limit such ownership, the Depositary shall, to the extent reasonably practicable, comply with the Company's instructions as to Receipts in respect of any such enforcement or limitation, and Owners and Beneficial Owners of Receipts shall comply with all such disclosure requirements and ownership limitations and shall cooperate with the Depositary's compliance with such Company instructions. The Company may from time to time request Owners to provide information (a) as to the capacity in which such Owners own or owned American Depositary Shares, (b) regarding the identity of any other persons then or previously interested in such American Depositary Shares and (c) regarding the nature of such interest and various other matters pursuant to applicable law or the Memorandum and Articles of Association of the Company or other such corporate document of the Company, all as if such American Depositary Shares were to the extent practicable the underlying Shares. Each Owner agrees to provide any information requested by the Company or the Depositary pursuant to this Section whether or not such person is still an Owner at the time of the request. The Depositary

agrees to use reasonable efforts to comply with written instructions received from the Company requesting that the Depositary forward any such requests to Owners and to forward to the Company any responses to such requests received by the Depositary.

### ARTICLE 4. THE DEPOSITED SECURITIES

### SECTION 4.01. Cash Distributions

Notwithstanding any rights under the Company's articles of association, any dividends paid on the Deposited Securities that are not paid to the Depositary or its nominee in Dollars will be paid by the Company in GBP. Whenever the Depositary shall receive any cash dividend or other cash distribution in respect of any Deposited Securities, the Depositary shall, subject to the provisions of Section 4.05 in the case of a dividend and/or distribution received in a currency other than Dollars, convert the amounts so received into Dollars. Promptly after the settlement of such conversion or, in the in the case of any cash dividend or other cash distribution received by the Depositary in Dollars, the Depositary shall as promptly as practicable, distribute the amount thus received (net of the fees and expenses of the Depositary as provided in Section 5.09) to the Owners entitled thereto, in proportion to the number of American Depositary Shares representing such Deposited Securities held by each of them; provided, however, that in the event that the Depositary shall be required to withhold and does withhold from such cash dividend or such other cash distribution an amount on account of taxes, the amount distributed to the Owner of the Receipts evidencing American Depositary Shares representing such Deposited Securities shall be reduced accordingly. The Depositary shall distribute only such amount, however, as can be distributed without attributing to any Owner a fraction of one cent. Any such fractional amounts shall be rounded down to the nearest whole cent and so distributed to Owners entitled thereto. The Company or its agent will remit to the appropriate governmental agency all amounts withheld and owing to such agency. The Depositary will forward to the Company or its agent such information from its records as the Company reasonably request to enable the Company or its agent to file necessary reports with governmental agencies, and the Depositary or the Company or its agent may file any such reports necessary to ob

Subject to the rules and regulations of any stock exchange upon which the American Depositary Shares may be traded, the Depositary shall endeavor to convert the funds as promptly as practicable and to distribute Dollars to Owners under this Section within five New York Stock Exchange trading days of the day on which the cash dividend or cash distribution on the Deposited Securities is received by the Depositary.

## SECTION 4.02. <u>Distributions Other Than Cash, Shares or Rights.</u>

Subject to the provisions of Sections 4.11 and 5.09, whenever the Depositary shall receive any distribution other than a distribution described in Section 4.01, 4.03 or 4.04, the Depositary shall cause the securities or property received by it to be distributed to the Owners entitled thereto, after deduction or upon payment of any fees and expenses of the Depositary or any taxes or other governmental charges, in proportion to the number of American Depositary

Shares representing such Deposited Securities held by them respectively, in any manner that the Depositary may deem equitable and practicable for accomplishing such distribution; provided, however, that if in the reasonable opinion of the Depositary such distribution cannot be made proportionately among the Owners entitled thereto, or if for any other reason (including, but not limited to, any requirement that the Company or the Depositary withhold an amount on account of taxes or other governmental charges or that such securities must be registered under the Securities Act of 1933 in order to be distributed to Owners or Beneficial Owners) the Depositary deems such distribution not to be feasible, the Depositary may, after consultation with the Company, adopt such method as it may deem equitable and practicable for the purpose of effecting such distribution, including, but not limited to, the public or private sale of the securities or property thus received, or any part thereof, and the net proceeds of any such sale (net of the fees and expenses of the Depositary as provided in Section 5.09) shall be distributed by the Depositary to the Owners entitled thereto, all in the manner and subject to the conditions described in Section 4.01; provided, however, that no distribution to Owners pursuant to this Section 4.02 shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any distribution of securities under this Section 4.02 if it has not received reasonably satisfactory assurances from the Company that the distribution does not require registration under the Securities Act of 1933.

## SECTION 4.03. <u>Distributions in Shares.</u>

If any distribution upon any Deposited Securities consists of a dividend in, or free distribution of, Shares, the Depositary may, and shall if the Company shall so request, distribute to the Owners of outstanding Receipts entitled thereto, in proportion to the number of American Depositary Shares representing such Deposited Securities held by them respectively, additional Receipts evidencing an aggregate number of American Depositary Shares representing the amount of Shares received as such dividend or free distribution, subject to the terms and conditions of the Deposit Agreement with respect to the deposit of Shares and the issuance of American Depositary Shares evidenced by Receipts, including the withholding of any tax or other governmental charge as provided in Section 4.11 and the payment of the fees and expenses of the Depositary as provided in Section 5.09; provided, however, that no distribution to Owners pursuant to this Section 4.03 shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any such distribution of Receipts if it has not received satisfactory assurances from the Company that such distribution does not require registration under the Securities Act of 1933. In lieu of delivering Receipts for fractional American Depositary Shares in any such case, the Depositary shall use reasonable efforts to sell the amount of Shares represented by the aggregate of such fractions and distribute the net proceeds, if any, all in the manner and subject to the conditions described in Section 4.01. If additional Receipts are not so distributed, each American Depositary Share shall thenceforth also represent the additional Shares distributed upon the Deposited Securities represented thereby.

## SECTION 4.04. Rights.

In the event that the Company shall offer or cause to be offered to the holders of any Deposited Securities any rights to subscribe for additional Shares or any rights of any other nature, the Depositary, after consultation with the Company, shall have discretion as to the

procedure to be followed in making such rights available to any Owners or in disposing of such rights on behalf of any Owners and making the net proceeds available to such Owners or, if by the terms of such rights offering or for any other reason, the Depositary may not either make such rights available to any Owners or dispose of such rights and make the net proceeds available to such Owners, then the Depositary shall allow the rights to lapse. If at the time of the offering of any rights the Depositary, after consultation with the Company, determines in its reasonable discretion that it is lawful and feasible to make such rights available to all or certain Owners but not to other Owners, the Depositary may, and at the request of the Company shall, distribute to any Owner to whom it determines the distribution to be lawful and feasible, in proportion to the number of American Depositary Shares held by such Owner, warrants or other instruments therefor in such form as it deems appropriate.

In circumstances in which rights would otherwise not be distributed, if an Owner of Receipts requests the distribution of warrants or other instruments in order to exercise the rights allocable to the American Depositary Shares of such Owner hereunder, the Depositary will make such rights available to such Owner upon written notice from the Company to the Depositary that (a) the Company has elected in its sole discretion to permit such rights to be exercised and (b) such Owner has executed such documents as the Company has determined in its sole discretion are reasonably required under applicable law.

If the Depositary has distributed warrants or other instruments for rights to all or certain Owners, then upon instruction from such an Owner pursuant to such warrants or other instruments to the Depositary from such Owner to exercise such rights, upon payment by such Owner to the Depositary for the account of such Owner of an amount equal to the purchase price of the Shares to be received upon the exercise of the rights, and upon payment of the fees and expenses of the Depositary and any other charges as set forth in such warrants or other instruments, the Depositary shall, on behalf of such Owner, exercise the rights and purchase the Shares, and the Company shall cause the Shares so purchased to be delivered to the Depositary on behalf of such Owner. As agent for such Owner, the Depositary will cause the Shares so purchased to be deposited pursuant to Section 2.02 of this Deposit Agreement, and shall, pursuant to Section 2.03 of this Deposit Agreement, execute and deliver Receipts to such Owner; provided, however, that in the case of a distribution pursuant to the second paragraph of this Section, such Receipts shall be legended in accordance with applicable U.S. laws, and shall be subject to the appropriate restrictions on sale, deposit, cancellation, and transfer under applicable United States laws.

If the Depositary determines that it is not lawful and feasible to make such rights available to all or certain Owners, it may sell the rights, warrants or other instruments in proportion to the number of American Depositary Shares held by the Owners to whom it has determined it may not lawfully or feasibly make such rights available, and allocate the net proceeds of such sales (net of the fees and expenses of the Depositary as provided in Section 5.09 and all taxes and governmental charges payable in connection with such rights and subject to the terms and conditions of this Deposit Agreement) for the account of such Owners otherwise entitled to such rights, warrants or other instruments, upon an averaged or other practical basis without regard to any distinctions among such Owners because of exchange restrictions or the date of delivery of any Receipt or otherwise.

The Depositary will not offer rights to Owners unless both the rights and the securities to which such rights relate are either exempt from registration under the Securities Act of 1933 with respect to a distribution to all Owners or are registered under the provisions of such Act; provided, that nothing in this Deposit Agreement shall create any obligation on the part of the Company to file a registration statement with respect to such rights or underlying securities or to endeavor to have such a registration statement declared effective. If an Owner of Receipts requests the distribution of warrants or other instruments, notwithstanding that there has been no such registration under the Securities Act of 1933, the Depositary shall not effect such distribution unless it has received an opinion from recognized counsel in the United States for the Company upon which the Depositary may rely that such distribution to such Owner does not require such registration.

The Depositary shall not be responsible for any failure to determine that it may be lawful or feasible to make such rights available to Owners in general or any Owner in particular.

## SECTION 4.05. Conversion of Foreign Currency.

Whenever the Depositary or the Custodian shall receive foreign currency, by way of dividends or other distributions or the net proceeds from the sale of securities, property or rights, and if at the time of the receipt thereof the foreign currency so received can in the reasonable judgment of the Depositary be converted on a reasonable basis into Dollars and the resulting Dollars transferred to the United States, the Depositary shall, as promptly as practicable, convert or cause to be converted, by sale or in any other manner that it may reasonably determine, such foreign currency into Dollars, and such Dollars shall be distributed to the Owners entitled thereto or, if the Depositary shall have distributed any warrants or other instruments which entitle the holders thereof to such Dollars, then to the holders of such warrants and/or instruments upon surrender thereof for cancellation. Such distribution may be made upon an averaged or other practicable basis without regard to any distinctions among Owners on account of exchange restrictions, the date of delivery of any Receipt or otherwise and shall be net of any expenses of conversion into Dollars incurred by the Depositary as provided in Section 5.09.

If such conversion or distribution can be effected only with the approval or license of any government or agency thereof, the Depositary shall file such application for approval or license, if any, as it may deem desirable.

If at any time the Depositary shall determine that in its reasonable judgment any foreign currency received by the Depositary or the Custodian is not convertible on a reasonable basis into Dollars transferable to the United States, or if any approval or license of any government or agency thereof which is required for such conversion is denied or in the reasonable opinion of the Depositary is not obtainable, or if any such approval or license is not obtained within a reasonable period as determined by the Depositary, the Depositary may distribute the foreign currency (or an appropriate document evidencing the right to receive such foreign currency) received by the Depositary to, or in its discretion may hold such foreign currency uninvested and without liability for interest thereon for the accounts of, the Owners entitled to receive the same.

If any such conversion of foreign currency, in whole or in part, cannot be effected for distribution to some of the Owners entitled thereto, the Depositary may in its discretion make such conversion and distribution in Dollars to the extent permissible to the Owners entitled thereto and may distribute the balance of the foreign currency received by the Depositary to, or hold such balance uninvested and without liability for interest thereon for the accounts of, the Owners entitled thereto.

## SECTION 4.06. Fixing of Record Date.

Whenever any cash dividend or other cash distribution shall become payable or any distribution other than cash shall be made, or whenever rights shall be issued with respect to the Deposited Securities, or whenever the Depositary shall receive notice of any meeting of holders of Shares or other Deposited Securities, or whenever for any reason the Depositary causes a change in the number of Shares that are represented by each American Depositary Share, or whenever the Depositary shall find it necessary, the Depositary shall fix a record date (a) for the determination of the Owners who shall be (i) entitled to receive such dividend, distribution or rights or the net proceeds of the sale thereof or (ii) entitled to give instructions for the exercise of voting rights at any such meeting or (iii) responsible for any fee or charges assessed by the Depositary pursuant to this Deposit Agreement, or (b) on or after which each American Depositary Share will represent the changed number of Shares. Subject to the provisions of Sections 4.01 through 4.05 and to the other terms and conditions of this Deposit Agreement, the Owners on such record date shall be entitled, as the case may be, to receive the amount distributable by the Depositary with respect to such dividend or other distribution or such rights or the net proceeds of sale thereof in proportion to the number of American Depositary Shares held by each of them, to give voting instructions and to act in respect of any other such matter.

## SECTION 4.07. Voting of Deposited Securities.

Upon receipt of notice of any meeting of holders of Shares or other Deposited Securities, if requested in writing by the Company, the Depositary shall, as soon as practicable thereafter, mail to the Owners of record as of the record date set by the Depositary under Section 4.06 (the "Voting Record Date") a notice, the form of which notice shall be approved of by the Company which shall contain (a) such information as is contained in such notice of meeting, (b) a voting instruction card in the form prepared by the Depositary after consultation with the Company, (c) a statement that the Owners of record as of the close of business on the Voting Record Date will be entitled, subject to any applicable provision of English law and of the Memorandum and Articles of Association of the Company and the provisions of or governing the Deposited Securities, to either (i) use such voting instruction card to request the Depositary, its Custodian or nominee (as appropriate) to appoint the Owner its proxy to attend at that meeting and vote with respect to the number of Shares or other Deposited Securities represented by American Depositary Shares evidenced by such Owner's Receipts or (ii) instruct the person nominated by the Depositary, its Custodian or nominee as its proxy as to the exercise of the voting rights pertaining to that number of Shares or other Deposited Securities, and (d) a brief statement as to the manner in which voting instructions may be given to the person nominated by the Depositary.

Upon the written request of an Owner of a Receipt as of the Voting Record Date received on or before the date established by the Depositary for that purpose (the "Instruction Date"), the Depositary shall endeavor to cause the appointment (or, if the Deposited Securities are registered in the name of or held by its Custodian or a nominee, the Depositary shall endeavor to procure that the Custodian or its nominee shall cause the appointment), subject to the articles of association of the Company, of that Owner as a proxy in respect of that meeting (including any adjournment of that meeting) to attend and vote the number of Deposited Securities represented by the American Depositary Shares evidenced by that Receipt.

Upon the written request of an Owner of a Receipt as of the Voting Record Date, received on or before the Instruction Date, the Depositary shall endeavor, in so far as practicable, to vote or cause to be voted the number of Shares or other Deposited Securities represented by the American Depositary Shares evidenced by such Receipt in accordance with the instructions set forth in such request. Neither the Depositary, nor the Custodian nor the nominee of either of them shall vote or attempt to exercise the right to vote that attaches to the Shares or other Deposited Securities other than in accordance with such written instructions from Owners given in accordance with this Section 4.07.

There can be no assurance that Owners generally or any Owner in particular will receive the notice described in the first paragraph of this Section 4.07 sufficiently prior to the Instruction Date to ensure that the Depositary will appoint the Owner its proxy or vote the Shares or other Deposited Securities as requested in accordance with the provisions set forth in the preceding paragraphs.

## SECTION 4.08. Changes Affecting Deposited Securities.

In circumstances where the provisions of Section 4.03 do not apply, upon any change in nominal value, change in par value, split-up, consolidation or any other reclassification of Deposited Securities, or upon any recapitalization, reorganization, merger or consolidation or sale of assets affecting the Company or to which it is a party, any securities which shall be received by the Depositary or a Custodian in exchange for or in conversion of or in respect of Deposited Securities, shall be treated as new Deposited Securities under this Deposit Agreement, and American Depositary Shares shall thenceforth represent, in addition to the existing Deposited Securities, the right to receive the new Deposited Securities so received in exchange or conversion, unless additional Receipts are delivered pursuant to the following sentence. In any such case the Depositary may, and shall if the Company shall so reasonably request, execute and deliver additional Receipts as in the case of a dividend in Shares, or call for the surrender of outstanding Receipts to be exchanged for new Receipts specifically describing such new Deposited Securities.

## SECTION 4.09. Reports.

The Company is subject to the periodic reporting requirements of the Securities Exchange Act of 1934 and accordingly files certain reports with the United States Securities and Exchange Commission (hereinafter called the "Commission"). Such reports and other information may be inspected and copied at public reference facilities maintained by the

Commission located at the date hereof at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549. The Depositary shall make available for inspection by Owners at its Principal Office any reports and communications, including any proxy soliciting material, received from the Company which are both (a) received by the Depositary as the holder of the Deposited Securities and (b) made generally available to the holders of such Deposited Securities by the Company. The Depositary shall also, upon the written request of the Company, send to the Owners copies of such reports when furnished by the Company pursuant to Section 5.06.

## SECTION 4.10. Lists of Owners.

Upon the written request of the Company, the Depositary shall, as promptly as practical, at the expense of the Company, furnish to it a list, as of a recent date, of the names, addresses and holdings of American Depositary Shares by all persons in whose names Receipts are registered on the books of the Depositary.

### SECTION 4.11. Withholding.

In the event that the Depositary determines that any distribution in property (including Shares and rights to subscribe therefor) is subject to any tax or other governmental charge which the Depositary is obligated to withhold, the Depositary may by public or private sale dispose of all or a portion of such property (including Shares and rights to subscribe therefor) in such amounts and in such manner as the Depositary reasonably deems necessary and practicable to pay such taxes or charges and the Depositary shall distribute the net proceeds of any such sale after deduction of such taxes or charges to the Owners entitled thereto in proportion to the number of American Depositary Shares held by them.

## ARTICLE 5. THE DEPOSITARY, THE CUSTODIANS AND THE COMPANY

## SECTION 5.01. Maintenance of Office and Transfer Books by the Depositary.

Until termination of this Deposit Agreement in accordance with its terms, the Depositary shall maintain in the Borough of Manhattan, The City of New York, facilities for the execution and delivery, registration, registration of transfers and surrender of Receipts in accordance with the provisions of this Deposit Agreement.

The Depositary shall keep books, at its Principal Office, for the registration of Receipts and transfers of Receipts which at all reasonable times shall be open for inspection by the Owners, provided that such inspection shall not be for the purpose of communicating with Owners in the interest of a business or object other than the business of the Company or a matter related to this Deposit Agreement or the Receipts.

The Depositary may close the transfer books, at any time or from time to time, when reasonably deemed expedient by it in connection with the performance of its duties hereunder or at the written reasonable request of the Company, provided that any such closing of

the transfer books shall be subject to the provisions of Section 2.06 which limit the suspension of withdrawals of Shares.

If any Receipts or the American Depositary Shares evidenced thereby are listed on one or more stock exchanges in the United States, the Depositary shall act as Registrar or appoint a Registrar or one or more co-registrars for registry of such Receipts in accordance with any requirements of such exchange or exchanges. The Company shall have the right, upon reasonable written request, to inspect the transfer and registration records of the Depositary relating to the Receipts, and to take copies thereof.

## SECTION 5.02. Prevention or Delay in Performance by the Depositary or the Company.

Neither the Depositary nor the Company nor any of their directors, employees, agents or affiliates shall incur any liability to any Owner or Beneficial Owner if, by reason of any provision of any present or future law or regulation of the United States or any other country, or of any governmental or regulatory authority or stock exchange, or by reason of any provision, present or future, of the Memorandum or Articles of Association of the Company, or by reason of any provision of any securities issued or distributed by the Company, or any offering or distribution thereof, or by reason of any act of God or war or terrorism or other circumstances beyond its control, the Depositary or the Company shall be prevented, delayed or forbidden from, or be subject to any civil or criminal penalty on account of, doing or performing any act or thing which by the terms of this Deposit Agreement or Deposited Securities it is provided shall be done or performed; nor shall the Depositary or the Company or any of their directors, employees, agents or affiliates incur any liability to any Owner or Beneficial Owner of any Receipt by reason of any nonperformance or delay, caused as aforesaid, in the performance of any act or thing which by the terms of this Deposit Agreement it is provided shall or may be done or performed, or by reason of any exercise of, or failure to exercise, any discretion provided for in this Deposit Agreement. Where, by the terms of a distribution pursuant to Section 4.01, 4.02, or 4.03, or an offering or distribution pursuant to Section 4.04, or for any other reason, such distribution or offering may not be made available to Owners, and the Depositary may not dispose of such distribution or offering on behalf of such Owners and make the net proceeds available to Such Owners, then the Depositary shall not make such distribution or offering, and shall allow any rights, if applicable, to lapse.

### SECTION 5.03. Obligations of the Depositary, the Custodian and the Company.

Neither the Company nor any of its directors, officers, employees or agents assumes any obligation nor shall any of them be subject to any liability under this Deposit Agreement to Owners or Beneficial Owners, except that the Company agrees to perform its obligations specifically set forth in this Deposit Agreement without negligence or bad faith.

Neither the Depositary nor any of its directors, officers, employees or agents assumes any obligation nor shall any of them be subject to any liability under this Deposit Agreement to any Owner or Beneficial Owner (including, without limitation, liability with

respect to the validity or worth of the Deposited Securities), except that the Depositary agrees to perform its obligations specifically set forth in this Deposit Agreement without negligence or bad faith.

Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be under any obligation to appear in or prosecute or defend any action, suit or other proceeding in respect of any Deposited Securities or in respect of the Receipts on behalf of any Owner, Beneficial Owner or other person. The parties understand that the Custodian is not a party to this Deposit Agreement and, accordingly, has no obligations whatsoever with respect to such proceedings, the responsibility of the Custodian being solely to the Depositary.

Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be liable for any action or nonaction by it in reliance upon the advice of or information from legal counsel, accountants, any person presenting Shares for deposit, any Owner or any other person believed by it in good faith to be competent to give such advice or information. Each of the Depositary, the Company and their respective directors, officers, employees and agents may rely and shall be protected in acting upon any written notice, request, direction or other document believed by such person to be genuine and to have been signed or presented by the proper party or parties.

The Depositary shall not be liable for any acts or omissions made by a successor depositary whether in connection with a previous act or omission of the Depositary or in connection with any matter arising wholly after the removal or resignation of the Depositary, provided that in connection with the issue out of which such potential liability arises the Depositary performed its obligations without negligence or bad faith while it acted as Depositary.

The Depositary and its agents will not be responsible for (i) any failure to carry out any instructions to vote any of the Deposited Securities or for the manner in which any such vote is cast, in each case to the extent the Depositary or its agents act without gross negligence or willful misconduct or (ii) for the effect of any such vote.

Notwithstanding anything to the contrary set forth in the Deposit Agreement or any Receipt, the Depositary and its agents may fully respond to any and all demands or requests for information maintained by or on its behalf in connection with the Deposit Agreement, any Owner or Owners, any Receipt or Receipts or otherwise related hereto to the extent such information is requested or required by or pursuant to any lawful authority, including without limitation laws, rules, regulations, administrative or judicial process, banking, securities or other regulators. Neither the Company nor the Depositary nor any of their respective agents shall be liable to Owners or holders of interests in American Depositary Shares or any other third party or parties for any indirect, special, punitive or consequential damages.

No disclaimer of liability under the Securities Act of 1933 is intended by any provision of this Deposit Agreement.

## SECTION 5.04. Resignation and Removal of the Depositary.

The Depositary may at any time resign as Depositary hereunder by written notice of its election so to do delivered to the Company, such resignation to take effect upon the appointment of a successor depositary and its acceptance of such appointment as hereinafter provided.

The Depositary may at any time be removed by the Company by written notice to the Depositary, such removal to become effective upon the appointment of a successor depositary and its acceptance of such appointment as hereinafter provided.

In case at any time the Depositary acting hereunder shall resign or be removed, the Company shall use its reasonable efforts to appoint a successor depositary, which shall be a bank or trust company having an office in the Borough of Manhattan, The City of New York. Every successor depositary shall execute and deliver to its predecessor and to the Company an instrument in writing accepting its appointment hereunder, and thereupon such successor depositary, without any further act or deed, shall become fully vested with all the rights, powers, duties and obligations of its predecessor; but such predecessor, nevertheless, upon payment of all sums due it and on the written request of the Company shall execute and deliver an instrument transferring to such successor all rights and powers of such predecessor hereunder, shall duly assign, transfer and deliver all right, title and interest in the Deposited Securities to such successor, and shall deliver to such successor a list of the Owners of all outstanding Receipts. Any such successor depositary shall promptly mail notice of its appointment to the Owners.

Any corporation into or with which the Depositary may be merged or consolidated shall be the successor of the Depositary without the execution or filing of any document or any further act.

## SECTION 5.05. The Custodians.

The Custodian shall be subject at all times and in all respects to the directions of the Depositary and shall be responsible solely to it. If the Depositary receives notice of the resignation of a Custodian and, upon the effectiveness of such resignation, there would be no Custodian acting hereunder, the Depositary shall, promptly after receiving such notice and upon consultation with the Company if practicable, appoint a substitute custodian or custodians, each of which shall thereafter be a Custodian hereunder. Whenever the Depositary in its reasonable discretion determines that it is in the best interest of the Owners to do so, it may, after consultation with the Company if practicable, appoint, a substitute or additional custodian or custodians, each of which shall thereafter be one of the Custodians hereunder. The Depositary shall require each such substitute or additional custodian to deliver to the Depositary, forthwith upon its appointment, an acceptance of such appointment satisfactory in form and substance to the Depositary.

Upon the appointment of any successor depositary hereunder, each Custodian then acting hereunder shall forthwith become, without any further act or writing, the agent

hereunder of such successor depositary and the appointment of such successor depositary shall in no way impair the authority of each Custodian hereunder; but the successor depositary so appointed shall, nevertheless, on the written request of any Custodian, execute and deliver to such Custodian all such instruments as may be proper to give to such Custodian full and complete power and authority as agent hereunder of such successor depositary.

### SECTION 5.06. Notices and Reports.

On or before the first date on which the Company gives notice, by publication or otherwise, of any meeting of holders of Shares or other Deposited Securities, or of any adjourned meeting of such holders, or of the taking of any action in respect of any cash or other distributions or the offering of any rights, the Company agrees to transmit to the Depositary and the Custodian a copy of the notice thereof in English but otherwise in the form given or to be given to holders of Shares or other Deposited Securities.

The Company will arrange for the translation into English, if not already in English, to the extent required pursuant to any regulations of the Commission, and the prompt transmittal by the Company to the Depositary and the Custodian of such notices and any other reports and communications which are made generally available by the Company to holders of its Shares. If requested by the Company, the Depositary will arrange for the mailing, at the Company's expense, of copies of such notices, reports and communications to all Owners. The Company will timely provide the Depositary with the quantity of such notices, reports, and communications, as reasonably requested by the Depositary from time to time, in order for the Depositary to effect such mailings.

# SECTION 5.07. <u>Distribution of Additional Shares, Rights, etc.</u>

If the Company or any affiliate of the Company determines to make any issuance or distribution of (1) additional Shares, (2) rights to subscribe for Shares, (3) securities convertible into Shares, or (4) rights to subscribe for such securities (each a "Distribution"), the Company shall notify the Depositary in writing in English as promptly as practicable and in any event before the Distribution starts and, if requested in writing by the Depositary, the Company shall promptly furnish to the Depositary a written opinion from U.S. counsel for the Company that is reasonably satisfactory to the Depositary, stating whether or not the Distribution requires, or, if made in the United States, would require, registration under the Securities Act of 1933. If, in the opinion of that counsel, the Depositary a written opinion as to whether or not there is a registration statement under the Securities Act of 1933 in effect that will cover that Distribution.

The Company agrees with the Depositary that neither the Company nor any company controlled by, controlling or under common control with the Company will at any time deposit any Shares, either originally issued or previously issued and reacquired by the Company or any such affiliate, unless a registration statement is in effect as to such Shares under the Securities Act of 1933.

The Company reserves full discretion as to whether in the future it may or may not register under said Act for purposes of offering and selling in the United States any Shares or any other securities, including any Shares or other securities which may be the subject of subscription or purchase rights pertaining to Deposited Securities at the time deposited under this Deposit Agreement.

### SECTION 5.08. Indemnification.

The Company agrees to indemnify the Depositary, its directors, employees, agents and affiliates and any Custodian against, and hold each of them harmless from, any liability or expense (including, but not limited to, the reasonable fees and expenses of counsel) which may arise out of or in connection with (a) any registration with the Commission of Receipts, American Depositary Shares or Deposited Securities or the offer or sale thereof in the United States or (b) acts performed or omitted, pursuant to the provisions of this Deposit Agreement and of the Receipts, as the same may be amended, modified or supplemented from time to time, (i) by either the Depositary or a Custodian or their respective directors, employees, agents and affiliates, or (ii) by the Company or any of its directors, employees, agents and affiliates excepting, however, any liability arising out of the negligence or bad faith of the Depositary or the Custodian or any of their respective directors, employees, agents or affiliates or the Registrar or any co-transfer agent.

The Depositary agrees to indemnify the Company, its directors, employees, agents and affiliates and hold them harmless from any liability or expense which may arise out of acts performed or omitted by the Depositary or its Custodian or their respective directors, employees, agents and affiliates in connection with the issuance of Pre-Released Receipts and the transactions contemplated by any Pre-Release Agreement or due to their negligence or bad faith.

Notwithstanding any other provision of this Deposit Agreement or the Receipts to the contrary, neither the Company nor the Depositary, nor any of their agents, shall be liable to the other for any indirect, special, punitive or consequential damages (collectively "Special Damages") except (i) to the extent such Special Damages arise from the gross negligence or willful misconduct of the party from whom indemnification is sought or (ii) to the extent Special Damages arise from or out of a claim brought by a third party (including, without limitation, Owners) against the Depositary or its agents, except to the extent such Special Damages arise out of the gross negligence or willful misconduct of the party seeking indemnification hereunder.

The Company and the Depositary agree that the indemnification provided in the first paragraph of this Section 5.08 shall apply to the Depositary's implementation of Profile and that, to the extent the relevant transfer is performed in connection with and in accordance with the arrangements and procedures related to Profile generally in effect, reliance by the Depositary upon information, or compliance with directions, it receives from a DTC participant claiming to act on behalf of an Owner of Direct Registration Receipts to register a transfer of American Depositary Shares to DTC or its nominee or to deliver American Depositary Shares to the DTC account of that DTC participant, without receipt by the Depositary of prior authorization from the Owner to register such transfer or make such delivery (unless such prior authorization is

required by Profile), shall not be deemed negligence or bad faith by the Depositary within the meaning of this Section 5.08, unless the Depositary had actual knowledge, or had reason to know (despite the absence of any investigation by it), that such directions were not authorized or were otherwise invalid.

## SECTION 5.09. Charges of Depositary.

The Company agrees to pay the fees, reasonable expenses and out-of-pocket charges of the Depositary and those of any Registrar only in accordance with agreements in writing entered into between the Depositary and the Company from time to time. The Depositary shall present its statement for such charges and expenses to the Company once every three months. The charges and expenses of the Custodian are for the sole account of the Depositary.

The following charges shall be incurred by any party depositing or withdrawing Shares or by any party surrendering Receipts or to whom Receipts are issued (including, without limitation, issuance pursuant to a stock dividend or stock split declared by the Company or an exchange of stock regarding the Receipts or Deposited Securities or a distribution of Receipts pursuant to Section 4.03), or by Owners, as applicable: (1) taxes and other governmental charges, (2) such registration fees as may from time to time be in effect for the registration of transfers of Shares generally on the Share register of the Company or Foreign Registrar and applicable to transfers of Shares to or from the name of the Depositary or its nominee or the Custodian or its nominee on the making of deposits or withdrawals hereunder, (3) such cable, telex and facsimile transmission expenses as are expressly provided in this Deposit Agreement, (4) such expenses as are incurred by the Depositary in the conversion of foreign currency pursuant to Section 4.05, (5) a fee of \$5.00 or less per 100 American Depositary Shares (or portion thereof) for the execution and delivery of Receipts pursuant to Section 2.03, 4.03 or 4.04 and the surrender of Receipts pursuant to Section 2.05 or 6.02, and (6) a fee for the distribution of securities pursuant to Section 4.02, such fee being in an amount equal to the fee for the execution and delivery of American Depositary Shares referred to above which would have been charged as a result of the deposit of such securities (for purposes of this clause 6 treating all such securities as if they were Shares) but which securities are instead distributed by the Depositary to Owners

The Depositary, subject to Section 2.09, may own and deal in any class of securities of the Company and its affiliates and in Receipts.

### SECTION 5.10. Retention of Depositary Documents.

The Depositary is authorized to destroy those documents, records, bills and other data compiled during the term of this Deposit Agreement at the times permitted by the laws or regulations governing the Depositary unless the Company requests that such papers be retained for a longer period.

### SECTION 5.11. Exclusivity.

Subject to Section 5.04, the Company agrees not to appoint any other depositary for issuance of American or global depositary receipts for the Shares so long as The Bank of New York is acting as Depositary hereunder.

## SECTION 5.12. <u>List of Restricted Securities Owners.</u>

The Company shall provide to the Depositary a list setting forth, to the actual knowledge of the Company, those persons or entities who beneficially own Restricted Securities as of the date hereof and the Company shall update that list on a regular basis as changes occur. The Company agrees to advise in writing each of the persons or entities so listed that such Restricted Securities, so long as they remain such, are ineligible for deposit hereunder. The Depositary (i) may rely on the list provided under this Section 5.12, as most recently updated, but shall not be liable for any action or omission made in reliance thereon and (ii) shall keep such a list strictly confidential, except as required by applicable law, legal process, regulation, judicial or administrative proceeding.

### ARTICLE 6. AMENDMENT AND TERMINATION

### SECTION 6.01. Amendment.

The form of the Receipts and any provisions of this Deposit Agreement may at any time and from time to time be amended by agreement between the Company and the Depositary without the consent of Owners or Beneficial Owners in any respect which they may deem necessary or desirable. Any amendment which shall impose or increase any fees or charges (other than taxes and other governmental charges, registration fees, cable, telex or facsimile transmission costs, delivery costs or other such expenses), or which shall otherwise prejudice any substantial existing right of Owners, shall, however, not become effective as to outstanding Receipts until the expiration of 30 days after notice of such amendment shall have been given to the Owners of outstanding Receipts. Every Owner and Beneficial Owner, at the time any amendment so becomes effective, shall be deemed, by continuing to hold such Receipt or any interest therein, to consent and agree to such amendment and to be bound by the Deposit Agreement as amended thereby. In no event shall any amendment impair the right of the Owner of any Receipt to surrender such Receipt and receive therefor the Deposited Securities represented thereby, except in order to comply with mandatory provisions of applicable law.

The Company and the Depositary shall each use their reasonable efforts to amend this Deposit Agreement as necessary to reflect changes in English or U.S. law and in the Company's Memorandum and Articles of Association.

### SECTION 6.02. Termination.

The Depositary shall, at any time at the direction of the Company, terminate this Deposit Agreement by mailing notice of such termination to the Owners of all Receipts then outstanding at least 30 days prior to the date fixed in such notice for such termination. The Depositary may likewise terminate this Deposit Agreement by mailing notice of such termination

to the Company and the Owners of all Receipts then outstanding, if at any time 60 days shall have expired after the Depositary shall have delivered to the Company a written notice of its election to resign and a successor depositary shall not have been appointed and accepted its appointment as provided in Section 5.04. On and after the date of termination, the Owner of a Receipt will, upon (a) surrender of such Receipt at the Principal Office of the Depositary, and (b) payment of the fee of the Depositary for the surrender of Receipts referred to in section 2.05, and (c) payment of any applicable taxes or governmental charges, be entitled to delivery, to him or upon his order, of the amount of Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, If any Receipts shall remain outstanding after the date of termination, the Depositary thereafter shall discontinue the registration of transfers of Receipts, shall suspend the distribution of dividends to the Owners thereof, and shall not give any further notices or perform any further acts under this Deposit Agreement, except that the Depositary shall continue to collect dividends and other distributions pertaining to Deposited Securities, shall sell rights and other property as provided in this Deposit Agreement, and shall continue to deliver Deposited Securities, together with any dividends or other distributions received with respect thereto and the net proceeds of the sale of any rights or other property, in exchange for Receipts surrendered to the Depositary (after deducting, in each case, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of this Deposit Agreement, and any applicable taxes or governmental charges) and for its obligations under Section 5.08 hereof. At any time after the expiration of six months from the date of termination, the Depositary may sell the Deposited Securities then held hereunder and may thereafter hold the net proceeds of any such sale, together with any other cash then held by it hereunder, without liability for interest, for the pro rata benefit of the Owners of Receipts which have not theretofore been surrendered, such Owners thereupon becoming general creditors of the Depositary with respect to such net proceeds. All such proceeds and cash shall be invested in direct obligations of the federal government of the United States. After making such sale, the Depositary shall be discharged from all obligations under this Deposit Agreement, except to account for such net proceeds and other cash (after deducting, in each case, the fee of the Depositary for the surrender of a Receipt, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of this Deposit Agreement, and any applicable taxes or governmental charges) and for its obligations under Section 5.08 hereof. Upon the termination of this Deposit Agreement, the Company shall be discharged from all obligations under this Deposit Agreement except for its obligations to the Depositary under Sections 5.08 and 5.09.

## **ARTICLE 7. MISCELLANEOUS**

SECTION 7.01. Counterparts.

This Deposit Agreement may be executed in any number of counterparts, each of which shall be deemed an original and all of such counterparts shall constitute one and the same instrument. Copies of this Deposit Agreement shall be filed with the Depositary and the Custodian and shall be open to inspection by any Owner or Beneficial Owner during business hours.

## SECTION 7.02. No Third Party Beneficiaries.

This Deposit Agreement is for the exclusive benefit of the parties hereto and, except with respect to indemnification of the Custodian as set forth in Section 5.08, shall not be deemed to give any legal or equitable right, remedy or claim whatsoever to any other person.

### SECTION 7.03. Severability.

In case any one or more of the provisions contained in this Deposit Agreement or in the Receipts should be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein or therein shall in no way be affected, prejudiced or disturbed thereby.

## SECTION 7.04. Owners and Beneficial Owners as Parties; Binding Effect.

The Owners and Beneficial Owners of Receipts from time to time shall be parties to this Deposit Agreement and shall be bound by all of the terms and conditions hereof and of the Receipts by acceptance thereof or any interest therein.

## SECTION 7.05. Notices.

Any and all notices to be given to the Company shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex or facsimile transmission confirmed by letter, addressed to Royal Dutch Shell plc, Shell Centre, London SE1 7NA, England, Attn: Company Secretary, or any other place to which the Company may have transferred its registered office with notice to the Depositary.

Any and all notices to be given to the Depositary shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex or facsimile transmission confirmed by letter, addressed to The Bank of New York, 101 Barclay Street, New York, New York 10286, Attention: American Depositary Receipt Administration, or any other place to which the Depositary may have transferred its Principal Office with notice to the Company.

Any and all notices to be given to any Owner shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex or facsimile transmission confirmed by letter, addressed to such Owner at the address of such Owner as it appears on the transfer books for Receipts of the Depositary, or, if such Owner shall have filed with the Depositary a written request that notices intended for such Owner be mailed to some other address, at the address designated in such request.

Delivery of a notice sent by mail or cable, telex or facsimile transmission shall be deemed to be effective at the time when a duly addressed letter containing the same (or a confirmation thereof in the case of a cable, telex or facsimile transmission) is deposited, postage prepaid, in a post-office letter box. The Depositary or the Company may, however, act upon any

cable, telex or facsimile transmission received by it, notwithstanding that such cable, telex or facsimile transmission shall not subsequently be confirmed by letter as aforesaid.

## SECTION 7.06. Compliance with U.S. Securities Laws.

Notwithstanding anything in this Deposit Agreement to the contrary, the Company and the Depositary each agrees that it will not exercise any rights it has under this Deposit Agreement to permit the withdrawal or delivery of Deposited Securities in a manner which would violate the U.S. securities laws, including, but not limited to, Section I.A.(1) of the General Instructions to the Form F-6 registration statement, as amended from time to time, under the Securities Act of 1933.

### SECTION 7.07. <u>Arbitration of Disputes.</u>

- (a) The Company and each Owner, but not the Depositary, shall be bound by the arbitration and exclusive jurisdiction provisions set out in articles 152, 153 and 154 of the Articles as if all references therein to "shareholder" were replaced with "Owner." Articles 152, 153 and 154 of the Articles shall accordingly be incorporated, <u>mutatis mutandis</u> into this Deposit Agreement.
  - (b) The Company will make a copy of the Articles available to Owners upon request.
- (c) All cross-references to the Articles in this Section 7.07 will be updated and amended without further action of any party in the event the Articles themselves are renumbered.

## SECTION 7.08. Governing Law.

This Deposit Agreement and the Receipts shall be interpreted and all rights hereunder and thereunder and provisions hereof and thereof shall be governed by the laws of the State of New York, except that the provisions of Section 7.07 and Article 25 of Exhibit A and the provisions incorporated by reference into that Section and that Article of Exhibit A shall be governed by the laws of England and Wales.

It is hereby acknowledged and agreed that Owners and Beneficial Owners of ADRs, as such, are not shareholders of the Company and have no direct rights of a shareholder against the Company. The rights of holders of and of the Company with respect to the Shares are governed exclusively by the Company's Memorandum and Articles of Association and the laws of England and Wales.

IN WITNESS WHEREOF, ROYAL DUTCH SHELL plc and THE BANK OF NEW YORK have duly executed this Deposit Agreement as of the day and year first set forth above and all Owners and Beneficial Owners shall become parties hereto upon acceptance by them of Receipts issued in accordance with the terms hereof or any interest therein.

# ROYAL DUTCH SHELL plc

By: /s/ Peter Voser

Name: Peter Voser

Title: Director and Chief Financial Officer

THE BANK OF NEW YORK, as Depositary

By: /s/ Joanne F. Di Giovanni

Name: Joanne F. Di Giovanni Title: Vice President

## **EXHIBIT A**

No.

## **AMERICAN DEPOSITARY SHARES**

(Each American Depositary Share represents two deposited Shares)

THE BANK OF NEW YORK
AMERICAN DEPOSITARY RECEIPT
FOR CLASS B ORDINARY SHARES
NOMINAL VALUE 0.07 EURO EACH OF
ROYAL DUTCH SHELL plc
(INCORPORATED UNDER THE LAWS OF ENGLAND AND WALES)

| The Bank of New York, as depositary (herein called the Depositary), hereby certifies that | , | 01 |
|---|---|----|
| registered assigns IS THE OWNER OF  |   |    |

## AMERICAN DEPOSITARY SHARES

representing deposited Class B ordinary shares (herein called Shares) of Royal Dutch Shell plc, incorporated under the laws of England and Wales (herein called the Company). At the date hereof, each American Depositary Share represents two Shares deposited or subject to deposit under the Deposit Agreement (as such term is hereinafter defined) at the principal London office of The Bank of New York.

THE DEPOSITARY'S PRINCIPAL OFFICE ADDRESS IS 101 BARCLAY STREET, NEW YORK, N.Y. 10286

A-1

## 1. THE DEPOSIT AGREEMENT.

This American Depositary Receipt is one of an issue (herein called Receipts), all issued and to be issued upon the terms and conditions set forth in the deposit agreement, dated as of May 19, 2005, as the same may be amended from time to time in accordance with its terms (the "Deposit Agreement"), by and among the Company, the Depositary, and all Owners and Beneficial Owners from time to time of American Depositary Shares issued thereunder, each of whom by accepting a Receipt or any interest therein agrees to become a party thereto and become bound by all the terms and conditions thereof. The Deposit Agreement sets forth the rights of Owners and Beneficial Owners of the Receipts and the rights and duties of the Depositary in respect of the Shares deposited thereunder and any and all other securities, property and cash from time to time received in respect of such Shares and held thereunder (such Shares, securities, property, and cash are herein called Deposited Securities). Copies of the Deposit Agreement are on file at the Depositary's Principal Office in New York City and at the office of the Custodian.

The statements made on the face and reverse of this Receipt are summaries of certain provisions of the Deposit Agreement and are qualified by and subject to the detailed provisions of the Deposit Agreement, to which reference is hereby made. Capitalized terms defined in the Deposit Agreement and not defined herein shall have the meanings set forth in the Deposit Agreement.

## 2. SURRENDER OF RECEIPTS AND WITHDRAWAL OF SHARES.

Upon surrender at the Principal Office of the Depositary of a Receipt endorsed in blank or accompanied by proper instruments of transfer in blank (in the case of a certificated Receipt), or surrendered pursuant to a proper instruction (in the case of a Direct Registration Receipt), and upon payment of the fee of the Depositary provided in this Receipt, and subject to the terms and conditions of the Deposit Agreement and English law, the Owner hereof is entitled to delivery, to him or upon his order, of the Deposited Securities at the time represented by the American Depositary Shares for which that Receipt is issued. Delivery of such Deposited Securities may be made (a) in the case of certificated Shares, by delivery of certificates in the name of such Owner or as ordered by him or certificates properly endorsed or accompanied by proper instruments of transfer to such Owner or as ordered by him and (b) in the case of any securities cash or other property to which such Owner is then entitled in respect of such Receipt, by delivery of those securities or that cash or other property to such Owner or as ordered by him. Such delivery will be made at the option of the Owner hereof, either at the office of the Custodian or at the Principal Office of the Depositary, provided that the forwarding of certificates for Shares or other Deposited Securities for such delivery at the Principal Office of the Depositary shall be at the risk and expense of the Owner hereof.

### 3. TRANSFERS, SPLIT-UPS, AND COMBINATIONS OF RECEIPTS.

The transfer of this Receipt is registrable on the books of the Depositary at its Principal Office by the Owner hereof in person or by a duly authorized attorney, upon surrender of this Receipt properly endorsed for transfer or accompanied by proper instruments of transfer

or, in the case of Direct Registration Receipts, upon receipt of a proper instruction, and funds sufficient to pay any applicable transfer taxes and the expenses of the Depositary and upon compliance with such regulations, if any, as the Depositary may establish for such purpose. This Receipt may be split into other such Receipts, or may be combined with other such Receipts into one Receipt, evidencing the same aggregate number of American Depositary Shares as the Receipt or Receipts surrendered. As a condition precedent to the execution and delivery, registration of transfer, split-up, combination, or surrender of any Receipt or withdrawal of any Deposited Securities, the Depositary, the Custodian, or Registrar may require payment from the depositor of the Shares or the presenter of the Receipt of a sum sufficient to reimburse it for any tax or other governmental charge and any stock transfer or registration fee with respect thereto (including any such tax or charge and fee with respect to Shares being deposited or withdrawn) and payment of any applicable fees as provided in this Receipt, may require the production of proof satisfactory to it as to the identity and genuineness of any signature and may also require compliance with any regulations the Depositary may establish consistent with the provisions of the Deposit Agreement or this Receipt, including, without limitation, this Article 3.

The delivery of Receipts against deposit of Shares generally or against deposit of particular Shares may be suspended, or the transfer of Receipts in particular instances may be refused, or the registration of transfer of outstanding Receipts generally may be suspended, during any period when the transfer books of the Depositary are closed, or if any such action is deemed necessary or advisable by the Depositary or the Company at any time or from time to time because of any requirement of law or of any government or governmental body or commission, or under any provision of the Deposit Agreement or this Receipt, or for any other reason, subject to the provisions of Section 7.07 of the Deposit Agreement. The Depositary shall not knowingly accept for deposit under the Deposit Agreement any Shares which would be required to be registered under the provisions of the Securities Act of 1933 for the public offer and sale thereof in the United States unless a registration statement is in effect as to such Shares for such offer and sale. The Depositary will use reasonable efforts to comply with written instructions of the Company that the Depositary shall not accept for the deposit hereunder any Shares identified in such circumstances as may reasonably be specified in such restrictions to facilitate the Company's compliance with the U.S. securities laws. Notwithstanding anything to the contrary in the Deposit Agreement, Owners shall be entitled to surrender Receipts and withdraw Deposited Securities as provided in Section 2.05 of the Deposit Agreement at any time, subject only to (i) temporary delays caused by closing the transfer books of the Depositary or the Company or the deposit of Shares in connection with voting at a shareholders' meeting, or the payment of dividends, (ii) the payment of fees, taxes and similar charges, (iii) compliance with any U.S. or foreign laws or governmental regulations relating to the Receipts or to the withdrawal of the Deposited Securities and (iv) any other reason that may at any time

### 4. LIABILITY OF OWNER OR BENEFICIAL OWNER FOR TAXES.

If any tax or other governmental charge shall become payable with respect to any Receipt or any Deposited Securities represented hereby, such tax or other governmental charge shall be payable by the Owner or Beneficial Owner hereof to the Depositary. The Depositary

may refuse to effect any transfer of this Receipt or any withdrawal of Deposited Securities represented by American Depositary Shares evidenced by such Receipt until such payment is made, and may withhold any dividends or other distributions, or may sell for the account of the Owner or Beneficial Owner hereof any part or all of the Deposited Securities represented by the American Depositary Shares evidenced by this Receipt, and may apply such dividends or other distributions or the proceeds of any such sale in payment of such tax or other governmental charge and the Owner or Beneficial Owner hereof shall remain liable for any deficiency. Neither the Company nor the Depositary shall be liable for failure of an Owner to comply with applicable tax laws or governmental charges.

## 5. WARRANTIES ON DEPOSIT OF SHARES.

Every person depositing Shares under the Deposit Agreement shall be deemed thereby to represent and warrant that such Shares and each certificate therefor are validly issued, fully paid, non-assessable, and were not issued in violation of any preemptive or similar rights of the holders of outstanding Shares and that the person making such deposit is duly authorized so to do. Every such person shall also be deemed to represent that such Shares are not, and American Depositary Shares representing such Shares would not be, Restricted Securities. Such representations and warranties shall survive the deposit of Shares and delivery of Receipts.

## 6. FILING PROOFS, CERTIFICATES, AND OTHER INFORMATION.

Any person presenting Shares for deposit or any Owner or Beneficial Owner of a Receipt may be required from time to time to file with the Depositary or the Custodian such proof of citizenship or residence, exchange control approval, or such information relating to the registration on the books of the Company or the Foreign Registrar, if applicable, to execute such certificates and to make such representations and warranties, as the Depositary may deem necessary or proper.

The Depositary may withhold the delivery or registration of transfer of any Receipt or the distribution of any dividend or sale or distribution of rights or of the proceeds thereof or the delivery of any Deposited Securities until such proof or other information is filed or such certificates are executed or such representations and warranties made. The Depositary shall notify the Company, upon its request, of the availability of any such proofs, certificates or other information and shall provide copies thereof to the Company as promptly as practicable upon request by the Company, unless such disclosure is prohibited by law. No Share shall be accepted for deposit unless accompanied by evidence satisfactory to the Depositary that any necessary approval, where relevant, has been granted by any governmental body in England and Wales that is then performing the function of the regulation of currency exchange.

## 7. CHARGES OF DEPOSITARY.

The Company agrees to pay the fees, reasonable expenses and out-of-pocket charges of the Depositary and those of any Registrar only in accordance with agreements in writing entered into between the Depositary and the Company from time to time. The Depositary shall present its statement for such charges and expenses to the Company once every

three months. The charges and expenses of the Custodian are for the sole account of the Depositary.

The following charges shall be incurred by any party depositing or withdrawing Shares or by any party surrendering Receipts or to whom Receipts are issued (including, without limitation, issuance pursuant to a stock dividend or stock split declared by the Company or an exchange of stock regarding the Receipts or Deposited Securities or a distribution of Receipts pursuant to Section 4.03 of the Deposit Agreement), or by Owners, as applicable: (1) taxes and other governmental charges, (2) such registration fees as may from time to time be in effect for the registration of transfers of Shares generally on the Share register of the Company or Foreign Registrar and applicable to transfers of Shares to or from the name of the Depositary or its nominee or the Custodian or its nominee on the making of deposits or withdrawals under the terms of the Deposit Agreement, (3) such cable, telex and facsimile transmission expenses as are expressly provided in the Deposit Agreement, (4) such expenses as are incurred by the Depositary in the conversion of foreign currency pursuant to Section 4.05 of the Deposit Agreement, (5) a fee of \$5.00 or less per 100 American Depositary Shares (or portion thereof) for the execution and delivery of Receipts pursuant to Section 2.03, 4.03 or 4.04 of the Deposit Agreement and the surrender of Receipts pursuant to Section 2.05 or 6.02 of the Deposit Agreement and (6) a fee for the distribution of securities pursuant to Section 4.02 of the Deposit Agreement, such fee being in an amount equal to the fee for the execution and delivery of American Depositary Shares referred to above which would have been charged as a result of the deposit of such securities (for purposes of this clause 7 treating all such securities as if they were Shares), but which securities are instead distributed by the Depositary to Owners.

The Depositary, subject to Article 8 hereof, may own and deal in any class of securities of the Company and its affiliates and in Receipts.

### 8. PRE-RELEASE OF RECEIPTS.

The Depositary will lend neither the Shares held under the Deposit Agreement nor the Receipts, provided, however, that the Depositary reserves the right to execute and deliver Receipts prior to the receipt of Shares pursuant to Section 2.02 of the Deposit Agreement on the terms and conditions set forth below and in the Deposit Agreement (a "Pre-Release"). The Depositary may receive Receipts in lieu of Shares as settlement of the Pre-Release of a Receipt. Subject to the terms and conditions of this Deposit Agreement, the Pre-Release of Receipts may occur only if (i) Pre-released Receipts are fully collateralized (marked to market daily) with cash or U.S. government securities in an amount equal to not less than 100% of the market value of the Pre-Released Receipts held by the Depositary for the benefit of Owners (but such collateral shall not constitute Deposited Securities), (ii) each recipient of Pre-released Receipts agrees in writing with the Depositary that such recipient (a) owns such Shares, (b) assigns all beneficial right, title and interest therein to the Depositary, (c) holds such Shares for the account of the Depositary and (d) will deliver such Shares to the Custodian as soon as practicable and promptly upon demand therefor and (iii) all Pre-released Receipts evidence not more than 20% of all American Depositary Shares (excluding those evidenced by Pre-released Receipts) and all Pre-Released Receipts evidence American Depositary Shares representing not more than 1% of all Shares outstanding or such other percentage of American Depositary Shares, as the

case may be, as the Company and the Depositary may from time to time agree in writing, except to the extent, if any, that either of such limitations is exceeded solely because of the withdrawal of Deposited Securities subsequent to the execution and delivery of Pre-Released Receipts in compliance with such limitation. The Depositary will also set limits with respect to the number of American Depositary Shares and Shares involved in transactions to be done under the Deposit Agreement with anyone person on a case by case basis as it deems appropriate. The Depositary may retain for its own account any compensation received by it in connection with the foregoing.

### 9. TITLE TO RECEIPTS.

It is a condition of this Receipt and every successive Owner and Beneficial Owner of this Receipt by accepting or holding the same consents and agrees, that the American Depositary Shares evidenced by this Receipt when properly endorsed or accompanied by proper instruments of transfer, shall be transferable as certificated securities in registered form under the laws of the State of New York. American Depositary Shares not evidenced by a Receipt (also referred to as Direct Registration Receipts) shall be transferable as uncertificated securities in registered form under the laws of the State of New York. The Depositary, notwithstanding any notice to the contrary, may treat the person in whose name American Depositary Shares are registered on the books of the Depositary as the absolute owner hereof for the purpose of determining the person entitled to distribution of dividends or other distributions or to any notice provided for in the Deposit Agreement or for all other purposes.

## 10. VALIDITY OF RECEIPT.

This Receipt shall not be entitled to any benefits under the Deposit Agreement or be valid or obligatory for any purpose, unless this Receipt shall have been executed by the Depositary by the manual signature of a duly authorized signatory of the Depositary; provided, however, that such signature may be a facsimile if a Registrar for the Receipts shall have been appointed and such Receipts are countersigned by the manual signature of a duly authorized officer of the Registrar.

### 11. REPORTS; INSPECTION OF TRANSFER BOOKS.

The Company is subject to the periodic reporting requirements of the Securities Exchange Act of 1934 and accordingly files certain reports with the United States Securities and Exchange Commission (hereinafter called the "Commission"). Such reports may be inspected and copied at public reference facilities maintained by the Commission located at the date hereof at Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549.

The Depositary will make available for inspection by Owners of Receipts at its Principal Office, any reports and communications, including any proxy soliciting material, received from the Company which are both (a) received by the Depositary as the holder of the Deposited Securities and (b) made generally available to the holders of such Deposited Securities by the Company. The Depositary will also, upon written request, send to Owners of Receipts copies of such reports when furnished by the Company pursuant to the Deposit Agreement. Any such reports and communications, including any such proxy soliciting material, furnished to the

Depositary by the Company shall be furnished in English to the extent such materials are required to be translated into English pursuant to any regulations of the Commission.

The Depositary will keep books, at its Principal Office, for the registration of Receipts and transfers of Receipts which at all reasonable times shall be open for inspection by the Owners of Receipts provided that such inspection shall not be for the purpose of communicating with Owners of Receipts in the interest of a business or object other than the business of the Company or a matter related to the Deposit Agreement or the Receipts.

## 12. <u>DIVIDENDS AND DISTRIBUTIONS</u>.

Notwithstanding any rights under the Company's articles of association, dividends paid on the Deposited Securities that are not paid to the Depositary or its nominee in Dollars will be paid by the Company in GBP. Whenever the Depositary receives any cash dividend or other cash distribution on any Deposited Securities distribution in respect of any Deposited Securities (including, without limitation, any dividend paid to holders of Shares in a foreign currency), the Depositary will, if at the time of receipt thereof any amounts received in a foreign currency can in the judgment of the Depositary be converted on a reasonable basis into United States dollars transferable to the United States, and subject to the Deposit Agreement, convert such dividend or distribution into Dollars. Promptly after the settlement of such conversion or, in the case of any cash dividend or other cash distribution received by the Depositary in Dollars, the Depositary shall as promptly as practicable, distribute the amount thus received (net of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.09 of the Deposit Agreement) to the Owners of Receipts entitled thereto; provided, however, that in the event that the Company or the Depositary is required to withhold and does withhold from any cash dividend or other cash distribution in respect of any Deposited Securities an amount on account of taxes, the amount distributed to the Owners of the Receipts evidencing American Depositary Shares representing such Deposited Securities shall be reduced accordingly. Subject to the rules and regulations of any stock exchange upon which the American Depositary Shares may be traded, the Depositary shall endeavor to convert the funds as promptly as practicable and to distribute Dollars to Owners pursuant to Section 4.01 of the Deposit Agreement within five New York Stock Exchange trading days of the day on which the cash dividend or cash distribution on the Deposited Securities is received by the Depositary.

Subject to the provisions of Section 4.11 and 5.09 of the Deposit Agreement, whenever the Depositary receives any distribution other than a distribution described in Section 4.01, 4.03 or 4.04 of the Deposit Agreement, the Depositary will cause the securities or property received by it to be distributed to the Owners entitled thereto, in any manner that the Depositary may deem equitable and practicable for accomplishing such distribution; provided, however, that if in the reasonable opinion of the Depositary such distribution cannot be made proportionately among the Owners of Receipts entitled thereto, or if for any other reason the Depositary deems such distribution not to be feasible, the Depositary may, after consultation with the Company, adopt such method as it may deem equitable and practicable for the purpose of effecting such distribution, including, but not limited to, the public or private sale of the securities or property thus received, or any part thereof, and the net proceeds of any such sale (net of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.09 of the Deposit

Agreement) will be distributed by the Depositary to the Owners of Receipts entitled thereto all in the manner and subject to the conditions described in Section 4.01 of the Deposit Agreement; provided, however, that no distribution to Owners pursuant to Section 4.02 of the Deposit Agreement shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any distribution of securities under Section 4.02 of the Deposit Agreement if it has not received satisfactory assurances from the Company that the distribution does not require registration under the Securities Act of 1933.

If any distribution consists of a dividend in, or free distribution of, Shares, the Depositary may, and shall if the Company shall so request, distribute to the Owners of outstanding Receipts entitled thereto, additional Receipts evidencing an aggregate number of American Depositary Shares representing the amount of Shares received as such dividend or free distribution subject to the terms and conditions of the Deposit Agreement with respect to the deposit of Shares and the issuance of American Depositary Shares evidenced by Receipts, including the withholding of any tax or other governmental charge as provided in Section 4.11 of the Deposit Agreement and the payment of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.09 of the Deposit Agreement; provided, however, that no distribution to Owners pursuant to this Section 4.03 shall be unreasonably delayed by any action or inaction of the Depositary or any of its agents. The Depositary may withhold any such distribution of Receipts if it has not received reasonably satisfactory assurances from the Company that such distribution does not require registration under the Securities Act of 1933 or is exempt from registration under the provisions of such Act. In lieu of delivering Receipts for fractional American Depositary Shares in any such case, the Depositary shall sell the amount of Shares represented by the aggregate of such fractions and distribute the net proceeds, all in the manner and subject to the conditions described in Section 4.01 of the Deposit Agreement. If additional Receipts are not so distributed, each American Depositary Share shall thenceforth also represent the additional Shares distributed upon the Deposited Securities represented thereby.

In the event that the Depositary determines that any distribution in property (including Shares and rights to subscribe therefor) is subject to any tax or other governmental charge which the Depositary is obligated to withhold, the Depositary may by public or private sale dispose of all or a portion of such property (including Shares and rights to subscribe therefor) in such amounts and in such manner as the Depositary reasonably deems necessary and practicable to pay any such taxes or charges, and the Depositary shall distribute the net proceeds of any such sale after deduction of such taxes or charges to the Owners of Receipts entitled thereto.

### 13. RIGHTS.

In the event that the Company shall offer or cause to be offered to the holders of any Deposited Securities any rights to subscribe for additional Shares or any rights of any other nature, the Depositary, after consultation with the Company, shall have discretion as to the procedure to be followed in making such rights available to any Owners or in disposing of such rights on behalf of any Owners and making the net proceeds available to such Owners or, if by

the terms of such rights offering or for any other reason, the Depositary may not either make such rights available to any Owners or dispose of such rights and make the net proceeds available to such Owners, then the Depositary shall allow the rights to lapse. If at the time of the offering of any rights the Depositary, after consultation with the Company, determines in its reasonable discretion that it is lawful and feasible to make such rights available to all or certain Owners but not to other Owners, the Depositary may, and at the request of the Company shall, distribute to any Owner to whom it determines the distribution to be lawful and feasible, in proportion to the number of American Depositary Shares held by such Owner, warrants or other instruments therefor in such form as it deems appropriate.

In circumstances in which rights would otherwise not be distributed, if an Owner of Receipts requests the distribution of warrants or other instruments in order to exercise the rights allocable to the American Depositary Shares of such Owner hereunder, the Depositary will make such rights available to such Owner upon written notice from the Company to the Depositary that (a) the Company has elected in its sole discretion to permit such rights to be exercised and (b) such Owner has executed such documents as the Company has determined in its sole discretion are reasonably required under applicable law.

If the Depositary has distributed warrants or other instruments for rights to all or certain Owners, then upon instruction from such an Owner pursuant to such warrants or other instruments to the Depositary from such Owner to exercise such rights, upon payment by such Owner to the Depositary for the account of such Owner of an amount equal to the purchase price of the Shares to be received upon the exercise of the rights, and upon payment of the fees and expenses of the Depositary and any other charges as set forth in such warrants or other instruments, the Depositary shall, on behalf of such Owner, exercise the rights and purchase the Shares, and the Company shall cause the Shares so purchased to be delivered to the Depositary on behalf of such Owner. As agent for such Owner, the Depositary will cause the Shares so purchased to be deposited pursuant to Section 2.02 of the Deposit Agreement, and shall, pursuant to Section 2.03 of the Deposit Agreement, execute and deliver Receipts to such Owner; provided, however, that in the case of a distribution pursuant to the second paragraph of this Article 13, such Receipts shall be legended in accordance with applicable U.S. laws, and shall be subject to the appropriate restrictions on sale, deposit, cancellation, and transfer under applicable United States laws.

If the Depositary determines that it is not lawful and feasible to make such rights available to all or certain Owners, it may sell the rights, warrants or other instruments in proportion to the number of American Depositary Shares held by the Owners to whom it has determined it may not lawfully or feasibly make such rights available, and allocate the net proceeds of such sales (net of the fees and expenses of the Depositary as provided in Section 5.09 of the Deposit Agreement and all taxes and governmental charges payable in connection with such rights and subject to the terms and conditions of the Deposit Agreement) for the account of such Owners otherwise entitled to such rights, warrants or other instruments, upon an averaged or other practical basis without regard to any distinctions among such Owners because of exchange restrictions or the date of delivery of any Receipt or otherwise.

The Depositary will not offer rights to Owners unless both the rights and the securities to which such rights relate are either exempt from registration under the Securities Act of 1933 with respect to a distribution to all Owners or are registered under the provisions of the Securities Act of 1933; provided, that nothing in the Deposit Agreement shall create any obligation on the part of the Company to file a registration statement with respect to such rights or underlying securities or to endeavor to have such a registration statement declared effective. If an Owner of Receipts requests the distribution of warrants or other instruments, notwithstanding that there has been no such registration under such Act, the Depositary shall not effect such distribution unless it has received an opinion from recognized counsel in the United States for the Company upon which the Depositary may rely that such distribution to such Owner is exempt from such registration. The Company will have no obligation under the Deposit Agreement to register such rights under the Securities Act of 1933.

The Depositary shall not be responsible for any failure to determine that it may be lawful or feasible to make such rights available to Owners in general or any Owner in particular.

## 14. CONVERSION OF FOREIGN CURRENCY.

Whenever the Depositary or the Custodian shall receive foreign currency, by way of dividends or other distributions or the net proceeds from the sale of securities, property or rights the Depositary shall, as promptly as practicable, convert or cause to be converted, by sale or in any other manner that it may reasonably determine, such foreign currency into Dollars, and such Dollars shall be distributed to the Owners entitled thereto or, if the Depositary shall have distributed any warrants or other instruments which entitle the holders thereof to such Dollars, then to the holders of such warrants or instruments upon surrender thereof for cancellation. Such distribution may be made upon an averaged or other practicable basis without regard to any distinctions among Owners on account of exchange restrictions, the date of delivery of any Receipt or otherwise and shall be net of any expenses of conversion into Dollars incurred by the Depositary as provided in Section 5.09 of the Deposit Agreement.

If such conversion or distribution can be effected only with the approval or license of any government or agency thereof, the Depositary shall file such application for approval or license, if any, as it may deem desirable.

If at any time the Depositary shall determine that in its reasonable judgment any foreign currency received by the Depositary or the Custodian is not convertible on a reasonable basis into Dollars transferable to the United States, or if any approval or license of any government or agency thereof which is required for such conversion is denied or in the reasonable opinion of the Depositary is not obtainable, or if any such approval or license is not obtained within a reasonable period as determined by the Depositary, the Depositary may distribute the foreign currency (or an appropriate document evidencing the right to receive such foreign currency) received by the Depositary to, or in its reasonable discretion may hold such foreign currency uninvested and without liability for interest thereon for the accounts of, the Owners entitled to receive the same.

If any such conversion of foreign currency, in whole or in part, cannot be effected for distribution to some of the Owners entitled thereto, the Depositary may in its discretion make such conversion and distribution in Dollars to the extent permissible to the Owners entitled thereto and may distribute the balance of the foreign currency received by the Depositary to, or hold such balance uninvested and without liability for interest thereon for the accounts of, the Owners entitled thereto.

### 15. RECORD DATES.

Whenever any cash dividend or other cash distribution shall become payable or any distribution other than cash shall be made, or whenever rights shall be issued with respect to the Deposited Securities, or whenever the Depositary shall receive notice of any meeting of holders of Shares or other Deposited Securities, or whenever for any reason the Depositary causes a change in the number of Shares that are represented by each American Depositary Share, or whenever the Depositary shall find it necessary, the Depositary shall fix a record date (a) for the determination of the Owners of Receipts who shall be (i) entitled to receive such dividend, distribution or rights or the net proceeds of the sale thereof, (ii) entitled to give instructions for the exercise of voting rights at any such meeting or (iii) responsible for any fees or charges assessed by the Depositary pursuant to the Deposit Agreement, or (b) on or after which each American Depositary Share will represent the changed number of Shares, subject to the provisions of the Deposit Agreement. The record date shall be fixed in accordance with any applicable rules of the New York Stock Exchange. The Depositary shall advise the Company and the New York Stock Exchange of any record date so fixed by the Depositary.

### 16. VOTING OF DEPOSITED SECURITIES.

Upon receipt of notice of any meeting of holders of Shares or other Deposited Securities, if requested in writing by the Company, the Depositary shall, as soon as practicable thereafter, mail to the Owners of record as of the record date set by the Depositary under Section 4.06 of the Deposit Agreement (the "Voting Record Date") a notice, the form of which notice shall be approved of by the Company which shall contain (a) such information as is contained in such notice of meeting, (b) a voting instruction card in the form prepared by the Depositary after consultation with the Company, (c) a statement that the Owners of record as of the close of business on the Voting Record Date will be entitled, subject to any applicable provision of English law and of the Memorandum and Articles of Association of the Company and the provisions of or governing the Deposited Securities, to either (i) use such voting instruction card to request the Depositary, its Custodian or nominee (as appropriate) to appoint the Owner its proxy to attend at that meeting and vote with respect to the number of Shares or other Deposited Securities represented by American Depositary Shares evidenced by such Owner's Receipts or (ii) instruct the person nominated by the Depositary, its Custodian or nominee as its proxy as to the exercise of the voting rights pertaining to that number of Shares or other Deposited Securities, and (d) a brief statement as to the manner in which voting instructions may be given to the person nominated by the Depositary.

Upon the written request of an Owner of a Receipt as of the Voting Record Date received on or before the date established by the Depositary for that purpose (the "Instruction

Date"), the Depositary shall endeavor to cause the appointment (or, if the Deposited Securities are registered in the name of or held by its Custodian or a nominee, the Depositary shall endeavor to procure that the Custodian or its nominee shall cause the appointment), subject to the articles of association of the Company, of that Owner as a proxy in respect of that meeting (including any adjournment of that meeting) to attend and vote the number of Deposited Securities represented by the American Depositary Shares evidenced by that Receipt.

Upon the written request of an Owner of a Receipt as of the Voting Record Date, received on or before the Instruction Date, the Depositary shall endeavor, in so far as practicable, to vote or cause to be voted the number of Shares or other Deposited Securities represented by the American Depositary Shares evidenced by such Receipt in accordance with the instructions set forth in such request. Neither the Depositary, nor the Custodian nor the nominee of either of them shall vote or attempt to exercise the right to vote that attaches to the Shares or other Deposited Securities other than in accordance with such written instructions from Owners given in accordance with Section 4.07 of the Deposit Agreement.

There can be no assurance that Owners generally or any Owner in particular will receive the notice described in the first paragraph of Section 4.07 of the Deposit Agreement sufficiently prior to the Instruction Date to ensure that the Depositary will appoint the Owner its proxy or vote the Shares or Deposited Securities as requested in accordance with the provisions set forth in the preceding paragraphs.

### 17. CHANGES AFFECTING DEPOSITED SECURITIES.

In circumstances where the provisions of Section 4.03 of the Deposit Agreement do not apply, upon any change in nominal value, change in par value, split-up, consolidation, or any other reclassification of Deposited Securities, or upon any recapitalization, reorganization, merger or consolidation, or sale of assets affecting the Company or to which it is a party, any securities which shall be received by the Depositary or a Custodian in exchange for or in conversion of or in respect of Deposited Securities shall be treated as new Deposited Securities under the Deposit Agreement, and American Depositary Shares shall thenceforth represent, in addition to the existing Deposited Securities, the right to receive the new Deposited Securities so received in exchange or conversion, unless additional Receipts are delivered pursuant to the following sentence. In any such case the Depositary may, and shall if the Company shall so reasonably request, execute and deliver additional Receipts as in the case of a dividend in Shares, or call for the surrender of outstanding Receipts to be exchanged for new Receipts specifically describing such new Deposited Securities.

### 18. LIABILITY OF THE COMPANY AND DEPOSITARY.

Neither the Depositary nor the Company nor any of their directors, employees, agents or affiliates shall incur any liability to any Owner or Beneficial Owner if, by reason of any provision of any present or future law or regulation of the United States or any other country, or of any other governmental or regulatory authority, or by reason of any provision, present or future, of the Memorandum and Articles of Association of the Company, or by reason of any provision of any securities issued or distributed by the Company, or any offering or distribution

thereof, or by reason of any act of God or war or terrorism or other circumstances beyond its control, the Depositary or the Company shall be prevented, delayed or forbidden from or be subject to any civil or criminal penalty on account of doing or performing any act or thing which by the terms of the Deposit Agreement or Deposited Securities it is provided shall be done or performed; nor shall the Depositary or the Company or any of their directors, employees, agents or affiliates incur any liability to any Owner or Beneficial Owner of a Receipt by reason of any non-performance or delay, caused as aforesaid, in the performance of any act or thing which by the terms of the Deposit Agreement it is provided shall or may be done or performed, or by reason of any exercise of, or failure to exercise, any discretion provided for in the Deposit Agreement. Where, by the terms of a distribution pursuant to Section 4.01, 4.02 or 4.03 of the Deposit Agreement, or an offering or distribution pursuant to Section 4.04 of the Deposit Agreement, such distribution or offering may not be made available to Owners of Receipts, and the Depositary may not dispose of such distribution or offering on behalf of such Owners and make the net proceeds available to such Owners, then the Depositary shall not make such distribution or offering, and shall allow any rights, if applicable, to lapse. Neither the Company nor the Depositary assumes any obligation or shall be subject to any liability under the Deposit Agreement to Owners or Beneficial Owners of Receipts, except that they agree to perform their obligations specifically set forth in the Deposit Agreement without negligence or bad faith. The Depositary shall not be subject to any liability with respect to the validity or worth of the Deposited Securities. Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be under any obligation to appear in or prosecute or defend any action, suit, or other proceeding in respect of any Deposited Securities or in respect of the Receipts on behalf of any Owner, Beneficial Owner or other person. The parties to the Deposit Agreement understand that the Custodian is not a party to the Deposit Agreement and, accordingly, the Custodian has no obligations whatsoever with respect to such proceedings, the responsibility of the Custodian being solely to the Depositary. Neither the Depositary nor the Company (nor any of their respective directors, officers, employees or agents) shall be liable for any action or nonaction by it in reliance upon the advice of or information from legal counsel, accountants, any person presenting Shares for deposit, any Owner or Beneficial Owner of a Receipt, or any other person believed by it in good faith to be competent to give such advice or information. Each of the Depositary, the Company and their respective directors, officers, employees and agents may rely and shall be protected in acting upon any written notice, request, direction or other document believed by such person to be genuine and to have been signed or presented by the proper party or parties. The Depositary and its agents will not be responsible for (i) any failure to carry out any instructions to vote any of the Deposited Securities or for the manner in which any such vote is cast, in each case to the extent the Depositary or its agents act without gross negligence or willful misconduct or (ii) for the effect of any such vote. Notwithstanding anything to the contrary set forth in the Deposit Agreement or any Receipt, the Depositary and its agents may fully respond to any and all demands or requests for information maintained by or on its behalf in connection with the Deposit Agreement, any Owner or Owners, any Receipt or Receipts or otherwise related hereto to the extent such information is requested or required by or pursuant to any lawful authority, including without limitation laws, rules, regulations, administrative or judicial process, banking, securities or other regulators. The Depositary shall not be liable for any acts or omissions made by a successor depositary whether in connection with a previous act or omission of the Depositary or in connection with a matter arising wholly after the removal or resignation of the Depositary, provided that in connection

with the issue out of which such potential liability arises, the Depositary performed its obligations without negligence or bad faith while it acted as Depositary. Neither the Company nor the Depositary nor any of their respective agents shall be liable to Owners or holders of interests in American Depositary Shares or any other third party or parties for any indirect, special, punitive or consequential damages. No disclaimer of liability under the Securities Act of 1933 is intended by any provision of the Deposit Agreement.

## 19. RESIGNATION AND REMOVAL OF THE DEPOSITARY; APPOINTMENT OF SUCCESSOR CUSTODIAN.

The Depositary may at any time resign as Depositary by written notice of its election so to do delivered to the Company, such resignation to take effect upon the appointment of a successor depositary and its acceptance of such appointment as provided in the Deposit Agreement. The Depositary may at any time be removed by the Company by written notice to the Depositary, such removal, to become effective upon the appointment of a successor depositary and its acceptance of such appointment as provided in the Deposit Agreement. Whenever the Depositary in its reasonable discretion determines that it is in the best interest of the Owners of Receipts to do so, it may appoint a substitute or additional custodian or custodians, after consultation with the Company.

### 20. AMENDMENT.

The form of the Receipts and any provisions of the Deposit Agreement may at any time and from time to time be amended by agreement between the Company and the Depositary without the consent of Owners or Beneficial Owners in any respect which they may deem necessary or desirable. Any amendment which shall impose or increase any fees or charges (other than taxes and other governmental charges, registration fees and cable, telex or facsimile transmission costs, delivery costs or other such expenses), or which shall otherwise prejudice any substantial existing right of Owners of Receipts, shall, however, not become effective as to outstanding Receipts until the expiration of thirty days after notice of such amendment shall have been given to the Owners of outstanding Receipts. Every Owner and Beneficial Owner of a Receipt at the time any amendment so becomes effective shall be deemed, by continuing to hold such Receipt or any interest therein, to consent and agree to such amendment and to be bound by the Deposit Agreement as amended thereby. In no event shall any amendment impair the right of the Owner of any Receipt to surrender such Receipt and receive therefor the Deposited Securities represented thereby except in order to comply with mandatory provisions of applicable law.

The Company and the Depositary shall each use their reasonable efforts to amend the Deposit Agreement as necessary to reflect changes in English or U.S. law and in the Company's Memorandum and Articles of Association.

## 21. TERMINATION OF DEPOSIT AGREEMENT.

The Depositary at any time at the direction of the Company, shall terminate the Deposit Agreement by mailing notice of such termination to the Owners of all Receipts then outstanding at least 30 days prior to the date fixed in such notice for such termination. The

Depositary may likewise terminate the Deposit Agreement by mailing notice of such termination to the Company and the Owners of all Receipts then outstanding if at any time 60 days shall have expired after the Depositary shall have delivered to the Company a written notice of its election to resign and a successor depositary shall not have been appointed and accepted its appointment as provided in the Deposit Agreement. On and after the date of termination, the Owner of a Receipt will, upon (a) surrender of such Receipt at the Principal Office of the Depositary and (b) payment of the fee of the Depositary for the surrender of Receipts referred to in Section 2.05 of the Deposit Agreement, and (c) payment of any applicable taxes or governmental charges, be entitled to delivery, to him or upon his order, of the amount of Deposited Securities represented by the American Depositary Shares evidenced by such Receipt, If any Receipts shall remain outstanding after the date of termination, the Depositary thereafter shall discontinue the registration of transfers of Receipts, shall suspend the distribution of dividends to the Owners thereof, and shall not give any further notices or perform any further acts under the Deposit Agreement, except that the Depositary shall continue to collect dividends and other distributions pertaining to Deposited Securities, shall sell rights and other property as provided in the Deposit Agreement, and shall continue to deliver Deposited Securities, together with any dividends or other distributions received with respect thereto and the net proceeds of the sale of any rights or other property, in exchange for Receipts surrendered to the Depositary (after deducting, in each case, the fee of the Depositary for the surrender of Receipts, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of the Deposit Agreement, and any applicable taxes or governmental charges) and for its obligation under Section 5.08 of the Deposit Agreement. At any time after the expiration of six months from the date of termination, the Depositary may sell the Deposited Securities then held under the Deposit Agreement and may thereafter hold uninvested the net proceeds of any such sale, together with any other cash then held by it thereunder, unsegregated and without liability for interest, for the pro rata benefit of the Owners of Receipts which have not theretofore been surrendered, such Owners thereupon becoming general creditors of the Depositary with respect to such net proceeds. After making such sale, the Depositary shall be discharged from all obligations under the Deposit Agreement, except to account for such net proceeds and other cash (after deducting, in each case, the fee of the Depositary for the surrender of a Receipt, any expenses for the account of the Owner of such Receipt in accordance with the terms and conditions of the Deposit Agreement, and any applicable taxes or governmental charges) and for its obligation under Section 5.08 of the Deposit Agreement. Upon the termination of the Deposit Agreement, the Company shall be discharged from all obligations under the Deposit Agreement except for its obligations to the Depositary with respect to indemnification, charges, and expenses.

## 22. COMPLIANCE WITH U.S. SECURITIES LAWS.

Notwithstanding anything in the Deposit Agreement to the contrary, the Company and the Depositary each agrees that it will not exercise any rights it has under the Deposit Agreement to permit the withdrawal or delivery of Deposited Securities in a manner which would violate the U.S. securities laws, including, but not limited to, Section I.A.(1) of the General Instructions to the Form F-6 registration statement, as amended from time to time, under the Securities Act of 1933.

### 23. DISCLOSURE OF INTERESTS.

To the extent that provisions of or governing any Deposited Securities (including the Company's Memorandum and Articles of Association or applicable English law) may require the disclosure of beneficial or other ownership of Deposited Securities, other Shares and other securities to the Company and may provide for blocking transfer and voting or other rights to enforce such disclosure or limit such ownership, the Depositary shall, to the extent reasonably practicable, comply with the Company's instructions as to Receipts in respect of any such enforcement or limitation, and Owners and Beneficial Owners of Receipts shall comply with all such disclosure requirements and ownership limitations and shall cooperate with the Depositary's compliance with such Company instructions. The Company may from time to time request Owners to provide information (a) as to the capacity in which such Owners own or owned American Depositary Shares, (b) regarding the identity of any other persons then or previously interested in such American Depositary Shares and (c) regarding the nature of such interest and various other matters pursuant to applicable law or the Memorandum and Articles of Association of the Company or other such corporate document of the Company, all as if such American Depositary Shares were to the extent practicable the underlying Shares. Each Owner agrees to provide any information requested by the Company or the Depositary pursuant to Section 3.04 of the Deposit Agreement whether or not such person is still an Owner at the time of the request. The Depositary agrees to use reasonable efforts to comply with written instructions received from the Company requesting that the Depositary forward any such requests to Owners and to forward to the Company any responses to such requests received by the Depositary.

#### 24. PROFILE MODIFICATION SYSTEM.

- (a) Notwithstanding the provisions of Section 2.04 of the Deposit Agreement, the parties acknowledge the Profile Modification System ("Profile") shall apply to Direct Registration Receipts upon acceptance thereof by DTC. Profile is a required feature of the Direct Registration System that allows a DTC participant, claiming to act on behalf of the Owner of Direct Registration Receipts, to direct the Depositary to register a transfer of the American Depositary Shares to DTC or its nominee and to deliver those American Depositary Shares to the DTC account of that DTC participant without receipt by the Depositary of prior authorization from the Owner to register such transfer.
- (b) In connection with and in accordance with the arrangements and procedures relating to Profile, the parties to the Deposit Agreement understand that the Depositary will not verify, determine or otherwise ascertain that the DTC participant that is claiming to be acting on behalf of an Owner in requesting a registration of transfer and delivery described in subsection (a) of Section 2.11 of the Deposit Agreement has the actual authority to act on behalf of the Owner. Each Owner agrees that neither the Company nor the Depositary shall have any liability for the Depositary's reliance upon information, or compliance with directions, it receives from a DTC participant as set forth in that subsection (a).

# 25. ARBITRATION OF DISPUTES.

- (a) The Company and each Owner, but not the Depositary, shall be bound by the arbitration and exclusive jurisdiction provisions set out in articles 152, 153 and 154 of the Articles as if all references therein to "shareholder" were replaced with "Owner." Articles 152, 153 and 154 of the Articles shall accordingly be incorporated, <u>mutatis mutandis</u> into the Deposit Agreement.
  - (b) The Company will make a copy of the Articles available to Owners upon request.
- (c) All cross-references to the Articles in Section 7.07 of the Deposit Agreement will be updated and amended without further action of any party in the event the Articles themselves are renumbered.